FRANCHISE DISCLOSURE DOCUMENT



Techy, LLC

(A Florida Limited Liability Company) 3000 South West 4th Avenue Fort Lauderdale, FL 33315 (877) 928-8905

Franchise@TechyCompany.com www.TechyCompany.com

The franchise we offer is the operation of a business for the installation, maintenance and repair of cell phones, computers, tablets, and smart home hardware and other electronic devices, and the sale of specified electronic devices, accessories and other supplies and items that we may designate from time to time under the trade name Techy Repairs & Smart Home Installs Powered by DrPhoneFix [®]. We may offer in limited markets, the option to add a Wrapped Techy Van or Techy Café to our single store franchise.

The total initial investment necessary to begin operation of a single store Inline franchise is \$173,600 to \$399,500. This includes \$22,500 that must be paid to the franchisor and a Start-Up Package of \$102,500 to \$227,500 paid to our affiliate, DrPhoneFix, Inc. In limited market areas we offer an In-Store franchise in Walmart facilities. The total initial investment necessary to begin operation ranges from \$172,500 to \$393,000. This includes \$22,500 that must be paid to the franchisor and a Start-Up Package of \$102,500 to \$227,500 paid to our affiliate DrPhoneFix, Inc. and requires you to sublease the store premises from DrPhoneFix, Inc. We may agree to a Wrapped Techy Van in some markets. The total initial investment for the Wrapped Techy Van is \$60,000 to \$83,000. This includes \$26,000 to \$36,500 that must be paid to our affiliate, DrPhoneFix, Inc. If the Wrapped Techy Van is added to the single store franchise, your total initial investment will include the single store total investment. In a limited number of territories, we may approve the use of a Wrapped Techy Van franchise without a physical store location in the territory. The total initial investment necessary to begin operation of a Wrapped Techy Van franchise without a physical store location in the territory ranges from \$82,500 to \$105,500. This includes \$22,500 that must be paid to the franchisor and \$26,000 to \$36,500 that must be paid to our affiliate, DrPhoneFix, Inc.

We also offer an Area Development Agreement that allows you to open a minimum of two stores within a designated development territory for a total initial investment of \$47,500 to \$80,000. This includes a development fee of \$25,000 that must be paid to the franchisor. You are required to develop an agreed number of stores within a designated territory with a reduced initial franchise fee after the first store, for each store to be developed. The initial franchise fee for the second store to be developed is \$10,000, and \$5,000 for the third store and each store thereafter. When you sign the Area Development Agreement you will pay the initial franchise fee for the first store plus one-half of the initial franchise fee for each store thereafter. (See Items 5 and 7 for additional information.)

This disclosure document summarizes certain provisions of your franchise agreement and other information in plain English (the "Disclosure Document"). Read this Disclosure Document and all accompanying agreements carefully. You must receive this Disclosure Document at least 14 calendardays before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no governmental agency has verified the information contained in this document.**

You may wish to receive your Disclosure Document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact the Franchise Administration Department at Techy, LLC, 3000 South West 4th Avenue, Fort Lauderdale, FL 33315, (877) 928-8905.

The terms of your Franchise Agreement will govern your franchise relationship. Do not rely solely upon this Disclosure Document to understand your Franchise Agreement. You should read and review all of your Franchise Agreement carefully with an advisor, such as an attorney or an accountant.

Buying a franchise is a complex investment. The information in this Disclosure Document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this disclosure document, is available from the Federal Trade Commission ("FTC"). You can contact the FTC at 1-877-FTC-HELP or in writing at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency, listed in Exhibit G, or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Date of Issuance: April 20, 2023

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTIONS	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Exhibits B and C.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Exhibit D includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only Techy, LLC business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be a Techy, LLC franchisee?	Item 20 or Exhibit B lists current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees.</u> You may have to pay royalties and other fees even if you are losing money.

Business model can change. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions.</u> You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions.</u> The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor.</u> Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

Renewal. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

<u>When your franchise ends.</u> The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit G.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About This Franchise

Certain states require that the following risk(s) be highlighted:

1. <u>Out-of-State Dispute Resolution</u>. The franchise agreement and area development agreement require you to resolve disputes with the franchisor by mediation, arbitration and/or litigation only in Florida. Out-of-state mediation, arbitration, or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to mediate, arbitrate, or litigate with the franchisor in Florida than in your own state.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

THIS MICHIGAN NOTICE APPLIES ONLY TO FRANCHISES WHO ARE RESIDENTS OF MICHIGAN OR LOCATE THEIR FRANCHISES IN MICHIGAN. NOTICE REQUIRED BY STATE OF MICHIGAN

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU.

Each of the following provision is void and unenforceable if contained in any documents relating to a franchise:

- (a) A prohibition on the right of a franchisee to join an association of franchisees.
- (b) A requirement that a franchisee assent to a release assignment, notation, waiver, or estoppels which deprives a franchisee of rights and protections provided in this act. This shall not preclude a franchisee, after entering into a franchise agreement, from setting any and all claims.
- (c) A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provision of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- (d) A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration of the franchisee's inventory, supplies, equipment, fixtures and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies of the franchise business are not subject to compensation. This subsection applies only if: (i) the term of the franchise is less than five (5) years and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under the trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchise does not receive at least six (6) months advance notice of the franchisor's intent to renew the franchise.
- (e) A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- (f) A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside the state.
- (g) A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for a good cause. This subdivision does not prevent a franchisor from exercising a right to first refusal to purchase the franchise. Good cause shall include, but not limited to:
- (i) The failure of the proposed transferee to meet the franchisor's then current reasonable qualification or standards.
 - (ii) The fact that the proposed transferee is a competitor of the franchisor or subfranchisor.
- (iii) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.

THIS MICHIGAN NOTICE APPLIES ONLY TO FRANCHISEES WHO ARE RESIDENTS OF MICHIGAN OR LOCATE THEIR FRANCHISES IN MICHIGAN.

- (iv) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.
- (h) A provision that requires the franchisee to resell to the franchisor items that are not uniquely identifies with the franchisor. This subdivision does not permit a provision that grants to a franchisor a right of refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (c).
- (i) A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENDORSEMENT BY THE ATTORNEY GENERAL.

Any questions regarding this notice should be directed to:

Michigan Department of Attorney General Consumer Protection Division Mennen Williams Building 525 West Ottawa Street Lansing, Michigan 48933 (517) 335-7567

TABLE OF CONTENTS

<u>ITEM</u> 1.	THE FRANCHISOR AND ANY PARENTS, PREDECESSORS	PAGE
1.	AND AFFILIATES	1
2.	BUSINESS EXPERIENCE	3
3.	LITIGATION	4
4.	BANKRUPTCY	5
5.	INITIAL FEES	5
6.	OTHER FEES	6
7.	ESTIMATED INITIAL INVESTMENT	10
8.	RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES	17
9.	FRANCHISEE'S OBLIGATIONS	22
10.	FINANCING	25
11.	FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND TRAINING	26
12.	TERRITORY	38
13.	TRADEMARKS	42
14.	PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION	44
15.	OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS	45
16.	RESTRICTIONS ON WHAT THE FRANCHISE MAY SELL	46
17.	RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION	46
18.	PUBLIC FIGURES	53
19.	FINANCIAL PERFORMANCE REPRESENTATIONS	53
20.	OUTLETS AND FRANCHISEE INFORMATION	54
21.	FINANCIAL STATEMENTS	57
22.	CONTRACTS	57
23.	RECEIPTS	57

EXHIBITS

Exhibit A Franchisee Operations Manual Table of Contents Exhibit B List of Franchisees Exhibit C List of Franchisees Who Have Left the System Exhibit D **Financial Statements** Exhibit E Franchise Agreement Exhibit E-1 In-Store (Walmart) Addendum to Franchise Agreement Exhibit E-2 Start-Up Package Addendum to Franchise Agreement Exhibit E-3 Techy Café Addendum to Franchise Agreement Exhibit E-4 Wrapped Techy Van Addendum to Franchise Agreement Exhibit F Area Development Agreement Exhibit F-1 In-Store (Walmart) Addendum to Area Development Agreement Exhibit G List of State Agencies/Agents for Service of Process Exhibit H State Specific and other Addenda and Riders Exhibit I Secured Promissory Note Exhibit J Security Agreement Exhibit K Franchisee Questionnaire Exhibit L Principal Owner's Statement Exhibit M Receipts

ITEM 1. THE FRANCHISOR, AND ANY PARENTS, PREDECESSORS AND AFFILIATES

To simplify the language, this disclosure document uses "Techy Repairs & Smart Home Installs Powered by DrPhoneFix", "we", "us", or "our" to mean Techy, LLC, the franchisor, and "you" or "your" to mean the individual, corporation or other entity that buys a franchise.

The Franchisor, Any Parents, Predecessors and Affiliates

We are a Florida Limited Liability Company formed August 3, 2016 and began offering franchises August 29, 2016. Our principal business address is 3000 South West 4th Avenue, Fort Lauderdale, Florida 33315. Our predecessor, DrPhoneFix Franchises, Inc. which was formed October 25, 2014, previously franchised nine stores and subsequent to our formation assigned the Franchise Agreements to us and ceased franchising. We also maintain support for the franchises, assigned to us. Our affiliate, DrPhoneFix, Inc. previously entered into license agreements to operate businesses using the DrPhoneFix name. As of the date of this disclosure document there are two (2) licensed operations in the State of Florida. We began offering franchises on August 29, 2016. On July 27, 2020, we amended our name from DrPhoneFix USA, LLC to Techy, LLC and began using our trademarked name and logo "Techy Repairs and Smart Home Installs Powered by DrPhoneFix." We have not sold franchises prior to August 29, 2016 of any type at any time. We have no parents.

DrPhoneFix Franchises, Inc., our predecessor, was located at 3000 South West 4th Avenue, Fort Lauderdale, Florida 33315, and was formed in October 2014 to launch a franchising program. DrPhoneFix Franchises, Inc. offered franchises from 2014 to August 29, 2016 when it sold and assigned all franchise agreements to us along with all of its registered trademarks and thereafter ceased operations. DrPhoneFix Franchise, Inc. did not sell franchises in any other line of business.

Our affiliate, DrPhoneFix, Inc. located at 3000 South West 4th Avenue, Fort Lauderdale, Florida 33315 owned 1 location at the end of 2015 and acquired two (2) locations in 2016. DrPhoneFix, Inc. will not participate in the franchising operations nor will it provide management or operational services or financial guarantees for Techy, LLC except that DrPhoneFix, Inc. will sublease Walmart locations to franchisees of the In-Store (Walmart) franchise concept and will provide an opening start-up package for franchisees and use may be made of some of the affiliate facilities and personnel for training purposes.

Our affiliate Cellular Supplies, Inc. located at 3000 South West 4th Avenue, Fort Lauderdale, Florida 33315 was formed May 24, 2006 to sell cellular phone accessories. Cellular Supplies, Inc. is owned by our chief executive officer and is a supplier of cellular phone accessories and has not sold franchises previously at any time.

On August 20, 2021 our affiliate, Techy Company LLC located at 3000 South West 4th Avenue, Fort Lauderdale, Florida 33315 and formed October 27, 2020, acquired certain franchise agreements of Experimax Franchising, LLC. These franchises are in the business of purchasing and reselling used electronic products such as laptop computers, tablets and cell phones plus new accessories and components for these products. Our affiliate will maintain support for the franchises acquired by it. Our affiliate will not participate in our franchising operations, nor will they provide any management or operational services or financial guarantees for our franchisees.

Agent for Service of Process

Our agents for service of process and states requiring franchise registration are disclosed in Exhibit G.

The Franchises Being Offered

Inline Location

Our Inline Franchise involves the franchising of our distinctive system to operate a business for the installation and sale of smart home products and repair of cell phones, tablets, computers, game consoles and other electronic devices as well as the sale and installations of other specified electronic devices, accessories, supplies and items we may designate from time to time. In our Inline Franchise you will locate the store site (usually in an in-line center) and submit the store location within a protected territory, for our approval.

In-Store (Walmart) Location

Our In-Store franchise is located within a Walmart store at available space within the store location. We may add to or delete the business format of our In-Store (Walmart) franchise from time to time.

Area Development Agreement

We offer and award to qualified applicants a franchise to operate a single Store using our marks, design and system (the "Techy Repairs & Smart Home Installs Powered by DrPhoneFix Systems" or "System"). In addition, we also offer qualified persons the right to operate a specified number of additional franchises in a designated territory under the terms of an Area Development Agreement (the "Development Agreement"). Pursuant to the Development Agreement you must open a certain number of Inline franchises within a specified period of time (see Item 5), or if the In-Store (Walmart) franchise addendum is entered into, you must open a certain number of In-Store (Walmart) franchises as they are made available in your designated Development Territory. You will pay a development fee, which will depend upon the number of stores to be opened within a defined geographical area over a required period of time. You will also sign the then current form of franchise agreement for each location, which may be different than the form of franchise agreement included in this franchise disclosure document and other factors specified in the Area Development Agreement (see Item 5).

Market and Competition

As a Techy Repairs & Smart Home Installs Powered by DrPhoneFix franchisee, you will be competing with other individually owned and other franchised businesses offering similar products and services in the general vicinity of your store. The Smart Home hardware, cell phone, computer and electronic device and accessories sales and repair industry is a competitive market.

Although there is never any guarantee of success, we believe that the market for the products and services offered by our franchisees is part of a growing and developing market with room for further development and expansion throughout the United States and other countries. You will compete with a large number of other store chains (franchised and non-franchised), including other

"Techy Repairs & Smart Home Installs Powered by DrPhoneFix franchised stores" (hereinafter "franchised store" or "franchised stores"). The success of your business will depend largely on your personal skills and abilities.

Industry Specific Regulations

As a reseller of used electronic products you will be subject to state and local laws regulating dealers in second-hand goods. These laws will require you to obtain a license to operate the business. To obtain the license you may be subject to a criminal background check. These laws will also regulate aspects of your business including gathering and returning information regarding the sellers of used products sold by your Store and the length of time that must expire before an item can be resold.

Federal, state and local government entities have enacted laws, rules, regulations and ordinances which may apply to the operation of your franchised store, including those which: (a) establish general standards, specifications and requirements for the construction, design and maintenance of the store premises; (b) regulate matters affecting the health, safety and welfare of your customers; (c) set standards pertaining to employee health and safety; (d) set standards and requirements for fire safety and general emergency preparedness; and (f) regulate the proper use, storage and disposal of hazardous materials. You should investigate whether there are regulations and requirements that may apply in the geographic area in which you are interested in locating your franchised business and should consider both their effect and cost of compliance. In addition to obtaining any applicable city or county business licenses, you must comply with all local, state and federal laws pertaining to businesses, including any business, retail sales, zoning, environmental, labor relations, sanitation, safety, fire, health codes, regulations and ordinances, as well as laws and regulations relating to access by persons with disabilities. You are solely responsible for identification of and compliance with all laws, ordinances and/or regulations applicable to your franchise. You should independently research and review the legal requirements of the industry with your own attorney before you sign any binding documents or make any investments. It also is your sole responsibility, on an ongoing basis, to investigate and satisfy all employment, worker's compensation, insurance, corporate, tax and similar laws and regulations, since they vary from region to region, can change over time, and may affect the operation of your business.

ITEM 2. BUSINESS EXPERIENCE

Chief Executive Officer and Director: William J. Daragan

William Daragan has been President and CEO since our formation August 3, 2016 (Fort Lauderdale, Florida). He has also served as President and CEO of DrPhoneFix, Inc. and DrPhoneFix Franchises, Inc., both companies located in Fort Lauderdale, Florida since their inception in January 2012 and October 2014, respectively. Prior to that he served and continues to serve as President and CEO of Cellular Supplies Inc. located in Fort Lauderdale since May 2006. Mr. Daragan is also President and CEO of Techy Company, LLC in Fort Lauderdale since its inception August 2021.

Chief Operating Officer: Timothy R. Phelps

Timothy Ryan Phelps has been Chief Operating Officer for our company since August 3, 2016. He has also been Vice President of Operations for DrPhoneFix, Inc. in Fort Lauderdale Florida since January 2012 and held the same position for DrPhoneFix Franchises, Inc. in Fort Lauderdale, FL since October 2014. Prior to joining us, Mr. Phelps was Operations Manager of two cell phone repair businesses; one in Homestead, Florida starting in 2008; the other in Plantation, Florida starting in 2011.

Director of Development: Daniel Daragan

Daniel Daragan has been our Director of Development since our formation August 3, 2016. Daniel Daragan has held the same position for DrPhoneFix Franchises, Inc. since its formation in October 2014. He also served as CEO of My Branding System from 2010 to 2013 located in Hollywood, Florida. Prior to that, he was a real estate investor, as well as President and CEO of London Ben Fish & Chips in Hollywood Florida for 17 years. Mr. Daragan began his career working for McDonald's Restaurant Corporation in South Florida where he supervised numerous employees and oversaw the operations of multiple locations and eventually served as a liaison between the company and franchisees.

ITEM 3. LITIGATION

Pending

- 1. Chance 308, LLC v. DrPhoneFix, Inc. and Techy, LLC, filed December 6, 2022 in Miami Dade County (Case No: 2022-023078-CA-01) by franchisee alleging breach of sublease, franchise agreement and unjust enrichment. Defendants filed a Motion to Compel Arbitration and the case is pending. Additionally, there is an eviction matter pending in the same court DrPhoneFix, Inc. v. Chance 308, LLC, filed February 14, 2023 in Miami Dade County (Case No: 2023-02502-CA-01). Defendant filed a motion to transfer and the matter is pending.
- 2. Troya Group Investment, LLC v. DrPhoneFix, Inc. d/b/a Techy by DrPhoneFix and William Daragan, filed May 25, 2022 in Broward County, FL (Case No: CACE 22-007625) by franchisee alleging fraudulent inducement and requesting recission of the Franchise Agreement. Defendants filed a Motion to Compel Arbitration which was granted April 2, 2023. The arbitration is pending.

Concluded

1. Techy, LLC v. Mohammad Noman, filed May 26, 2022 in the United States District Court Southern Florida (Case No: 0:22-cv-61010-WPD) by franchisor for willful trademark infringement, unfair competition and breach of contract. On February 28, 2022 DrPhoneFix, Inc. filed suit for eviction in Pittsburgh's Magisterial District Court (Case No: MJ-05317-LT-0000022-2022). The parties entered into a settlement agreement in which defendant vacated the premises and agreed to pay \$33,000. A joint stipulation of dismissal was filed in the federal case and an order of dismissal entered August 31, 2022. The eviction case was closed on March 22, 2022.

- 2. New Age Cell, Inc. v. DrPhoneFix Franchises, Inc., filed June 26, 2018 in the Circuit Court of Broward County, Florida (Case No: CACE 18-015484)(03) by Franchisee for violation of Florida Statues, unjust enrichment, breach of contract and fraudulent inducement. Defendant filed a motion to compel arbitration. Defendant filed a motion to compel arbitration and the parties subsequently entered into a General Release on February 6, 2019 in which the Franchise Agreement contractual obligations of confidentiality and covenants not to compete survived. There was no monetary consideration exchanged. A Notice of Voluntary Dismissal With Prejudice was filed February 21, 2019.
- 3. Smart Phone Fix Clermont, LLC, Smart Phone Fix Cape Coral, LLC, Smart Phone Fix Naples North, LLC, Smart Phone Fix, LLC, Kayton Phone Fix Corp. and Kayton Phone Fix 2 Corp. v. Techy, LLC, Daniel Daragan, G and E Realty Group, Inc. and Walmart, Inc. (Case No: CACE 18-017305) filed July 19, 2018 by franchisees after termination of the Franchise Agreement by Techy, LLC. The complaint alleges violation of Florida statutes, fraud, negligent misrepresentation and third party beneficiary. Defendants filed a motion to compel arbitration, which was granted July 8, 2020, and Plaintiff thereafter filed a Notice of Voluntary Dismissal Without Prejudice on May 18, 2021.

There is no other litigation required to be disclosed in this Disclosure Document.

ITEM 4. BANKRUPTCY

No bankruptcy information is required to be disclosed in this Disclosure Document.

ITEM 5. INITIAL FEES

Initial Franchise Fee.

At the time the Franchise Agreement is signed, a non-refundable \$15,000 franchise fee must be paid to Techy, LLC. Proceeds from the initial Franchise fee are used to help defray the cost of developing and improving our services, for expenses of preparing and when required, registering this disclosure document, legal fees, accounting fees, cost of obtaining and screening franchisees, to defray expenses incurred in connection with the operation of our corporate office, developing and refining standards, developing procedures, techniques and other elements of our franchise system and to provide compensation for our efforts.

Area Development Agreement.

We may also offer you an Area Development Agreement (the "Development Agreement") that will allow qualified persons to develop additional store locations in a designated Development Territory at progressively reduced fees. The minimum number of stores that must be opened under the area development program is two stores. Upon signing a Development Agreement, you will pay an initial, lump sum, non-refundable Development Fee consisting of \$15,000 for the first franchised location, plus one-half of the progressively reduced Initial Franchise Fees (\$10,000 second store, \$5,000 third store and every additional store) for each additional franchised store to be opened under the Development Agreement. An Area Developer will also sign a separate Franchise Agreement for each location.

Area Development Agreement

		Amount Paid On	
Number of Franchise	Total	Signing	
Stores Required To	Development	Development	Balance Due At the Time You Sign Each of
Be Opened	Fee	Agreement	the Remaining Franchise Agreements
Two	\$25,000	\$20,000	0 (1) + 5k(2)
Three	\$30,000	\$22,500	0(1) + 5k(2) + 2.5k(3)
Four	\$35,000	\$25,000	0(1)+5k(2) + 2.5k(3) + 2.5k(4)
Five	\$40,000	\$27,500	0(1)+5k(2)+2.5k(3)+2.5k(4)+2.5k(5)

Note that the second franchise has a reduced fee of \$10,000 in addition to the \$15,000 fee for the first franchise, for a total Development Fee of \$25,000. However, only half of the fee for the second franchise is paid when signing the Development Agreement. The remaining half (\$5,000) is paid upon signing the franchise agreement to open the second franchise.

For each additional area development franchise in excess of 2, the initial franchise fee will be \$5,000 of which \$2,500 will be payable upon signing each additional Franchise Agreement.

Training Fee.

We provide certified classroom and on-the-job training consisting of seventy-two (72) hours of classroom training and forty (40) hours of on-the-job training at a cost of \$7,500.

Start-Up Package.

We offer a Start-Up Package for our Inline and In-Store (Walmart) franchises. The Start-Up Package ranges in price from \$102,500 to \$227,500 and differs primarily in the size of the leased space, amount of inventory and the cabinetry provided. The package includes improvements and preparation of the leased space, cabinetry, shelving, furnishings, furniture, signage, equipment, inventory, computer, POS and related software. The Start Up Package is acquired from our affiliate, DrPhoneFix, Inc.

Wrapped Techy Van.

We offer an optional Wrapped Techy Van. The Wrapped Techy Van includes inventory, computer hardware and software, signage (including the wrap), equipment and shelving that is acquired from our affiliate, DrPhoneFix, Inc.

ITEM 6. OTHER FEES

Type of Fee (1)	Amount	Due Date	Remarks
Weekly Continuing Royalty Fee	7% of Gross Revenues received by you from the previous week's operations. If your	Payable on Monday at 3 pm of each week with a report detailing all Gross Revenues (2)	(1) Amounts will be debited electronically by Electronic Funds Transfer ("EFT") from your bank account. Subject to inflation adjustment.

Type of Fee (1)	Amount	Due Date	Remarks
	gross reaches \$7,500 for the week, all royalties for that week will be 6%	received during the previous 7-day week (Monday through Sunday).	(2) In the absence of a timely weekly report we will debit a minimum fee of \$500 from your bank account and resolve any difference or discrepancy when the report is made. A \$100 fee per occurrence of insufficient EFT will be charged.
Interest and Late Fees	Interest is highest applicable legal rate for open account business credit. (Not to exceed 1.5% per month). Additionally, we may require you to pay an administrative late fee of Fifty Dollars (\$50) for each late report and/or late payment.	Payable on demand	Interest on all amounts owed us. We may also require payment by cashier's check for repeated late payments.
Successor Fee	20% of the then- current Initial Franchise Fee	At the time you sign the Successor Franchise Agreement	
Marketing and Technology Fund Fee (3)	2% of Gross Revenues of previous month or \$785 whichever is greater.	Fee will be electronically debited (Electronic Funds Transfer "EFT") from your bank in the same manner as the Weekly Continuing License fee, on the 5th day of each month.	May be increased up to 10% per year upon a thirty (30) day written notice.
Local Marketing Expenses	1.5% of Gross Revenues for previous month or \$500 whichever is greater.	Must spend this minimum every month.	Paid to local vendors. Subject to inflation adjustment. As described 9.6 of the franchise agreement.
Advertising Cooperative	To be determined	To be determined	See Note ⁽⁴⁾ below
Insurance Reimbursement	Will vary under circumstances	As incurred	If you fail to obtain the required insurance coverage for your store, we may obtain the

Type of Fee (1)	Amount	Due Date	Remarks
			coverage at your expense plus an administrative fee of 15% Franchise Agreement 10.6.
Transfer Fee	If you transfer your franchise, your franchise agreement requires payment of a \$10,000 transfer fee to us and a \$7,500 Training fee.	At the time of your application for a transfer	The transferee fee is non refundable. (Franchise Agreement, 14.3).
Transfer Fee – Area Development Agreement	\$7,500	On transfer of any interest in the Area Development Agreement	We may, but are not required to, waive or reduce this fee at our sole discretion.
Audit Fee	Cost of Audit and 1½ % interest per month on understated amount. Your cost might range from \$500 to \$1,500 depending on the complexity of the audit.	Due when billed following any such audit.	You pay this fee only if the audit reflects an understatement of 2% or more for any audited month.
Cellphone Accessory Support Fee	\$100	Monthly by Electronic Funds Transfer on the 10 th day of each month.	Paid monthly but waived when you purchase \$300/month or more of cellphone accessories.
Call Log PBX Software	\$50	Monthly by Electronic Funds Transfer on the 10 th of each month.	Payable to Us
Non Compliance Fee	\$250 per day	If incurred.	If an inspection of your store finds unapproved vendor items.
Manual Replacement Fee	\$250	As needed.	If you lose or misplace your Manual, you will pay us a replacement fee.
Seminars and Training Conferences	Cost of attending regional and national conferences as we require.	Periodic Seminars and Conferences will be held at our company headquarters in Florida or at a different location of our choosing. These events will not exceed 8 days.	Fee is payable prior to attendance at the event and expenses are paid as incurred. You will not be required to attend more than two such conferences per year.

Type of Fee (1)	Amount	Due Date	Remarks
Additional Training Requested by You at Your Store	Currently \$400 per day, plus travel and expenses.	Immediately after notice from us.	If, at your request, and subject to availability, we may send one of our staff members to the franchised business to provide further assistance, we will charge you a daily rate for that assistance, plus the travel expenses for our employee.
Costs and Attorney's Fees	Will vary under the circumstances.	Immediately upon notice from us.	You only pay if we succeed in any arbitration or litigation we bring against you, or in defending any claim you bring against us.
Indemnification	Will vary under the circumstances.	As incurred.	You have to reimburse us if we are sued or held liable for claims arising out of your business.
Customer Satisfaction/Quality Control Measures	To be determined.	To be determined	We may institute programs for auditing customer satisfaction and/or other quality control measures. We may require you to pay program costs.
Review and Approval of Supplier	\$1,000	On submission by you of supplier for review.	Payable only if you submit a proposed supplier for our review.
Unauthorized Failure to Be Open	\$250 per unauthorized day you are closed without written permission from us.	As it occurs, the fee will be an EFT debit from your bank account.	You pay this fee to us if you are not authorized to be closed.
Management Fee (only on default by you)	\$300 per day plus expenses.	Deducted from funds of your Businesses	Expenses include compensation, other costs, travel and living expenses of appointed manager.
Area Development Extension	\$5,000 to \$20,000	If schedule extended	Fee varies with amount of time only if extended (see Item 12)

- (1) All fees described in this Item 6 are the same for each Franchise unless otherwise noted. All fees are imposed by and payable to us and are non-refundable unless otherwise noted. You must participate in an electronic funds transfer and reporting program, which will authorize us to use a pre-authorized bank debit system.
- (2) <u>Gross Revenues</u> includes all charges and/or Revenues which are, or could be, received or earned by you (and/or any Affiliate of yours):

All sales and/or billings, whether collected or not, will be included in Gross Revenues, with no deduction for credit card or other charges. Gross Revenues does <u>not</u> include sales tax collected

and paid when due to the appropriate taxing authority and provable customer refunds, adjustments and credits.

- (3) Marketing and Technology Fund Fee. You pay this amount to us to support the creative development and production by us (or our agents selected by us) of all print and electronic (worldwide websites), promotion, publicity, digital and other marketing and advertising programs as we may from time to time research, develop and provide. We reserve the right to decide the amount and nature of these expenditures and how, when and where all expenditures are made Proceeds from the marketing fund will not be utilized for the direct personal benefit of any specific, individual Franchisee or on any pro rata basis whatsoever, but will be utilized generally for local, regional or national marketing and advertising programs to support the franchise system as a whole (See Item 11).
- (4) <u>Advertising Cooperatives</u>. In the future we may form one or more local, regional or national cooperatives which you will be required to join if the cooperative covers your area. You may be required to contribute such amounts as are determined by the cooperative.

ITEM 7. ESTIMATED INITIAL INVESTMENT

YOUR ESTIMATED INITIAL INVESTMENT ¹ TECHY INLINE FRANCHISE

Description	Estimated Amount	Method of Payment	When Due	To Whom Payable
Initial Franchise Fee ²	\$15,000 (For Single Store) ²	Lump Sum	On Signing Franchise Agreement	Us
Training ³	\$7,500	Lump Sum	On Signing Franchise Agreement	Us
Start -Up Package ⁴	\$102,500 - \$227,500	Lump Sum	On Signing Franchise Agreement	DrPhoneFix, Inc.
Travel and Expenses While Training ⁵	\$1,000 - \$20,000	Lump Sum	As Incurred	Transportation Lines, Hotel Facilities, etc.
Real Estate 3 Months Initial Rent ⁶	\$9,600 - \$19,500	As Arranged	As Incurred	Lessor/Owner of Real Estate
Utility Deposits, Application Fees, Permits ⁷	\$1,000 - \$4,000	Lump Sum	As Incurred	Utility Companies, Cities and Municipalities
Additional Inventory Cost	\$750 - \$4,000	Lump Sum	As Incurred	Designated Suppliers
Business License / Occupational License	\$1,000 - \$4,000	Lump Sum	As Incurred	City Agencies, normally depends upon the number of employees.

Description	Estimated Amount	Method of Payment	When Due	To Whom Payable
Insurance 8	\$750 - \$1,500	As Arranged	As Incurred	Insurance Agencies
Legal and Accounting	\$2,000 - \$20,000	As Arranged	As Incurred	Attorney, Accountant
Surveillance System and Alarm System	\$500 - \$1,500	Lump Sum	As Incurred	Designated Suppliers/ Vendors
Opening Day Marketing Program	\$2,000 - \$3,000	As Arranged	As Incurred	Designated Suppliers/Vendors
Marketing First Sixty Days Program 9	\$4,000 - \$6,000	As Arranged	As Incurred	Designated Suppliers/Vendors
Miscellaneous Costs	\$1,000 - \$6,000	As Arranged	As Incurred	Us/Vendors/ Designated Suppliers
Additional Funds – Six (6) Months ¹⁰	\$25,000 - \$60,000	As Arranged	As Incurred	Employees, Designated Suppliers, Vendors
TOTAL	\$173,600 - \$399,500			

NOTES:

- (1) <u>Estimated Initial Investment</u>. This table shows the estimated expenditures required to develop a single Franchise Techy Store.
- (2) <u>Initial Franchise Fee Will Vary</u>. If you are approved to sign an Area Development Agreement, you pay a lump sum, non-refundable Development Fee of \$15,000 for your first franchised store, plus one-half of the progressively reduced Initial Franchise Fee for each additional store to be opened under your agreement. The initial amount paid is credited against the balance of the remaining Initial Franchise Fees to be paid by you for each store as they are subsequently opened. Development of three store locations requires a commitment of \$30,000 in franchise fees, of which \$22,500 would be paid upon signing the Development Agreement, with the balance being paid upon signing individual Franchise Agreements for each location. (See Item 5 for detailed explanation).
- (3) Training. Cost for obtaining certified training.
- (4) <u>Start-up Package</u>. You will pay DrPhoneFix, Inc. a Start-Up Package fee for preparing the leased space in a ready to use condition which will include furnishings, computer, POS and software, signage, equipment, and start-up inventory. Your start-up package fee will be primarily dependent upon the size of the leased premises, the amount of inventory and cabinetry you choose. The range of the Start-Up Package is set out below.

Description	Low	High
Tenant Improvement and Remodeling	\$29,000	\$68,500
Cabinetry, Shelving, Furnishings, Furniture	\$28,000	\$60,000
Computer Hardware/Software, POS and PC	\$2,500	\$4,000
Signage	\$7,000	\$15,000
Equipment	\$26,000	\$50,000
Inventory	\$10,000	\$30,000
Total	\$102,500	\$227,500

- (5) <u>Travel and Expenses While Training.</u> You are responsible for arranging and paying the expenses for transportation, meals and lodging for you and others while attending the training program. The amount you spend will depend on several factors, including the distance you travel, your method of travel and the type of accommodations you choose. We base this estimate upon attendance by three people.
- (6) Real Estate Deposits and Initial Rent. Real estate costs can vary widely depending upon a multitude of factors related to location, including whether the property is purchased or leased, local market conditions, the size, type, condition and location of the property and zoning requirements. The size of a typical site is between 500 and 1,800 square feet. The Techy Café requires additional space between 100 to 500 square feet.
- (7) <u>Utility Deposits, Permits</u>. These figures do not include any special connection fees.
- (8) <u>Insurance</u>. You must maintain in force policies of insurance issued by carriers we have approved covering various risks. We may specify the types and amounts of coverage required under these policies and require different and/or additional kinds of insurance at any time, including excess liability insurance. Each insurance policy must name us and our Affiliates as additional named insureds, will contain a waiver of all subrogation rights against us and our Affiliates and any successors and assigns, and must provide for 30 days' prior written notice to us of any material modifications, cancellation, or expiration of the policies. (See Item 8 for insurance requirement details).
- (9) <u>Marketing First Sixty Days.</u> During the sixty (60) day period after opening, you must spend between \$4,000 to \$6,000 on Franchisor approved marketing of your store.
- (10) Additional Funds. An estimate of funds to cover business (not personal) expenses during the first six (6) months of operation of your business. You will need capital to support ongoing costs of your business, such as payroll, utilities, taxes, loan payments and other expenses, to the extent that Revenues do not cover business costs. You may need additional funds during the first six (6) months of initial operation or afterwards. We do not furnish or authorize our salespersons or any other persons or entities to furnish estimates as to the capital or other reserve funds necessary to reach "break-even" or any other financial position, nor should you rely on any such estimates. The six (6) month period from beginning a business covers the time by which most Franchisees are fully operational. In addition, the estimates presented relate only to costs associated with the franchised business and do not cover any personal, "living," unrelated business or other expenses you may have. We recommend that, in addition to the amount estimated here, you have sufficient personal savings and/or income so that you will be self-sufficient for a period of time.

The estimates assume that you will lease the location of your franchise. Ranges are shown for some items because they depend upon factors that may vary such as site, location, as well as other local market conditions and circumstances. The costs listed are estimates only. All fees and payments are non-refundable, unless otherwise stated or permitted by the payee. You should review these estimates carefully with your business advisor before making a decision to enter into a Franchise Agreement.

YOUR ESTIMATED INITIAL INVESTMENT ¹ IN-STORE (WALMART) FRANCHISE

		KE (WALMAKI)		T- W1
7	Estimated	Method		To Whom
Description	Amount	Payments	When Due	Payable
Initial Franchise	\$15,000 – (For		On Signing	
Fee ²	Single Store)	Lump Sum	Franchise	Us
1 00	bingle btore)		Agreement	
			On Signing	
Training ³	\$7,500	Lump Sum	Franchise	Us
			Agreement	
Start-Up	\$102,500 -		On Signing	DrPhoneFix,
Package ⁴	\$227,500	Lump Sum	Franchise	Inc.
1 ackage	\$227,300		Agreement	mc.
Travel &				Transportation
Expenses while	\$1,000 - \$20,000	Lump Sum	As Incurred	Lines, Hotel
Training ⁵		_		Facilities, Etc.
Real Estate				Sublease /
three (3)	\$0,000 \$15,500	A A 1	A - T.,	
Months Initial	\$9,000 - \$15,500	As Arranged	As Incurred	DrPhoneFix,
Report ⁶				Inc.
Application	#1 000 #1 000	Y 0		Cities and
Fees, Permits	\$1,000 - \$4,000	Lump Sum	As Incurred	Municipalities
Additional				
Inventory &	\$750 - \$4,000	Lump Sum	As Incurred	Designated
Equipment	Ψ.εσ ψ.,σσσ	24p 24		Suppliers
				City Agencies,
Business /				Normally
Occupational	\$1,000 - \$4,000	Lump Sum	As Incurred	Depends Upon
License	φ1,000 φ1,000	Zamp Sam		the Number of
License				Employees
				Insurance
Insurance	\$750 - \$1,500	As Arranged	As Incurred	Agencies
Legal &				Attorney,
Accounting	\$2,000 - \$20,000	As Arranged	As Incurred	Accountant
Opening				Designated
Marketing	\$2,000 - \$3,000	As Arranged	As Incurred	Suppliers /
Program	Ψ2,000 - Ψ3,000	As Arranged	As incurred	Vendors
Marketing First				Designated
•	\$4,000 - \$6,000	As Arranged	As Incurred	•
Sixty Days	\$4,000 - \$0,000	As Arranged	As incurred	Suppliers /
Program ⁷				Vendors
Miscellaneous	¢1,000, ¢7,000	A - A - 1	A T 1	Vendors /
Costs	\$1,000 - \$5,000	As Arranged	As Incurred	Designated
				Suppliers
Additional				Employees,
Funds Six (6)	\$25,000 - \$60,000	As Arranged	As Incurred	Designated
Months 8	+ -25,000 φου,ου	110111111111111111111111111111111111111		Suppliers,
1.10110110				Vendors

Description	Estimated Amount	Method Payments	When Due	To Whom Pavable
TOTAL	\$172,500 - \$393,000			

- (1) <u>Estimated Initial Investment.</u> The table shows the estimated expenditures required for the single In-Store (Walmart) franchise.
- (2) <u>Initial Franchise Fee Will Vary</u>. If you are approved to sign an Area Development Agreement, you pay a lump sum, non-refundable Development Fee of \$15,000 for your first franchised stores, plus one-half of the progressively reduced Initial Franchise Fee for each additional store to be opened under your agreement. The initial amount paid is credited against the balance of the remaining Initial Franchise Fees to be paid by you for each store as they are subsequently opened. Development of five store locations requires a commitment of \$40,000 in franchise fees, of which \$27,500 would be paid upon signing the Development Agreement, with the balance being paid upon signing individual Franchise Agreements for each location. (See Item 5 for detailed explanation).
- (3) <u>Training</u>. Cost for obtaining certified training.
- (4) <u>Start-up Package</u>. You will pay DrPhoneFix, Inc. a Start-Up Package fee for preparing the leased space in a ready to use condition which will include furnishings, computer, POS and software, signage, equipment, and start-up inventory. Your start-up package fee will be primarily dependent upon the size of the leased premises, the amount of inventory and cabinetry you choose. The range of the Start-Up Package is set out below.

Description	Low	High
Tenant Improvement and Remodeling	\$29,000	\$68,500
Cabinetry, Shelving, Furnishings, Furniture	\$28,000	\$60,000
Computer Hardware/Software, POS and PC	\$2,500	\$4,000
Signage	\$7,000	\$15,000
Equipment	\$26,000	\$50,000
Inventory	\$10,000	\$30,000
Total	\$102,500	\$227,500

- (5) <u>Travel and Expenses While Training.</u> You are responsible for arranging and paying the expenses for transportation, meals and lodging for you and others while attending the training program. The amount you spend will depend on several factors, including the distance you travel, your method of travel and the type of accommodations you choose. We base this estimate upon attendance by three people.
- (6) Real Estate Deposits and Initial Rent. Real Estate Deposit and Rent will vary depending upon a multitude of factors related to the location and size of the leased space in a Walmart store. You will sublease the premises from DrPhoneFix, Inc. who will have an agreement for the space prior to your entering a sublease agreement. The size of the typical In-Store franchise location in a Walmart store will vary from 250 square feet to 1,000 square feet. The Techy Café requires additional space between 100 to 500 square feet.
- (7) <u>Marketing First 60 Days After Opening.</u> During the sixty (6) day period after opening, you must spend between \$4,000 to \$6,000 on Franchisor approved marketing of your store.

(8) Additional Funds. An estimate of funds to cover business (not personal) expenses during the first six (6) months of operation of your business. You will need capital to support ongoing costs of your business, such as payroll, utilities, taxes, loan payments and other expenses, to the extent that Revenues do not cover business costs. You may need additional funds during the first six (6) months of initial operation or afterwards. We do not furnish or authorize our salespersons or any other persons or entities to furnish estimates as to the capital or other reserve funds necessary to reach "break-even" or any other financial position, nor should you rely on any such estimates. The six (6) month period from beginning a business covers the time by which most Franchisees are fully operational. In addition, the estimates presented relate only to costs associated with the franchised business and do not cover any personal, "living," unrelated business or other expenses you may have. We recommend that, in addition to the amount estimated here, you have sufficient personal savings and/or income so that you will be self-sufficient for a period of time.

The estimates assume that you will lease the location of your franchise. Ranges are shown for some items because they depend upon factors that may vary such as site, location, as well as other local market conditions and circumstances. The costs listed are estimates only. All fees and payments are non-refundable, unless otherwise stated or permitted by the payee. You should review these estimates carefully with your business advisor before making a decision to enter into a Franchise Agreement.

You should review these figures carefully with a business advisor (such as an accountant) before making any decision to purchase the franchise.

YOUR ESTIMATED INITIAL INVESTMENT WRAPPED TECHY VAN OPTION ¹

	Estimated	Method		To Whom
Description	Amount	Payments	When Due	Payable
Van Purchase	\$29,000 - \$38,500	As Arranged	As Incurred	Supplier
Inventory ²	\$7,000 - \$10,000	As Arranged	As Incurred	DrPhoneFix, Inc.
Van Computer Hardware & Software ³	\$5,500 -\$8,000	As Arranged	As Incurred	DrPhoneFix, Inc.
Signage, Including Wrap ⁴	\$6,500 - \$9,500	As Arranged	As Incurred	DrPhoneFix, Inc.
Equipment & Shelving	\$7,000 - \$9,000	As Arranged	As Incurred	DrPhoneFix, Inc.
Business License & Permits	\$1,000 - \$2,500	Lump Sum	As Incurred	Governmental Agencies
Insurance ⁵	\$500 - \$1,000	As Arranged	Monthly	Vendor
Legal & Accounting	\$1,500 - \$2,000	As Arranged	As Incurred	Attorney, Accountant
Advertising & Marketing	\$2,000 - \$2,500	As Arranged	As Incurred	Vendors
TOTAL	\$60,000 - \$83,000			

(1) <u>Wrapped Techy Van.</u> You will be required to outfit the Van in accordance with our requirements set forth in your Operations Manual. You are further required to lease or purchase a Wrapped Techy Van in accordance with the specifications set forth in your

Operations Manual and additionally have the van wrapped pursuant to the requirements set forth in your Operations Manual. If you obtain our permission, you may purchase the van, your cost will vary greatly, depending upon whether you finance the Wrapped Techy Van or pay it in full at purchase. The figures under this category assume you will lease the van and incur monthly lease payments.

In a limited number of territories, we may approve the use of a Wrapped Techy Van without a physical store location in the territory. In this event, a franchisee would incur an Initial Franchise Fee of \$15,000 and Training Fee of \$7,500 paid to us, in addition to the Estimated Initial Investment for the Wrapped Techy Van Option.

- (2) <u>Inventory.</u> Your Wrapped Techy Van is required to carry inventory in addition to your store. We estimate the additional inventory cost for your Wrapped Techy Van is between \$4,000 to \$5,000.
- (3) <u>Van Computer Hardware & Software.</u> Your Wrapped Techy Van must be equipped with the required POS System and current Repair Desk System.
- (4) <u>Signage, Including Wrap.</u> Our estimate includes the required signage and wrap for your Wrapped Techy Van.
- (5) <u>Insurance</u>. You must maintain in force policies of insurance covering your mobile van using carriers we have approved covering various risks. We may specify the types and amounts of coverage required under these policies and require additional or different kinds of insurance at any time, including naming us and our Affiliates as additional insureds and excess liability insurance. All insurance will contain a waiver of subrogation rights, against us and our Affiliates and any successors and assigns, and must provide for 30 days prior written notice to us of any material modifications, cancellation, or expiration of the policies.

YOUR ESTIMATED INITIAL INVESTMENT FOR AREA DEVELOPERS

Type of Expenditure	Amount	Method of Payment	When Due	To Whom Payment Is to be Made
Area Development Fee (Note 1)	\$25,000	Two Installments	\$20,000 On Signing Development Agreement; Balance Due Upon Signing 2nd Franchise Agreements	Us
Legal and Accounting	\$2,500 - \$5,000	As Third Party Specifies	As Incurred	Attorney, Accountant
Additional Funds	\$20,000 - \$50,000	As Supplier Requires	Monthly Except Bi- Weekly Payroll	Start-Up Capital, Employees
Total (Note 2,3)	\$47,500 - \$80,000 (excluding real property)			

<u>Note 1</u>: The Area Development Fee listed above is for two (2) stores, the minimum number of stores required for an Area Development Agreement. The Area Development Fee for three stores is \$30,000 and each store location after your third location is \$5,000 per store location.

Your Development Schedule for the required signing of each Franchise Agreement and required opening date for each store will be set forth in the Area Development Agreement. You will pay an initial, lump sum, non-refundable Development Fee consisting of \$15,000 for the first franchised location, plus one-half of the progressively reduced Initial Franchise Fees (\$10,000 second store, \$5,000 third store and every additional store) for each additional franchised store to be opened under the Development Agreement.

Note 2: All fees payable to us are fully earned when paid and non-refundable. Area Development Fees are payable to us upon signing the Area Development Agreement.

Note 3: These payments are only estimates and your costs may be higher. You should review these figures carefully with a business advisor, accountant or attorney before making any decision to enter into an Area Development Agreement. We do not offer financing for your initial investment. The availability and terms of financing with third party lenders will depend upon factors such as the availability of financing generally, your credit-worthiness and policies of lending institutions concerning the type of business to be operated.

You should review these figures carefully with a business advisor (such as an accountant) before making any decision to purchase the franchise.

ITEM 8. RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

You are required to sell or offer for sale all products, items and services approved by us and not deviate from our standards and specifications by offering or selling unapproved products, items or services.

Pursuant to the Inline Franchise Agreement, you must obtain our approval for the site of your franchise store and if you are leasing the premises, your lease documentation must contain a conditional assignment and assumption of lease as required by the Franchise Agreement. We also require that your landlord give written consent to the conditional assignment and assumption of lease and our right to cure certain defaults of your lease should you not do so. You will pay our affiliate Dr. Phone Fix, Inc. a Start-Up Package fee for preparing the premises in a ready to use condition which includes furnishings, computer, POS system with software, signage, equipment and start-up inventory. The amount of your Start-Up Package fee will depend primarily upon the size of your premises, amount of inventory and cabinetry (See Item 7).

Pursuant to the In-Store (Walmart) franchise, you will enter into a Franchise Agreement for a predetermined space within a Walmart store location. You will pay our affiliate DrPhoneFix, Inc. a Start-Up Package fee for preparing the leased space in a ready to use condition which will include furnishings, computer, POS system with software, signage, equipment, and start-up inventory. The amount of your Start-Up Package fee will depend primarily upon the size of the leased premises, amount of inventory and cabinetry (See Item 7). You will also sign a sublease agreement with our affiliate for the Walmart space. The monthly rental will depend upon location and size of the premises (See Item 7).

When you enter into a sublease agreement with DrPhoneFix, Inc. they will receive an override amount which is greater than the amount DrPhoneFix, Inc. is required to pay the underlying landlord.

In some markets we may offer the Wrapped Techy Van option and you will enter into the Inline Franchise Agreement with a Wrapped Techy Van Addendum. You will pay our affiliate DrPhoneFix, Inc. a flat fee for the Wrapped Van, inventory, computer hardware and software and signage (See Item 7).

You must use products, signs, furnishings, supplies, fixtures and equipment which comply with our standards and specifications. Specifications may include minimum standards for delivery, performance, design, and appearance, and local zoning, sign and other restrictions. We formulate and modify our standards and specifications for products and services based upon the collective experience of our franchisees. Our standards and specifications are described in the Franchise Agreement, the Operations Manual, and other written documents. We have the right, under each franchise concept in the Franchise Agreement, to change the standards and specifications applicable to operation of the franchise, including standards and specifications for signs, furnishings, products, supplies, fixtures and equipment. You recognize that you may incur an increased cost to comply with these changes; however, no change will materially alter your fundamental rights under the Franchise Agreement for each concept. We will notify you of any change to our standards and specifications by way of written amendments to the Operations Manual or otherwise in writing. We have the right to require you to purchase certain products, signs, furnishings, supplies, fixtures, computer hardware and software and equipment from approved suppliers identified in the Operations Manual or otherwise in writing. We reserve the right, under each concept, in the Franchise Agreement, to require you to purchase certain items from us, our affiliate or an unaffiliated designated supplier.

We provide you with a list of approved suppliers, distributors, service vendors, suggested manufacturers for operating the franchise ("Approved Vendor List") which can be found in our Confidential Operations Manual and a list of approved inventory, products, fixtures, furniture, equipment, signs, stationery, supplies and other items or services necessary to operate your store ("Inventory List"). If you would like to sell or use any product, material or supply or purchase any products from a vendor not on either of these lists, you must notify us and may need to submit samples and other information to us so that we can make an informed decision as to whether the product or supplier meets our standards. We currently approve suppliers upon request submitted upon our "Supplier Approval Form" and payment of a supplier approval fee of \$1,000. Based upon information and samples you supply us, we will test the items supplied and review the proposed supplier's financial records, business reputation, delivery performance, credit, and other information we request from you on the supplier. We will complete our review promptly, generally within 90 days. We have the right to revoke our approval of particular products or suppliers when we determine, in our sole discretion, that such products or suppliers no longer meet our standards. Upon receipt of written notice of such revocation, you must cease purchasing products from such supplier. You must use products purchased from approved suppliers solely in connection with the operation of your store and not for any competitive business purpose. Once we approve a supplier, we have the exclusive right to negotiate price terms directly with the supplier on your behalf and on behalf of all System franchisees. We may receive rebates from approved suppliers from time to time as part of our purchasing negotiations and auditing services.

We apply the following general criteria in approving or revoking a proposed supplier:

- (1) Ability to provide product
- (2) Quality of products and services
- (3) Production and delivery capacity
- (4) Proximity to Franchised Stores in order to be able to make timely deliveries of product
- (5) Dependability of the supplier
- (6) Financial capability
- (7) Customer relations
- (8) Concentration of purchases to obtain better prices and services

We may inspect or re-inspect the facilities and products of any approved supplier and revoke our approval if the supplier fails to continue to meet our criteria and specifications. As a condition of approval, you and/or any supplier must reimburse us for all costs and expenses incurred by us associate with any testing, including travel and lodging expenses incurred where we deem it necessary to visit a supplier's facilities.

Nothing contained in this Disclosure Document or in Franchise Agreement requires us to approve an inordinate number of suppliers of a given item or approve suppliers, which, in our reasonable judgment, would result in higher costs to our franchisees or prevent, in our sole judgment, our effective and economical supervision of suppliers.

Purchasing arrangements for products, materials and supplies are in place. We will continue to negotiate purchase arrangements with suppliers including price terms for the benefit of franchisees and us. The only material benefits you will receive for using approved suppliers are lower prices and better services and quality if we are successful in our negotiations on behalf of the System.

We or our affiliates reserve the right to derive revenue from your required purchases from us or our approved suppliers. Except for the receipt by us of .5 - 20 % of the total purchase price from the sale of certain apparel and supplies, we do not currently derive any revenue or other material consideration as a result of required purchases or leases. Our affiliate, Cellular Supplies, Inc., is currently the only approved supplier of cellular phone accessories and is owned by our chief executive officer. For our In-Store (Walmart) franchise concept located at Walmart stores, you will enter into a sub-lease agreement with our affiliate, DrPhoneFix, Inc. who will also as part of the Start-Up Package prepare the leased space in a ready to use condition, which will include furnishings, computer, POS and software, equipment and start-up inventory. Our Chief Executive Officer owns an interest in DrPhoneFix, Inc. We estimate that up to 78% of your initial investment will be paid to our affiliate for the Start-Up Package, and miscellaneous supplies. We estimate that your required purchases and leases will account for approximately 85% - 95% of all purchases and leases necessary to establish the franchised business, and approximately 70% - 80% of all purchases and leases necessary to operate the franchised business after opening. You will not receive any material benefit from your use of designated or approved suppliers. Other than as described above, there are currently no purchasing or distribution cooperatives in existence for our franchise system. We reserve a right to collect rebates, although there are no arrangements in place. We expect to derive revenue from leases of software and from sales of products. We have no history of such sales. As of the date of this Disclosure Document, we are not an approved supplier of any product or service. Revenues, if any, derived from the sale of products through suppliers, whether in the form of rebates, commissions, license fees, or similar payments or similar payments will also be retained by us. During the year 2022 we had \$1,420,376 total revenue. Our affiliate Cellular Supplies Inc. had revenue of \$453,973 from franchisee purchases and our affiliate DrPhoneFix, Inc. had \$102,374 from franchisee purchases and \$1,647,654 received from rental payments. We had no revenue derived from franchisee purchases and no dollars from vendor rebates.

Trademarked Products

In the future we may develop certain Trademarked Products ("Trademarked Products") that are uniquely developed for us, which bear our logo and private label. When we do so, you must use or carry an adequate supply or representative inventory of these Trademarked Products. You will purchase Trademarked Products only from us or other designated sources which manufacture the Trademarked Products to our precise specifications.

Software:

We may develop special software programs such as accounting, inventory control, or POS, which will be proprietary to us and which contain confidential information. We cannot estimate the cost you will incur to purchase or lease the software. We have no trademarked software at this time. Currently, we require you to use "LightSpeed" as your point of sale software and "EmailMeForms" which we pay for, as a utility program. Both of these programs are subject to improvement and upgrade expenses. We require you to pay the monthly fees to us which we forward in full to the software provider.

Insurance:

You must obtain and maintain, at your expense, insurance protecting you, us and our officers, directors and employees against any loss, liability, personal injury, death or property damage or expense concerning operating your franchised store as we may reasonably require. We must be named as an additional insured in each policy or policies.

The liability insurance may not be limited in any way by reason of any insurance that we may maintain. Within 90 days of signing the Franchise Agreement, and before you acquire interest in the real property on which you will develop and operate your franchise, you must furnish us with an Accord Form Certificate of Insurance as proof of complying with our insurance requirements. The certificate must state that you will provide us with a 30-day written notice prior to any change in your policies. The certificate will reflect proof of payment of premiums. Maintenance of the insurance and your performance of the obligations under the insurance provisions of the Franchise Agreement will not relieve you of our indemnification requirements set out in the Franchise Agreement. We may modify the minimum insurance requirements and notify you of the changes in writing.

You are also required to obtain: worker's compensation with limits in compliance with your state law and employer's liability insurance, with coverage of One Million Dollars (\$1,000,000) combined single limit, as well as such other insurance as may be required by statute or rule of the state in which the franchised business is located or operated.

Additionally you must obtain comprehensive general liability insurance and product liability insurance with limits of \$1,000,000) combined single limit including the following coverages: broad form contractual liability, personal injury (employee and contractual inclusion deleted); insuring us and you against all claims suits, obligations, liabilities and damages, including

attorneys' fees, for actual or alleged personal injuries or property damage relating to the franchised business, provided that the required amounts herein may be modified from time to time by us to reflect inflation or future experience with claims; automobile liability insurance on company vehicles, including owned, hired and non-owned vehicle coverage, with a combined single limit of at least One Million Dollars (\$1,000,000); and loss of income insurance (in an amount sufficient to cover the Marketing and Development Fund fees due under the Franchise Agreement, for a period of at least six months); rental value insurance (in an amount sufficient to cover the rents and other fees due the landlord and/or Merchants' Association under the lease, if any, during any period of business interruption or inability to operate the store) or such greater amounts of insurance as required by the lease for the store; additional insurance and types of coverage as may be required by the terms of any lease for the franchised business, or as may be required by us, including an umbrella policy with limits of One Million Dollars (\$1,000,000).

All insurance policies must be issued by carriers rated AA or better by Alfred M. Best and Company, Inc. who are authorized to do business in the state where the store is located, must contain the types and minimum amounts of coverage, exclusions and maximum deductibles as we prescribe from time to time, must name us as additional insured, must provide for 30 days prior written notice to us of any material modification, cancellation or expiration of such policy and must include all other provisions we may require from time to time.

Insurance policies shall indemnify both the Franchisee (as named insured), the Franchisor and any other party having an insurable interest in either your operations or our operations as an Additional Insured from an actual or alleged claim by a third party caused by or occurring in conjunction with the operation of the franchise or otherwise in conjunction with the conduct of business by you pursuant to the Franchise Agreement. We reserve the right to adjust the limits of indemnification (up or down) or to require you to procure and maintain other additional coverage prescribed from time to time by us and issued by insurance carriers rated AA or better by Alfred M. Best and Company, Inc. We may increase the minimum liability protection requirements annually and require different or additional kinds of insurance to reflect inflation, changes in standards of liability or higher damage awards in public, product or motor vehicle liability litigation, or other relevant change in circumstances.

You shall procure and maintain at your sole cost and expense a policy of Commercial Fire Insurance on a Special Form Basis providing for One Hundred Percent (100%) replacement cost of your franchised store, Personal Property, Leasehold Improvements and Betterments and Real Property (including signs and plate glass). You shall also provide Business Interruption and Extra Expense on a One Hundred Percent (100%) co-insurance basis. We shall be named as loss Payee on the Business Interruption coverage on a primary basis so as to enable you to provide us with an uninterrupted stream of payments (as if you were still in operation) for a period of a minimum of six months.

You shall further carry any additional insurance covering such additional risks or providing higher limits as we may reasonably request.

You shall submit to us annually a copy of the certificate of, or other evidence of, the renewal or extension of each insurance policy or policies. We reserve the right to reject any policy with exclusions or sub limits that are not satisfactory to us.

If you at any time fail or refuse to maintain in effect any insurance coverage required by us, or to furnish satisfactory evidence thereof, we at our option and in addition to its other rights

and remedies hereunder, may, but shall not be required to, obtain such insurance coverage on behalf of you, and you shall promptly execute any applications or other forms or instruments required to obtain any such insurance and pay to us, on demand, any costs and premiums incurred by us plus an administrative fee of 15% of that cost (See Items 7 and 11).

Records and Reports:

You must retain for 3 years all books and records related to your franchised store, including daily sales reports, invoices, payroll records, check stubs, sales tax records and return, cash Revenues and disbursement journals, general ledgers and any other financial records that Franchisor designate or by law. You must submit to us before 10th of each quarter, P&L and Balance Sheet reports for the preceding quarter.

Advertising by Franchisee

You may not engage in any advertising program or use any other advertising, including local advertising placed on television, print or any other media, or prepare or use any marketing materials, unless we have approved them in writing. All references to "advertising" shall include, but shall not be limited to, electronic advertising, whether by internet, social media platforms or otherwise. You may not make any use of any "advertising" without our express written permission as described in Item 11 of this document. You are required to spend \$3,000 for advertising your Opening Day and for a Inline Franchise, and between \$3,000 to \$5,000 for advertising your store during the first sixty (60) days after opening.

Computer Equipment

You must purchase or lease business computing, faxing and printing equipment and software as described in Items 7 and 11.

ITEM 9. FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the franchise and other agreements. It will help you find more detailed information about your obligations in these agreements and in other items of this disclosure document.

	Section in Franchise Agreement	Item in Disclosure
Obligation	and Area Development Agreement	Document
a. Site selection and acquisition /	3.1-3.2 of Franchise Agreement;	5, 7, 11
lease	Exhibit E-1 In-Store (Walmart)	
	Addendum to Franchise Agreement;	
	Exhibit E-2 Start-Up Package	
	Addendum to Franchise Agreement;	
	Exhibit E-3 Techy Café Addendum to	
	Franchise Agreement;	
	Exhibit E-4 Wrapped Techy Van	
	Addendum to Franchise Agreement;	

Obligation	Section in Franchise Agreement and Area Development Agreement	Item in Disclosure Document
- Constant	4.1 of Area Development Agreement	
	("ADA");	
	Exhibit F-1 In-Store (Walmart)	
	Addendum to Area Development	
	Agreement	
b. Pre-opening purchases/leases	3.1, 3.2, 3.4, 4, 10.4, 10.5, and 10.6 of	5, 6, 7, 8
	Franchise Agreement;	
	Exhibit E-1 In-Store (Walmart)	
	Addendum to Franchise Agreement;	
	Exhibit E-2 Start-Up Package	
	Addendum to Franchise Agreement;	
	Exhibit E-3 Techy Café Addendum to	
	Franchise Agreement;	
	Exhibit E-4 Wrapped Techy Van	
	Addendum to Franchise Agreement;	
	1.1, 4.1 of Area Development	
	Agreement;	
	Exhibit F-1 In-Store (Walmart)	
	Addendum to Area Development	
	Agreement	
c. Site development and other	3.1, 3.2, 3.4 and 3.5 of Franchise	5, 7
pre-opening requirements	Agreement;	
	Exhibit E-1 In-Store (Walmart)	
	Addendum to Franchise Agreement;	
	Exhibit E-2 Start-Up Package	
	Addendum to Franchise Agreement;	
	Exhibit E-3 Techy Café Addendum to	
	Franchise Agreement;	
	Exhibit E-4 Wrapped Techy Van	
	Addendum to Franchise Agreement;	
	4.1 of Area Development Agreement;	
	Exhibit F-1 In-Store (Walmart)	
	Addendum to Area Development	
	Agreement	
d. Initial and ongoing training	5.1 - 5.2, and 10.5 of Franchise	5, 11
	Agreement;	
	6.1 of Area Development Agreement	
e. Opening	3.1, 3.6 and 3.7 of Franchise	6, 7
	Agreement	
	4.1 of Area Development Agreement	
f. Fees	5.2, 5.3, 9.1 - 9.2, 9.4, 9.7, 10.1, 10.8,	4, 5, 7
	10.9, 11.1, 11.3 C, 14.3 (9), and 15.3	
	H of Franchise Agreement;	

Obligation	Section in Franchise Agreement and Area Development Agreement	Item in Disclosure Document
	5,1,5.2 of the Area Development Agreement	
g. Compliance with standards and policies/Operations Manual	1.1, 2.1 D., 2.3, 3.1 - 3.3, 4, 5.4, 10.1, 10.2, 10.4, 10.6, 11, 14.3, and 15.3 of Franchise Agreement; 12.1 of the Area Development Agreement	8, 11
h. Trademarks and proprietary information	6.1 - 6.4, 8.1, and 17.2 of Franchise Agreement	13
i. Restrictions on products/services offered	2.1, 2.3, 10.2 Franchise Agreement; In Store (Walmart) Addendum	8
j. Warranty and customer service requirements	None	None
k. Territorial development and sales quotas	None	12
On-going product/service purchases	4, 10.1, 10.2, 10.3 and 10.6 of Franchise Agreement; Exhibit E-1 In-Store (Walmart) Addendum to Franchise Agreement	8
m. Maintenance, appearance and remodeling requirements	10.1 of Franchise Agreement	6, 17
n. Insurance	10.6 of Franchise Agreement	6
o. Advertising	2.3, 3.7, 11 of Franchise Agreement	7, 11
p. Indemnification	7.4 of Franchise Agreement 11.1 and 13.1 of Area Development Agreement	None
q. Owner's participation/ management/staffing	5.1, 10.6 of Franchise Agreement;6.1 of the Area DevelopmentAgreement	6, 15
r. Records/Reports	12 of Franchise Agreement; 12.1 of Area Development Agreement	6
s. Modification of the Agreement	Section 19.21 Franchise Agreement; Section 13.1 Area Development Agreement	Franchise Agreement – no modifications except by written agreement, but Confidential Operations Manual and System Standards are subject to change.
t. Integration/merger clause	Section 19 Franchise Agreement; Sections 12.1 and 13.1 Area Development Agreement	Only terms of the Franchise Agreement (including Confidential Manuals, System Standards, exhibits, and addenda) are binding (subject to state law). Any

Obligation	Section in Franchise Agreement and Area Development Agreement	Item in Disclosure Document
		other promises may not be enforceable.
u. Renewal	15.2 – 15.3 of Franchise Agreement; 3.1 of the Area Development Agreement	6, 17
v. Post-Termination obligations	8.2, 17.1 - 17.4, and 18 of Franchise Agreement; Exhibit E-1 In-Store (Walmart) Addendum to Franchise Agreement; Exhibit E-2 Start-Up Package Addendum to Franchise Agreement Exhibit E-3 Techy Café Addendum to Franchise Agreement; Exhibit E-4 Wrapped Techy Van Addendum to Franchise Agreement; 7.1, 12.1 of Area Development Agreement; Exhibit F-1 In-Store (Walmart) Addendum to Area Development Agreement	17
w. Non-Competition covenants	8.2, 17.4 of Franchise Agreement;12.1 of Area DevelopmentAgreement;ADA Addendum	17
x. Dispute resolution ⁽¹⁾	19.1 - 19.8 of Franchise Agreement;12.1 of Area Development Agreement	9, 17
y. Other (Owner's Guaranty)	Exhibit 1 of the Franchise Agreement and Exhibit 1.3 of the Area Development Agreement	10

ITEM 10. FINANCING

We have no obligation to provide any financing, but we may in our sole discretion, finance up to 40% of the Inline Franchise Agreement tenant improvements, cabinetry, furnishings, furniture, and Start-Up Package cost to qualified prospective franchisees under specified terms and conditions. Our decision to finance up to 40% of the specified cost will be based, in part, on your credit-worthiness, and our then-current financing policies.

If we agree to finance a portion of the initial franchise fee, you must sign a Promissory Note when you sign the Franchise Agreement; an example of the promissory note is attached as Exhibit I to the Disclosure Document. The annual percentage rate of interest charged is 8% and the total amount must be paid in 24 equal monthly installment payments by automatic bank draft starting the 10th day of the month following signing the promissory note and continuing thereafter on the 10th day of each succeeding month.

We may require any of your direct or indirect owners, your subsidiaries or affiliates, or spouse to sign a guaranty in the form attached as Exhibit I to the Disclosure Document.

The payment of the note may be accelerated (at our option) in certain enumerated occurrences including, among other things, default in payment or other obligations of you under the Franchise Agreement or other agreements, termination or transfer of the franchise or any significant assets, a misrepresentation by you, failure to pay taxes, death, abandonment of the franchised business, failure to furnish financial information, appointment of a receiver or if, in our reasonable opinion, your financial responsibility and/or business viability becomes unsatisfactory. Default under the Promissory Note can result in a termination of the Franchise Agreement (Promissory Note, Exhibit I).

The Promissory Note may be prepaid and there is no prepayment penalty.

The Promissory Note contains no waiver of defenses or similar provisions, except that it does contain waivers of presentment, demand, notice of dishonor, notice of protest and requires you to pay all cost and expenses of collection and/or enforcement, including reasonable attorney's fees if you default on any payment and obligation under the note (Promissory Note, Exhibit I; Security Agreement, Exhibit J).

We may sell, assign, or discount any Promissory Note or other obligation arising out of the Franchise Agreement to a third party. If we sell or assign your Promissory Note, it will not affect our obligation to provide the services to you that are described in the Franchise Agreement, but the third party may be immune under the law to any defenses to payment you may have against us.

We may periodically agree with third party lenders to make financing available to qualified franchisees and we may in our sole discretion, refer you to a third party lender for financing. We have no control over whether financing will be offered to you by any third party lender. The lender is not obligated to provide financing to you or to any other franchisee that the lender finds does not meet its credit requirements and loan criteria.

We do not currently derive income from referrals or placement of financing with any third party lender. Currently, we have no arrangements with third party lenders.

We do not guarantee your obligations with third parties.

ITEM 11. FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND TRAINING

Except as listed below, we are not required to provide you with any assistance.

Pre-Opening Assistance.

Before you begin operating your franchised store:

A. We will either approve the location of your business for our Inline Franchise or provide the location for our In-Store (Walmart) Franchise and provide a territory

surrounding the site of your business where we will not place another Techy Repairs & Smart Home Installs Powered by DrPhoneFix Franchisee (Franchise Agreement Sections 2.2, 2.3 and 2.4, Exhibits 2.2, 3.3 B and E.1).

You must select a site for your Inline Franchise business and submit the location for our approval within 90 days of signing your agreement and be conducting business at the approved location within 180 days or your franchise will be terminated without any refund of fees or expenses. We do not own any real estate locations that we would lease to you and will not be responsible for locating your business site. However, we will recommend independent leasing services that you can enlist to assist you in finding a location to submit to us for approval. We will provide you with written notice of acceptance or non-acceptance of the proposed store site within two weeks after written notice to us of your choice of location and your submission to us of the proposed lease. Acceptance of the proposed location is also dependent on our written approval of the lease (Franchise Agreements Sections 3.1 A and 3.6).

Factors we consider in reviewing a proposed Inline Franchise site include the real estate characteristics, area demographics, trade area population, visibility, traffic flow, available parking, accessibility, proximity to residential and commercial settings, distance away from competition, terms of lease, and size and physical condition of building (Franchise Agreement Section 3.1).

You must obtain a business license from a state approved provider as required by the relevant governing agencies before opening. All matters related in any way to opening and operating your business are your sole responsibility, regardless of any assistance we may choose to provide. Neither we, nor any other person or entity associated with us will have any liability for any site-related or store development matter; however, we will provide general floor plan specifications for you to follow. You will engage our affiliate, DrPhoneFix, Inc. for a Start-Up Package fee to prepare the store premises in a ready to use condition which will include furnishings, computer, POS and software, signage, equipment and start-up inventory (Franchise Agreements Sections 3.3, 3.4, 3.5 and Exhibit E-1, E-2 and E-3).

- B. We will provide you and your initial store manager and one other person with the training described later in the Training portion of this section (Franchise Agreement, Section 5.2).
- C. We will furnish you with standards, specifications and other requirements for operating your franchised business (Franchise Agreement, Section 3.3).
- D. We will provide you with Inline Franchise Agreement site selection guidelines at the time you sign the franchise agreement (Franchise Agreement, Section 3.1).
- E. We will provide you with initial and any updated approved supplier and approved service provider lists upon signing the franchise agreement (Franchise Agreement, Section 10.2).

Typical Length of Time to Open Your Business

The typical length of time between the signing of the Franchise Agreement, the payment of the Initial Franchise Fee and the opening of a Franchise is approximately 90 days. Failure to be open for business within 180 days from the signing date may, at our option result in termination of your franchise. Factors affecting this length of time include locating a site, completion of training, construction, financing, permits or licenses (Franchise Agreement, Section 3.6).

Our Obligations During the Operation of Your Techy Repairs & Smart Home Installs Powered by DrPhoneFix Franchise.

During the operation of your franchise:

- A. We will loan you a copy of the Confidential Franchisee Operations Manual. A copy of the Table of Contents to the Manuals is attached to this Disclosure Document as Exhibit A; (Franchise Agreement, Section 5.4).
- B. We will provide limited guidance in the operation of your Franchised Business. We may provide this guidance electronically, in writing or telephonically, through training programs and/or on site consultations, among other methods (Franchise Agreement, Section 5.3).
- C. We will furnish advice and guidance to you with respect to your Opening Program (Franchise Agreement, Section 3.7).
- D. We will provide you a representative to assist in the opening of your franchise for a total of 7 days covering a period immediately before, during and after the opening of your franchise (Franchise Agreement, Section 5).
- E. We will review and approve all advertisements and promotional items (Franchise Agreement, Section 11).
- F. We may modify the system and you will make the required modification, including the adoption and use of new or modified trade names, trademarks, service marks or copyrighted materials, new items, new products, new equipment or techniques (Franchise Agreement, Section 6 and 10).

Your Local Advertising Requirements.

Each month you must spend the greater of \$500 or 1.5% of the previous month's Gross Revenues for local advertising and promotion of your franchised store. We may require you to submit verification of your expenditures. You must also appropriately list your phone numbers and location in white and classified directory listings and advertising. The value of discounts, coupon redemptions and/or products or services given without charge may not be used to meet your local advertising obligation under this Section.

Your advertising will be in good taste and conform to ethical and legal standards. You must submit to us, prior to dissemination, for our approval samples of all advertising and promotional materials for any media, including the Internet. You will not use any advertising or

promotional materials or programs without our prior written approval. We can require that a brief statement regarding the franchise opportunity availability of Techy Repairs & Smart Home Installs Powered by DrPhoneFix be included in advertising used by you and/or that brochures regarding purchase of franchises be displayed in your franchised store.

All use, if any, of the Internet, World Wide Web, social media platforms or other electronic media by you in connection with your franchised store will be as specified by us. We may require that any use be through us, using a designated Internet/Intranet Service Provider (which can be us or an Affiliate), and that all pages be accessed through a designated site and/or meet our specifications.

Advertising Cooperative.

As of the date of this disclosure document, we have not established any local or regional advertising cooperatives ("Co-op Advertising Region"). If we do so in the future, you must participate in the Co-op Advertising Region for the region in which your store is located. We may exempt certain franchisees from participation at our sole discretion. We will notify you in writing if you must join a Co-op Advertising Regional and the initial amount of the contributions to the Co-op Advertising Region. We will determine the area of each Co-op Advertising Region (Franchise Agreement, Section 10.3).

Each Co-op Advertising Region must comply with the then current governing cooperative agreement. A copy of the governing documents (if one has been established) is available upon request. At all meetings of the Co-op Advertising Regions, each participating franchisee is entitled to one vote per franchise that franchisee operates in the Co-op Advertising Region and we and our affiliates are entitled to one vote for each company-owned or affiliate-owned store in the Co-op Advertising Region or such other vote as may reasonably be determined by us.

We will determine your minimum contribution to the Co-op Advertising Region. However, each Co-op Advertising Region may increase the contribution by majority vote. We or our affiliate, as applicable, will contribute to the Co-op Advertising Region for each of our company or affiliate-owned Stores located in the Co-op Advertising Region on the same basis as franchisees.

Each Co-op Advertising Region will decide the use of funds available to it for advertising or marketing, subject to our written approval. Any disputes (other than pricing) arising among or between you, other franchisees, and/or the Co-Op Advertising Region may be resolved by us.

Our decisions will be final and binding on all parties. No Co-op Advertising Region may appoint, or pay from the funds collected by the Co-op Advertising Region fees or costs of, any advertising agency or buying group without prior written consent. Co-op fees will be credited to required local advertising expenditures.

Advertising Council.

We do not have an advertising council.

Computer, POS Hardware and Software.

You must purchase, use, maintain and update, at your expense, software, computer and other systems meeting our specifications. There are no contractual limitations on the frequency and cost of upgrades and updates to the systems or programs. You must comply with our thencurrent terms of use, policies and any other requirements regarding any internet/intranet sites we establish for the franchise system. We do not provide, nor do we have any obligation to provide ongoing maintenance repairs, upgrades or updates of any software, computers or any other systems.

You must obtain and maintain at your own expense accounting, sales reporting and records retention systems conforming to any requirements set by us. We reserve the right to use, and to have full access to, all cash register, Point Of Sale, computer and any other systems, and the financial information and data they contain. We may charge a reasonable fee for the license, modification, maintenance or support of software or any other goods and/or services that we furnish to you in related to any of the systems.

Your Start-Up Package will include the Point of Sale equipment which is Just Repair Desk and includes required software.

You must maintain your POS system online to allow us access to system data and information. Supplier and/or licensor charges for use, maintenance, support and/or updates of and to the required systems are payable by you upon receipt.

Neither we nor any of the Franchisor-Related Persons/Entities will have any liability and/or obligation about any failures, errors or any other occurrences relating to any computer or system hardware or software without an express written warranty from us, even if recommended or specified by us.

You must comply with the requirements and standard processes to maintain accurate and up to date accounting records. You will provide monthly and other reports as determined by us, including updating master file records to comply with changes in accounting practices. As with all computer and internet information we have complete access to all of the information at all times.

Your Start-Up Package will also include a fax machine, printer, scanner and copier which normally is all inclusive like the Brother MFC 9700 fax/printer/scanner/copier machine.

Alarm and Surveillance Systems.

Alarm System

You are required to install an alarm system approved by us. The keypad should be close to the employee entry door.

Surveillance System

We require you to install a multi-camera surveillance system in your store as a deterrent to theft, documentation of incidents and monitoring your business. Your system should stream live to your smart phone to monitor operations.

It is best, and more cost effective, to have all of these systems purchased, pre-wired and ready for installation during initial construction or remodeling. Please refer to Item 7 for costs and other information on these systems.

Our Access to Your Systems.

In order to maintain high quality standards we will reserve the right to audit and monitor each franchised location at any time through physical or electronic access to all information of your computer systems, POS, surveillance and website systems and to all information referred to in this Item 11 and we reserve the right to use this information and data at our discretion, for our benefit and for the benefit of the franchise system as a whole. We have no plans to do so, but may in the future require you to purchase different or other software or hardware. In the event we require changes, you will receive written notification and be given a reasonable time in which to purchase the required software or hardware. We will have independent access to the information generated and stored in the systems, and there are no contractual limitation on our right to access any of the above information.

Confidential Operating Manual.

We will loan you one copy of our confidential operations manual ("Manual") which contains mandatory and suggested specifications, standards and operation procedures as we prescribe and may also include information relative to your Franchise Agreement. The Manual may be amended or modified from time to time to reflect changes in our System. You must keep the Manual confidential and current, and may not copy any part of the Manual. The table of contents for the Manual is listed in Exhibit "A". There are 110 pages in the Manual.

Training.

We may charge a reasonable fee for training of additional and/or subsequent managers. If in our judgment any trainee fails to successfully complete any part of the training program the trainee must repeat that portion of the training program until successful completion of all portions, at your expense, including the payment to us of \$400 per day of such additional training.

Our initial training schedule is set out below and will be divided between our Headquarters location in Fort Lauderdale, Florida, a facility operated by an affiliate in the Fort Lauderdale, Florida area. Training will be conducted by Timothy Phelps, Eleimar Lemus and Michael Alfonso. Mr. Phelps is our director of training and will oversee training. His work history is set out in Item 2. Eleimar Lemus is our Tech Trainer and has been with us since September 2020. She has previously worked as a technician and trainer (2018 – 2020) and managed a wireless store and implemented organizational systems and training guides for GoKrazyWireless in Cooper City, FL (2017-2018). Michael Alfonso joined our organization in 2022 as our Lead Technician/Trainer/District Manager. From 2021 to 2022, Mr. Alfonso served as Lead Technician and Manager for iFixandRepair in Plantation, Florida. Prior to this, Mr. Alfonso was Leach Technician and Manager at DrPhoneFix Coral Springs in Coral Springs, Florida. And from 2018 to 2019, Mr. Alfonso was Operations Manager at Pet Supermarket Warehouse in Sunrise, Florida.

We provide mandatory initial training prior to opening. We do not have a formal training schedule but will provide training at a mutually agreeable time prior to the date you open for business. You or your approved manager must successfully complete our training program before

operating the Techy Store. Your Store must be open within 180 days of executing the Franchise Agreement. Training consists of 72 hours of classroom training at our headquarters or an affiliate store location in Fort Lauderdale, Florida and 40 hours of on-the-job training at an affiliate store location in the Fort Lauderdale, Florida area. You and your manager must attend additional and/or refresher training programs as we may reasonably require to correct, improve and/or enhance your operations, the System and its members. In addition, we can require successful completion of training by all of your supervisory personnel. You will be responsible for all travel, living, incidental and other expenses for you and your personnel attending the Certification Training Program, repeated sessions and any other voluntary or mandatory training programs, seminars or meetings.

TRAINING SCHEDULE (CLASSROOM)

	TRAINING SCHEDO	CLASSROOM	
DAY	CIDIECT		LOCATION
	SUBJECT	(Hours)	LOCATION
1	9:30 AM-12:30 PM: Introduction to	8	Headquarters
	Techy Repairs & Smart Home Installs		Fort Lauderdale, FL
	Powered by DrPhoneFix: Email set up,		
	Account set up, Website overview.		
	12:30-1:00 PM: Lunch		
	1:00 PM-6:00 PM: Introduction to Training		
2	9:30 AM-12:30 PM: Introduction to	8	Headquarters
2	smartphone repair.	8	Fort Lauderdale, FL
	smartphone repair.		Tort Lauderdaie, TL
	12:30-1:00 PM: Lunch		
	1:00 PM-6:00 PM: Introduction to Apple		
	Products		
3	9:30 AM-12:30 PM: iPhone Repair	8	Headquarters
	Training	Ü	Fort Lauderdale, FL
			1 011 2400010410, 12
	12:30-1:00 PM: Lunch		
	12:30 PM-6:00 PM: iPhone Repair		
	Training		
4	9:30 AM-12:30 PM: iPhone Repair	8	Headquarters
	Training		Fort Lauderdale, FL
			2 of Landordaio, 1 L
	12:30-1:00 PM: Lunch		
	12.00 1.00 1.11. 201011		
	1:00 PM-6:00 PM: iPhone Repair		
	Training		
5	9:30 AM-12:30 PM: iTouch Training	8	Headquarters
	6		Fort Lauderdale, FL
	12:30-1:00 PM: Lunch		ŕ
U			

		CLASSROOM	
DAY	SUBJECT	(Hours)	LOCATION
	1:00 PM-6:00 PM: iPad/iTouch Repair		
	Training		
6	9:30 AM-12:30 PM: iTouch Training	8	Headquarters
			Fort Lauderdale, FL
	12:30-1:00 PM: Lunch		
	1:00 PM-6:00 PM: iTouch/Intro to iPad		
7	Repair Training 9:30 AM-12:30 PM: iPad Repair	8	Affiliate Store
,	Training	в	Location
	Truming		Fort Lauderdale, FL
	12:30-1:00 PM: Lunch		1 011 2000010010, 12
	1:00 PM-6:00 PM: iPad Repair Training		
8	9:30 AM-12:30 PM: HTC Repair	8	Affiliate Store
	Training		Location
	12 20 1 00 70 4		Fort Lauderdale, FL
	12:30-1:00 PM: Lunch		
	1.00 DM C.00 DM Common Demain		
	1:00 PM-6:00 PM: Samsung Repair		
	Training/Recap		
9	9:30 AM-12:30 PM: Sales, Customer	8	Affiliate Store
	Service, POS, & Inventory Training		Location Fort Londondolo El
	12:30-1:00 PM: Lunch		Fort Lauderdale, FL
	12.00 1.00 1.111 20.1011		
	1:00 PM-6:00 PM: Recap		
	Total Hours	72	

TRAINING SCHEDULE (ON THE JOB)

	TRAINING SCHEDULE		
DAY	SUBJECT	ON THE JOB TRAINING	LOCATION
1	9:30 AM-12:30 PM: Sales, Customer	8	Affiliate Store
	Service, POS, & Inventory Training		Location
			Fort Lauderdale, FL
	12:30 PM-1:00 PM: Lunch		
	1:00 PM-6:00 PM: Recap & Store		
	Acclimation And Setup		
2	9:30 AM-12:30 PM: Sales, Customer	0	Affiliate Store
	Service, POS, & Inventory Training	8	Location
			Fort Lauderdale, FL
	12:30 PM-1:00 PM: Lunch		
	1.00 DM < 00 DM D		
	1:00 PM-6:00 PM: Recap & Store		
3	Acclimation And Setup	0	A CC'1'
3	9:30 AM-12:30 PM: Sales, Customer	8	Affiliate Store
	Service, POS, & Inventory Training		Location
	12:30 PM-1:00 PM: Lunch		Fort Lauderdale, FL
	12:50 PM-1:00 PM: Lunch		
	1:00 PM-6:00 PM: Recap & Store		
	Acclimation And Setup		
4	9:30 AM-12:30 PM: Sales, Customer		Affiliate Store
	Service, POS, & Inventory Training	8	Location
	,		Fort Lauderdale, FL
	12:30 PM-1:00 PM: Lunch		·
	1:00 PM-6:00 PM: Recap & Store		
	Acclimation And Setup		
5	9:30 AM-12:30 PM: Sales, Customer	0	Affiliate Store
	Service, POS, & Inventory Training	8	Location
	12 20 DV 1 00 DV 1		Fort Lauderdale, FL
	12:30 PM-1:00 PM: Lunch		
	1:00 PM-6:00 PM: Recap & Store		
	Acclimation And Setup		
	Accimiation And Scrup		
	Total Hours	40	

We may eliminate or shorten training for persons previously trained or with comparable experience, however, any manager that has not successfully completed our Certification Training Program must pass the then-current version of our training exam.

If we offer you the opportunity to open a Techy Café at your franchised location you will receive an additional day of initial training in Customer Service, POS, and Inventory Training for the Techy Café.

Marketing and Technology Fund

We will maintain a Marketing and Technology Fund into which all franchisees will contribute 2% of Gross Revenues for the previous month or \$785 whichever is greater. Several services are supported by this Fund and the amount of your fee may reasonably be increased up to 10% per year in response to rising costs. Monthly payments to the Marketing and Technology Fund will be made through our electronic funds transfer system described in Item 6.

If we have company-owned stores in the future they will not contribute to the fund. Any stores owned jointly or separately by any of our officers or directors will not contribute to the fund. The fund will be administered by us and annual unaudited financial statements prepared at the end of each year, accounting for the placement of all Marketing and Technology Fund fees. Upon your written request we will provide to you an annual unaudited financial statement of the Fund for the previous year. We will not receive any payment or fee for providing services to the fund. Surplus funds at the end of any year will be carried over into the following year. Some advertising media will be generated by us and other items may be produced by selected agencies.

The Marketing Fund may be used to meet all costs of administering (including personnel and salaries), directing, preparing, placing and paying for international, national, regional or local advertising to promote and enhance the image, identity or patronage of Stores owned by us or our affiliates and by licensees.

We will determine how and where the money will be spent at our sole discretion whether local, regional or national. We are not required to spend any amount on advertising in your area or territory. Marketing will not be designated for the benefit of any specific franchisee, but for the benefit of the system as a whole as we may determine from time to time. The Marketing and Technology Fund supports our Website which also contains each franchisee's individual Location Website. The Website is a key part of our marketing "tool chest" for the entire franchise system. Its major purpose is to establish and promote the Techy Repairs & Smart Home Installs Powered by DrPhoneFix brand that will attract retail business for our franchisees and to provide an internet presence for each franchisee.

We will benefit from contacts by new franchisee candidates through this website, and our franchise opportunity information and application pages will be present within the site. As with all other advertising materials, stationery and business cards, the site will also contain references to the fact that "Franchises Are Available" and that "Each" Techy Repairs & Smart Home Installs Powered by DrPhoneFix® Franchise Is Independently Owned and Operated." We do not directly use any funds from the Marketing and Advertising Fund to solicit new franchise sales.

The Website contains all of the basic information about the franchise system and the high quality items and services offered by our Techy Repairs & Smart Home Installs Powered by DrPhoneFix franchises.

For the fiscal year ending December 31, 2022, advertising cost amounted to approximately \$49,447. The purposes and percent of National Advertising Funds contributed by franchisees in 2022 were:

Area of Advertising	<u>Amount</u>	<u>Percentage</u>
Production	\$ 4,945	10%
Media Placement	\$ 42,030	85%
Administrative Expenses	\$ 2,472	5%

Services Provided Pursuant to the Technology and Marketing Fund Include:

Hosting

Full Hosting and management services for your individual Location Website will be provided to give your franchise an internet presence, including your business address and telephone information, customer testimonials, photographs showing your personnel and store and other information about your local franchise. You do not have any responsibility or expense for designing, hosting, modifying, managing or operating this website presence.

Web Site

You may utilize only the website and Internet presence provided by us. You may not use any other form of website or other Internet advertising;

Franchise Location Finder

Site visitors will find your franchised store in the Location Finder portions of the Website; and your Media Manager Subscription and Password will provide you access to downloadable print and digital marketing materials and programs administered and managed by us (newspaper, magazine and direct mail advertising materials), and you will have the ability to order television and radio advertisements and bulk advertising materials, as may be made available by us from time to time. There is no cost for downloadable materials, but you must pay the cost of video duplication and for the tagging, printing and shipping of all ordered materials.

If any automatic electronic funds withdrawal is denied and remains unpaid for a period of seven days, your subscription password account will be deactivated. We reserve the right to make reasonable increases in the Marketing and Technology Fund Fee if we deem it necessary to do so because of increased costs to us.

We may develop proprietary software or alternate sources to provide your Location Website hosting and/or other services. If so, we or our designee shall license the software to you and we may need to make a reasonable increase in the Marketing and Technology Fund Fee as a result. You agree to pay the increased fee required by us and you will comply with all specifications and standards as prescribed by us from time to time in the Operations Manual.

We will also provide you information on our approved Advertising and Graphic Design Provider. In addition to the materials approved by us pursuant to the Marketing and Technology Fund, you may request the Provider to create advertising materials on your behalf. If you do, you must pay the Provider directly for the total cost of any such materials and the materials must be approved by us prior to being ordered by you.

We will decide the amount and nature of all expenditures of the Marketing and Technology Fund and how and where all expenditures are made. In accordance with Section 16 of the Franchise Agreement services provided pursuant to this fund may be deactivated upon a franchisee's failure to pay the monthly fee in a timely manner and will be reinstated only when payments and penalties are brought current.

Gift Card Program

If we develop a program for all franchisees to sell or otherwise issue gift cards or coupon certificates (together "Gift Cards") you must participate by offering Gift Cards to your customers and honoring all Gift Cards presented to you as payment for products, regardless of whether the Gift Card was issued by you or another franchised store. You shall sell, issue, and redeem (without any offset against any License Fees) Gift Cards in accordance with procedures and policies specified by us in the Manual or otherwise in writing.

We do not maintain any other Advertising or Marketing Fund or any Advertising Council or Cooperative of any kind.

Referrals

If you refer someone to us who is not already in our system, then we will provide you a referral fee of \$10,000 if that person purchases a franchise from us. To be eligible, you must have received our signed acknowledgement form. You are then entitled to the referral fee when payment in full is received for the initial franchise fee from the person that you refer to us. We may cancel or modify this referral policy at any time.

Existing Franchisees may receive a prospective franchisee conversion fee of \$500 if they answer questions of prospective candidates who become franchisees. We do not control the content of any communications between existing franchisees and prospective franchisees, and existing franchisees do not act as our agents or representatives in any manner. We may modify or cancel this conversion fee policy at any time.

Techy Shop Program

If we sell any item from our internet sites or National Website [TechyShopp.com] and the customer resides in your Protected Territory, we will pay you a commission of twenty-five percent (25%) of the gross revenue received from the sale.

System Standards

We may evaluate your franchise (including inspections, field service visits, customer comments/surveys and secret shopper reports) for compliance with our System Standards using the same methodology and scoring system as we use to evaluate any franchised store owned and/or operated by us and/or our Affiliates. Your franchised store will receive a System Standards Score for categories being scored at that time. We will compare your scores with the average score in each category as achieved by other franchised stores (which may include company-owned outlets) in other geographic areas as we believe appropriate for evaluation purposes.

ITEM 12. TERRITORY

We reserve the absolute right to adjust the territorial boundaries as we determine to be necessary in our sole discretion. The boundaries of your Protected Territory may also depend upon topographical features which clearly define a contiguous area, like rivers, lakes, mountains, major freeways, etc.

The In-Store (Walmart) franchise is for a specific location. The Inline franchise (which may include the Techy Café and Wrapped Techy Van) is for a Protected Territory with the Store location within the Protected Territory to be approved by us.

We will grant you a geographic area in which to operate your franchised store (the "Protected Territory"). The typical territory granted to each Inline (which may include the Techy Café and/or Wrapped Techy Van) and In-Store (Walmart) franchisee is an area with a 1 mile radius from the approved site or having a daytime population of at least 50,000 (US Census Bureau) people, whichever is the lesser area; however in more rural areas your Protected Territory may consist of a larger boundary and in a more densely populated metropolitan area (particularly in midcity districts) we will negotiate with you the size of your "Protected Territory." Depending on the characteristics of your approved location, one of the above classifications will apply.

You are required to select a proposed site for the Inline Franchise within the designated Site Selection Area within 90 days from the date of signing the Franchise Agreement (the Site Selection Period). You must submit a proposed site location within the first 45 days of the Site Selection Period. We will notify you of our acceptance or rejection of any proposed Site within 30 days of our receipt of all information we request concerning the site. However, if you have submitted a proposed Site to us later than the 45 day period, or for which we require further information from you, and we have not notified you of our acceptance or rejection before expiration of the Site Selection period, we will have 30 additional days to notify you of our decision.

If you are in good standing under the Franchise Agreement, you may relocate your Store within the existing Protected Territory and with our prior written consent. You will be responsible for all relocation costs and expenses. If your Store is damaged, condemned or otherwise rendered unusable, or if there is a change in character of the location of the Store sufficiently detrimental to its business potential to warrant relocation, you agree to relocate your Store within the Protected Territory.

The Protected Territory for the Inline Franchise will be designated by us once the store location is accepted by us and we have approved of your proposed lease, and will be described in Exhibit 2.2 to the Franchise Agreement. For the In-Store (Walmart) Franchise your location will be within a Walmart store and you will enter into a sub-lease agreement with our affiliate, DrPhoneFix, Inc. Your sublease is expressly made subject to all obligations and conditions of the underlying lease. In the event the underlying lease is terminated or expires, irrespective of the cause or circumstance, then the Sublessor (DrPhoneFix, Inc.) has a right to immediately terminate your sublease agreement without any liability to us or to DrPhoneFix, Inc., except that you may obtain from DrPhoneFix, Inc. any unearned fees paid in advance. If your sublease expires or is terminated you are required to find an acceptable location and develop the site as a Inline Franchise open to the public within ninety (90) days of date of expiration or termination or if there is an acceptable In-Store (Walmart) franchise location available within a one (1) mile radius of your Territory, we will offer the location to you upon the same terms and conditions as a new In-Store (Walmart)

franchise without any Initial Franchise Fee. However, we have no obligation to provide you with an alternative In-Store (Walmart) location in the event your sublease was terminated or expires. (Once a proposed site has been selected and lease negotiations entered into, a copy of the lease must be provided to us for our review and approval prior to execution. Our approval of the lease indicates only that we believe that its terms fall within the acceptable criteria we have established as of the time of our approval. (Franchise Agreement – Section 3, and Exhibit 3.2B). You may not relocate your franchised store without our prior written approval.

We will not sell another franchise or open a company-owned store in your Protected Area during the term of the Franchise Agreement. We may develop, use, and license within or outside your Protected Territory other products or services involving Marks other than those used in connection with the System. You may not solicit or market outside your Protected Territory by means of (a) Direct marketing by telephone, mail, flyers or other means; or (b) Any marketing method targeting anywhere outside your Protected Territory, except for our written approval of television, radio and print advertising media that may incidentally be covered by such advertising. If a city is located partially within your Protected Territory, then you may utilize the name of the city in your marketing methods and materials.

Except by written agreement with us you do not receive the right to acquire or open additional franchised Techy Repairs & Smart Home Installs Powered by DrPhoneFix stores within your Protected Territory.

Rights We Reserve: Franchise Agreement

You will not receive an exclusive territory. You may face competition from other franchisees, from stores we own, or from other channels of distribution or competitive brands that we control. We retain (in our sole discretion) the right to:

- 1. establish, and grant to other franchisees the right to establish franchises anywhere outside the Protected Territory, on such terms and conditions as we deem appropriate (even immediately outside the border of the Protected Territory), but not within the Protected Territory of the franchised store you open under the Franchise Agreement and continue to operate under it;
- 2. operate, and grant franchises to others to operate businesses, whether inside or outside the Protected Territory, specializing in the sale of products or services, other than a Competitive Business or a franchised store, using some of the Marks and/or all of the Marks pursuant to such terms and conditions as we deem appropriate;
- 3. operate, and grant franchises to others to operate businesses or provide other services, whether inside or outside the Protected Territory, that do not use any of the Marks;
- 4. market and sell goods and services competitive with goods and services offered by franchised stores, inside and outside of the Protected Territory through channels of distribution including but not limited to small square footage outlets like an "express" store or a kiosk; stores housed within other Walmart locations or retail facilities, such as a department store, hotel, or casino, and/or direct mail operations or Internet sites (including our National Web site [TechyShopp.com] through our Techy Shop program). Other examples of non-traditional sites include but are not limited to (1) military bases; (2) public transportation facilities, airports, train stations, bus stations; (3) sports stadiums, arenas and facilities, including race tracks; (4) student

unions or other similar buildings on college or university campuses; (5) amusement and theme parks, fairs, festivals and special events; and (6) enclosed shopping centers or "closed traffic" malls (all collectively referred to herein as "Special Sites"). Special Sites will be excluded from your Protected Territory; and

5. purchase, merge, acquire, be acquired by or affiliate with an existing competitive or non-competitive franchise or non-franchise network, chain or any other business regardless of the location of that other business' facilities, and that following such activity we may operate, franchise or license those other businesses and/or facilities under any names or marks other than, while this agreement is in effect, the Marks, regardless of the location of these businesses and/or facilities, which may be within the Protected Territory or immediately outside its border. There is no minimum sales quota for maintaining your Protected Territory or other circumstance that grants us the right to modify your territory if you are otherwise in compliance with all of your agreements with us.

Area Development Agreement

We may, but are not required to, enter into an Area Development Agreement with you which provides for the development of a specified minimum number of franchised stores within a defined geographic area (the "Development Area") over a specified period of time for each franchise (the "Development Period"). An Area Development Fee for the Development Area is required, based upon the number of franchised stores to be developed. You must enter into the thencurrent Franchise Agreement for each franchised store established under the Area Development Agreement. You are not entitled to additional development rights beyond those specified in the Area Development Agreement. You are required to comply with all site selection procedures in the Franchise Agreement for each location developed by you. Each site is subject to our acceptance, which will not be unreasonably withheld. We also have the right to refuse to grant a franchise for a proposed franchised store if you do not then meet financial criteria established by us.

Area Development Agreement: Minimum Development Quota

Your Area Development Agreement will contain a Minimum Development Quota specifying the number of Techy Repairs & Smart Home Installs Powered by DrPhoneFix franchised stores you must open during each Development Period and the cumulative number of franchised stores you must have opened through the end of all the Development Periods agreed upon. Techy Repairs & Smart Home Installs Powered by DrPhoneFix franchised stores will not count towards meeting the Minimum Development Quota for any Development Period until they have been fully constructed, developed and have opened operations in accordance with their respective franchise agreements with us. We alone determine if any franchised store has "opened" for purposes of meeting the Development Schedule, any Minimum Development Quota and for any Development Period. If a franchised store is permanently closed after having been opened, you must develop and open a substitute franchised store within one year from the date of its permanent closing separate and apart from the Development Schedule.

Rights We Reserve: Area Development Agreement

You will not receive an exclusive territory. You may face competition from other franchisees, from stores we own, or from other channels of distribution or competitive brands that we control. We retain (in our sole discretion) the right to:

- 1. Establish, and grant to other franchisees the right to establish, franchised stores anywhere outside your Territory, on such terms and conditions as we deem appropriate (even immediately outside the border of your Territory, but not within the Territory specified in the Franchise Agreement of any franchised store you operate under the Area Development Agreement and continue to operate);
- 2. Operate and grant franchises to others to operate businesses, whether inside or outside your Territory, specializing in the sale of products or services, other than a Competitive Business or a franchised store, using certain of the Marks and pursuant to such terms and conditions as we deem appropriate;
- 3. Operate and grant franchises to others to operate businesses, or provide other services, whether inside or outside your Territory, that do not use any of the Marks;
- 4. Market and sell, inside and outside of the geographic area, through channels of distribution other than full service franchised stores (like Internet, e-commerce, mail order or grocery, retail or convenience stores or through special purpose sites including military bases, public transportation facilities, sport facilities (including race tracks), Walmart locations, airports, student unions or similar buildings on college or university campuses, amusement and theme parks, special events, and through other sites which are covered/closed traffic malls, all of which are designated ("Special Sites"). Special sites designated above are not granted a Protected Territory; and
- 5. Purchase, merge, acquire, be acquired by or affiliate with an existing competitive or non-competitive franchise or non-franchise network, chain or any other business regardless of the location of that other business' facilities, and that following such activity we may operate, franchise or license those other businesses and/or facilities under any names or marks other than, while the Area Development Agreement is in effect, the Marks, regardless of the location of these businesses and/or facilities, which may be within the Protected Territory or immediately outside its border.

Designated Territory: Default Under the Area Development Agreement

We have the right to terminate an Area Development Agreement if you default under its terms or under the terms of any Franchise Agreement or other agreement you have with us. If you do not achieve the Minimum Development Quota specified in the Area Development Agreement, we, in our sole control, may, (but are not required to do so):

- 1. Terminate the Area Development Agreement;
- 2. Have the right to operate (directly or through affiliates) or grant franchises for the operation of franchised store within the Designated Territory;
- 3. Grant you an extension under the Development Schedule for such time period as we specify for a non-refundable extension fee of from \$5,000 to \$20,000 depending upon the amount of time extended; or
- 4. Reduce the Designated Territory and the Development Schedule to a size and magnitude that we estimate you are capable of operating in accordance with the Area Development Agreement.

We reserve the right to establish other channels of distribution for our services. These channels may compete with franchised businesses. We will not compensate you for any sales made through these channels.

ITEM 13. TRADEMARKS

Under the Franchise Agreement, we grant you the non-exclusive right to use our primary proprietary Mark, "Techy Repairs & Smart Home Installs Powered by DrPhoneFix [®]" and our other Marks with our written authorization, in connection with your franchise for your use and only in the manner authorized and permitted by us. The Area Development Agreement does not grant you the right to use the proprietary Mark or System. You may not directly or indirectly contest our rights in the Mark. We received an assignment of all ownership rights to the Marks from DrPhoneFix Franchises, Inc.

Our marks have been registered with the United States Patent and Trademark Office ("USPTO") on the Principal Register as follows:

1.)



Design Plus Words, Letters, and/or Numbers

Registration Number: 6416189 Registration Date: July 13, 2021

2.)



Design Plus Words, Letters, and/or Numbers

Registration Number: 6092382 Registration Date: June 30, 2020

3.) DrPhoneFix

Standard Character Mark Registration Number: 4647373 Registration Date: December 2, 2014 4.)



Design Plus Words, Letters, and/or Numbers

Registration Number: 4639249

Registration Date: November 18, 2014

There are no currently effective material determinations of the USPTO, the Trademark Trial and Appeals Board, the Trademark Administrator of any state or any court relating to the Mark. There is no pending infringement, opposition or cancellation involving the Mark; no known superior rights or infringing uses actually known to us that could materially affect your use of our Mark; and no pending material litigation involving the Mark. All affidavits and renewals required to be filed have been filed.

There are no currently effective agreements that significantly limit our rights to use or license the use of the Marks listed in this section in a manner material to the franchise to be sold. On September 1, 2021, we entered into a non exclusive license agreement with our affiliate, Techy Company LLC. In accordance with the agreement, our affiliate will be permitted to use our Marks to enable franchisees under franchise agreement acquired by Techy Company, LLC from Experimax Franchising LLC to operate those franchises.

The franchise agreement grants you the non-exclusive right to use our current and future Marks to identify the products and services offered by us. We have the right to require you to modify or discontinue your use of any Mark(s). If we exercise this right, we will provide advance notice to all franchisees. We will have no liability or obligation for your modification or discontinuance of any Mark or promotion of a substitute trademark, service mark or trade dress.

You must follow our rules when using our Marks. You must receive our approval when choosing your corporate name and you cannot use the Marks as part of the corporate or other legal entity name or with modifying words, designs or symbols without our consent. All of your usage of the Marks and any goodwill you establish is to our exclusive benefit and you retain no right in the Mark on termination or expiration of the franchise agreement. You must also obtain fictitious or assumed name registrations as we require, or under applicable law.

Neither the franchise agreement nor area development agreement contain any provisions under which we are required to defend or indemnify you against any claims of infringement or unfair competition arising out of your use of the Marks. You must immediately notify us of any apparent infringement of or challenge to your use of any Mark or claim by any person of any rights to any Mark. We will have the sole discretion to take any action we deem appropriate and will have the right to control exclusively, any litigation or USPTO or other administrative proceedings arising out of any infringement, challenge or claim or otherwise relating to any Mark. You must execute any and all instruments and documents, provide assistance and do such things as, in the opinion our counsel, may be necessary or advisable to protect our interest in any litigation or USPTO or other proceeding or otherwise to protect our interest in the Mark(s). We have no obligation under the franchise agreement or area development agreement to protect you against or reimburse you for

any damages for which you are held liable.

We may not be able to prevent anyone who began using the name "Techy Repairs & Smart Home Installs Powered by DrPhoneFix" or "Techy" or any variation thereof before our use of it from continuing their use of that name in the area of prior use. The name "Techy Repairs & Smart Home Installs Powered by DrPhoneFix" or "Techy" as abbreviated, may be in use by other businesses in the United States who are not our franchisees or in any way affiliated with us. You will be responsible for finding out whether the name "Techy Repairs & Smart Home Installs Powered by DrPhoneFix" or "Techy" as abbreviated, is already being used in your granted Territory. Under the Franchise Agreement you release us from any liability to you caused by any prior use of the name "Techy Repairs & Smart Home Installs Powered by DrPhoneFix" or "Techy" as abbreviated or any variation thereof by anyone else.

ITEM 14. PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

There are no patents that are material to the Franchise.

There currently are no effective determinations of the Copyright Office (Library of Congress) or any court regarding any of the copyrighted materials, although we claim common law copyrights in our manuals and other materials. There are no agreements in effect which significantly limit our right to use or license the copyrighted materials. Finally, there are no infringing uses actually known to us which could materially affect your use of the copyrighted materials in any state. No agreement requires us to protect or defend any copyrights or you in connection with any copyrights.

Confidential Operating Manual(s) and Other Materials.

During the term of the Franchise Agreement, we will loan to you at no charge our Confidential Operating Manual(s) in which we assert a copyright interest. The Confidential Operating Manual(s), its supplements, and any other materials or information designated by us is confidential. You will not provide your employees access to the Confidential Operating Manual(s) unless necessary to operate your franchised store. You and your employees will sign a Proprietary Information Non-Disclosure Agreement ("NDA") protecting our copyrights and confidential information and a Release and Assignment of Rights to Improvements Agreement agreeing to give up any rights to Improvements you make to the franchise. Anyone whom you transfer any interest in you, or to whom you transfer the franchise, must sign both agreements. You must obtain signatures of all employees on the NDA. Managers must also sign the NDA and Release. Personal supervision is not required if you are an individual, although it is recommended. You must devote full-time best efforts to the success of the franchise, at all times. Furthermore, each person must sign a personal guarantee to be personally bound by, and personally liable for the breach of, every confidentiality and non-competition provision, and must execute a "Employee Confidentiality Agreement" in the form required by the Franchise Agreement, Exhibit 8.1.

You must use your best efforts to keep confidential all provisions in the Confidential Operating Manual(s), including any supplements or amendments that we provide. You are responsible for keeping your copy of the Confidential Operating Manual(s) up-to-date. The provisions in our master copy will control any disputes that arise. You agree to comply with every provision in the Confidential Operating Manual(s) and every revision to the Confidential Operating

Manual(s) that we may make from time to time, provided such revisions do not implement new or different requirements which alter the fundamental terms and conditions of the Franchise Agreement.

We will loan you a replacement copy if you lose or misplace your copy or supplements but we may require a replacement charge of \$250. You must not photocopy any part of the Confidential Operating Manual(s) without our written consent.

In general, our proprietary information includes "Confidential Information" as defined in Article 8 of the Franchise Agreement, some of which is contained in our Manuals, and includes among other things: a) techniques, policies, procedures, information, systems, and knowledge regarding the development, marketing, operation and franchising of a franchised store; b) information regarding Trade Secrets, repair techniques and methods, and suppliers of Products and Services used in a franchised store; c) all information regarding customers, including any statistical and/or financial information and all lists. We disclose to you Confidential Information and your only interest in the Confidential Information is the right to use it under your Franchise Agreement.

You must agree to maintain the confidentiality of all the Confidential Information during and after the term of the Franchise Agreement and that you will not use any of the Confidential Information in any other business or in any manner we do not specifically authorize in writing. You also agree to fully and promptly disclose all ideas, techniques and other similar information relating to the Franchised Business that are conceived or developed by you and/or your employees. We will have an ongoing right to use, and to authorize others to use, these ideas, etc. without compensation or other obligation.

You must cause each of your employees, agents, principals, and affiliates to sign and deliver to you an agreement containing substantially the same provisions described in this Item 14 (in forms specified by us) and will deliver to us copies on request (Franchise Agreement, Section 8.1)

ITEM 15. OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

Your franchised store must be personally managed on a full-time basis by a person who has successfully completed mandatory training and met then-current standards as specified by us. We recommend that you participate in the supervision of the franchise. We may, however, agree in writing to allow someone other than an owner to act as the on-premise manager. That person must complete our training program and cannot have any interest or business relationship with any of our competition. You and your initial manager must attend our Certification Training Program.

Each of your employees, agents, principals and Affiliates must sign a form of confidentiality agreement as approved by us including but not limited to a Proprietary Information Non-Disclosure Agreement, a Continuing Guaranty of your full performance, and the "Employee Confidentiality Agreement." (Exhibits 1 and 8.1 of the Franchise Agreement). You will hire all employees of your Franchised Business and will be solely responsible for their supervision, possible termination, terms of employment and compensation and proper training.

If the Franchisee is a business entity, we do not require the manager to have an equity interest in the Franchise.

ITEM 16. RESTRICTIONS ON WHAT THE FRANCHISE MAY SELL

You must use or offer to the retail customer, only those products, services, equipment, programs and other items in the operation of your Techy Store that we designate in the Franchise Agreement, the Confidential Operations Manual, or specifically approve in writing. You are required to sell all goods or services authorized by us and to develop, maintain and operate your Techy Store in accordance with each Techy System Standard, as may be modified and supplemental by us from time to time in our business judgement, whether such changes require additional investments and/or changes by you in operations or other areas of your franchised business.

ITEM 17. RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

This table lists certain important provisions of the franchise and related agreements. You should read these provisions in the agreement attached to this disclosure document.

Provisions in Franchise Agreement

		Section in Franchise	
	Provision	Agreement	Summary
a.	Length of the franchise term	2.1; Exhibit E-1 In-Store (Walmart) Addendum to Franchise Agreement	10 years for new Franchise and 10 years or expiration date of the remaining term of the Franchise Agreement for an acquisition, in our sole discretion. 10 years for new Franchise, but the lease for your Walmart location only provides a term of 2 years with two 1 year renewals. After the Walmart, term you may be required
b.	Renewal or extension of the term	15.1	to relocate within your Protected Territory. If in good standing, you may renew for an additional term of 10 years.
c.	Requirements for you to renew or extend	15.1 – 15.5	Renewal of your Franchise Agreement means that upon expiration of your initial 10 year term, you may apply for a successor franchise for one additional 10 year period provided: you must give written notice in first 90 days of 9 th year; possess or renew your lease; agree to any requested

	Provision	Section in Franchise Agreement	Summary
	TTOVISION	Agreement	modifications; be in compliance with all agreements with us; execute then current agreement; sign release of all claims; attend any require refresher training at your expense; pay fee of 20% of then current franchise fee for new franchisees. The then-current agreement may be materially different than the original franchise agreement.
d.	Termination by you	None	None; mandatory notice of claims by you.
e.	Termination by us without cause	16.4	Any default under any agreement between us or any Franchisor-Related Persons/Entities and you or any owner or affiliate may be regarded as a default under the Franchise Agreement.
f.	Termination by us with cause	16.1 – 16.4	We can terminate your Franchise if you commit any one of several listed violations. Any default under any agreement between us or any Franchisor-Related Persons/Entities and you or any owner or affiliate may be regarded as a default under the Franchise Agreement.
g.	"Cause" defined-curable defaults	16.2. 16.4	Failure to: submit any information due; make any payments due us, any affiliate, supplier or creditor; meet obligations to others; comply with Manuals and all agreements; maintain required insurance; correct dangerous conditions; comply with dispute resolution procedures; cure default under (sub) lease or Operating Agreement. Any default under any agreement between us or any Franchisor-Related Persons/Entities and you or any owner or affiliate may be regarded as a default under the Franchise Agreement.
h.	"Cause" defined-non-curable defaults	16.1, 16.4	Fail to meet site selection, development, opening and other requirements, abandon or fail to operate your franchised store for more than 7 consecutive days, lose

		Section in Franchise	
	Provision	Agreement	Summary
i.	Your obligations on termination/non-renewal	17.1 – 17.4	right to possession of Premises and fail to relocate, misrepresentations or omissions by you in application for Franchise; bankruptcy; conviction of felony or other misconduct that adversely affects the franchise; unauthorized transfer or surrender of control; unauthorized use of Confidential Information, the Marks or Manuals; violation of non-compete provisions; commit any fraud or misrepresentation to us or any third party; 5 or more material customer complaints in any 12-month period. Failure to maintain records or cooperate in any audit or inspection. Any default under any agreement between us or any Franchisor-Related Persons/Entities and you or any owner or affiliate may be regarded as a default under the Franchise Agreement. Cease operations; cease use of Marks; de-identify business; pay amounts owed; return Manuals and materials; return of signage; transfer phone numbers to us or our designee; cancel all fictitious business name registrations. Certain obligations continue, including payment, indemnity, non-
j.	Assignment of contract by us	14.1	No restrictions on our right to transfer; our obligations are terminated upon transfer.
k.	"Transfer" by you-definition	14.2	Requires our written approval.
1.	Our approval of transfer by you	14.2-14.4	Transfer subject to our prior written consent.
m.	Conditions for our approval of transfer	14.2-14.4	You must be in compliance with all agreements and the Manuals; You must have continuously operated the Franchised Business for at least 5 years; Transferee must qualify; Transferee assumes your

	Provision	Section in Franchise	Summony
	Provision	Agreement	Summary
			obligations; all amounts and reports due us are remitted; all debts
			payable to third parties are assumed
			by transferee; all required reports or
			other documents submitted;
			Transferee pays a transfer fee of
			\$10,000; Transferee pays a \$7,500
			training fee and satisfactorily
			completes training; Transferee
			obtains all required permits,
			licenses and insurance; Lessors and
			other parties consent to transfer;
			execution of Franchise Agreement
			by transferee and payment the
			transfer fee; execution of release;
			amount financed subordinate to
			obligations of transferee to us;
			execution of non-competition
			agreement; and compliance with
			laws and regulations. Additional requirements apply for transfer to a
			controlled business entity.
n.	Our right of first refusal to acquire	14.7	We have the right to match offer,
111.	your business	14.7	require terms and conditions.
	Jour dustriess		require terms and conditions.
о.	Our option to purchase your	Exhibit 3.2 B	We may take over your lease rights.
	business		
p.	Your death or disability	14.5	Transfer to third party within 6
			months. Subject to all transfer
			requirements and restrictions.
q.	Non-competition covenants	8.2	No involvement in any Similar
-1'	during the term		Business.
r.	Non-competition covenants after	8.2, 14.2, 17.4, 19.6	No involvement in Similar Business
	the franchise is terminated or		for 2 years within a 10 mile radius.
	expires		
s.	Modification of the agreement	19.1, 19.8	Agreement may be modified in
			writing by all parties. Manuals and
			System are subject to change by us
			and you must promptly comply.
t.	Integration/merger clause	21	Only the terms of the Franchise
			Agreement are binding (subject to
			State law). Any representations or
			promises outside the disclosure
			document and franchise agreement
			may not be enforceable.

		Section in Franchise	G.
	Provision	Agreement	Summary
u.	Dispute resolution by arbitration	19.1 – 19.11	Except for a few types of claims, all
	or mediation		disputes are resolved through face-
			to-face meeting, mediation, and/or
			binding arbitration at a neutral site
			in the county in which our then
			current headquarters is located;
			limited rights of appeal and pre-trial
			discovery; and limitation of types of
			damages.
v.	Choice of forum	19.2	A site in the county in which our
			then current headquarters is located
			(subject to state law).
w.	Choice of law	19.8	Laws of the State of Florida apply,
			but Federal Arbitration Act
			preempts (subject to state law).

Provisions in Area Development Agreement

Pr	rovision	Section in Area Development	Summary
		Agreement	
a.	Length of the franchise term	2.1	Term is variable depending on number of franchised stores to be developed. Expires on the earlier of: (i) the latest Required Date for Opening of a Development store; or the actual Opening Date for the last scheduled franchised store.
b.	Renewal or extension of the term	None	None
c.	Requirements for you to renew or extend	None	None
d.	Termination by you	None	None
e.	Termination by us without cause	7.1 B	Any default under any agreement, lease or other obligation relating to the Franchise Business may be regarded as a default under the Area Development Agreement.
f.	Termination by us with cause	7.1, 7.1 B	We can terminate your Franchise if you commit any one of the violations listed below. Any default under any agreement, lease or other obligation relating to the Franchise Business may be

Provision		Section in Area	Summary
		Development	·
		Agreement	
			regarded as a default under the
			Area Development Agreement.
g.	"Cause" defined-curable defaults	7.1 B	Any default under any agreement,
			lease or other obligation relating
			to the Franchise Business may be
			regarded as a default under the
			Area Development Agreement.
h.	"Cause" defined-non-curable defaults	7.1, 7.1 B, 12.1	Unauthorized transfer; failure to
			meet Development Schedule; you
			commit to a lease without our
			approval, or begin the
			development and/or operation of
			a Development store, without
			having complied with the terms
			of the Area Development
			Agreement; bankruptcy (or any
			similar type of dissolution such as
			insolvency, receivership,
			assignment for the benefit of
			creditors, 30 day unsatisfied
			judgment, foreclosure on lien); conviction of felony or offense
			which affects the good will
			associated with the Marks; any
			breach of the Agreement; default
			under any agreement, lease or
			other obligation relating to the
			Franchise Business may be
			regarded as a default under the
			Area Development Agreement.
i.	Your obligations on termination/non-	12.1, 13.1	Your obligations on termination
	renewal	, ,	or non-renewal of the Area
			Development Agreement include:
			no further rights to develop
			franchised stores, cease operating
			as an area developer, cancel
			assumed names, pay amounts
			owed, assign telephone numbers
			to us and comply with any other
			provisions that survive
			termination.

Provision		Section in Area Development Agreement	Summary	
j.	Assignment of contract by us	8.1	No restrictions on our right to transfer; our obligations are terminated upon transfer.	
k.	"Transfer" by you-definition		Transfer, sale, assignment, gift encumbrance, devise, bequest of franchise, Area Development Agreement or any interest in the area development business or in you, the area developer.	
1.	Our approval of transfer by you	8.1	Transfer subject to our prior written consent.	
m.	Conditions for our approval of transfer	8.1	Payment of transfer fee. We may apply applicable transfer provisions of the most current store Franchise Agreement between us. Interest in all Development Stores must be transferred to the same assignee. For a transfer to a third party, the transferee must meet our qualifications and you must also transfer all franchise agreements for franchised stores in the developed territory.	
n.	Our right of first refusal to acquire your business	8.1, 12.1	We may apply applicable right of first refusal provisions of the most current store Franchise Agreement between us.	
о.	Our option to purchase your business	8.1, 12.1	We may apply applicable right of purchase of the most current store Franchise Agreement between us.	
p.	Your death or disability	None	None	
q.	Non-competition covenants during the term	12.1	Subject to the provisions of Franchise Agreement.	
r.	Non-competition covenants after the franchise is terminated or expires	12.1	Subject to the provisions of Franchise Agreement.	
S.	Modification of the agreement	4.1, 13.1	Agreement may be modified in writing by all parties.	
t.	Integration/merger clause	13.1	Only the terms of the Franchise Agreement are binding (assuming they comply with State law). Any other promises are not enforceable.	
u.	Dispute resolution by arbitration or mediation	12.1	Except for a few types of claims, all disputes are resolved through	

Provision	Section in Area Development Agreement	Summary
		mediation, and/or binding arbitration at a neutral site in the county in which our then current headquarters is located; limited rights of appeal and pre-trial discovery; and limitation of types of damages.
v. Choice of forum	12.1	Face-to-face meeting, mediation, arbitration and litigation at a neutral site in the county in which our then current headquarters is located. Please see the statespecific addenda to the Disclosure Document in Exhibit H and the Franchise Agreement.
w. Choice of law	12.1	Laws of the State of Florida apply, but Federal Arbitration Act preempts.

ITEM 18. ARRANGEMENTS WITH PUBLIC FIGURES

We do not use any public figure to promote our franchise.

ITEM 19. FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a Franchisor to provide information about the actual or potential financial performance of its Franchised and/or Franchisor owned outlets, if there is a reasonable basis for the information and if the information is included in this disclosure document. Financial performance information that differs from that included in item 19 may be given only if: (1) a Franchisor provides the actual records of an existing outlet you are considering buying; or (2) a Franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

We do not make any representations about a franchisee's future financial performance or the past financial performance of our affiliate-owned or franchise outlet. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting the Franchise Administration Department at 3000 South West 4th Avenue, Fort Lauderdale, FL 33315, the Federal Trade Commission, and the appropriate state regulatory agencies.

ITEM 20. OUTLETS AND FRANCHISEE INFORMATION

Table 1Systemwide Outlet Summary
For Year 2020, 2021 and 2022

Outlet Type	Year	Outlets at the Start of Year	Outlets at the End of Year	Net Change
	2020	29	33	+4
Franchise	2021	33	37	+4
	2022	37	46	+9
C	2020	5	3	-2
Company- Owned*	2021	3	8	+5
Owned*	2022	8	10	+2
	2020	34	36	+2
Total Outlets	2021	36	45	+9
	2022	45	56	+11

* There are no company stores but our affiliate, DrPhoneFix, Inc., operates ten locations and previously entered into two (2) licenses to operate a business using our trademarked name outlined in Exhibit "B".

Table 2
Transfers of Outlets
From Franchisees to New Owners
(Other than the Franchisor)
For Years 2020, 2021 and 2022

		Number of			Number of
State	Year	Transfers	State	Year	Transfers
	2020	3		2020	1
Florida	2021	5	Pennsylvania	2021	0
	2022	6		2022	0
				2020	4
			TOTALS	2021	5
				2022	6

Table 3Status of Franchised Outlets
For Year 2020, 2021 and 2022

State	Year	Outlets at Start of Year	Outlets Opened	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of Year
	2020	3	0	3	0	0	0	0
Colorado	2021	0	0	0	0	0	0	0
<u> </u>	2022	0	0	0	0	0	0	0
	2020	0	0	0	0	0	0	0
Delaware	2021	0	0	0	0	0	0	0
	2022	0	2	1	0	0	0	1
	2020	22	10	0	0	0	2	30
Florida	2021	30	6	0	0	2	0	34
<u> </u>	2022	34	9	4	0	5	0	34
	2020	1	0	0	0	1	0	0
Georgia	2021	0	0	0	0	0	0	0
	2022	0	5	0	0	0	0	5
	2020	0	0	0	0	0	0	0
Maryland	2021	0	0	0	0	0	0	0
	2022	0	1	0	0	0	0	1
	2020	0	0	0	0	0	0	0
Minnesota	2021	0	0	0	0	0	0	0
	2022	0	1	0	0	0	0	1
	2020	1	0	0	0	0	0	1
Ohio	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2020	1	0	0	0	0	0	1
Pennsylvania	2021	1	0	0	0	0	0	1
	2022	1	1	2	0	0	0	0
	2020	0	0	0	0	0	0	0
Tennessee	2021	0	0	0	0	0	0	0
	2022	0	1	0	0	0	0	1
	2020	1	0	0	0	0	0	1
Texas	2021	1	0	0	0	0	0	1
	2022	1	1	0	0	0	0	2
	2020	29	10	3	0	1	2	33
Total*	2021	33	6	0	0	2	0	37
	2022	37	21	7	0	5	0	46

Table 4
Status of Company-Owned Outlets*
For Year 2020, 2021 and 2022

State	Year	Outlets at Start of Year	Outlets Opened	Outlets Reacquired From Franchisee	Outlets Closed	Outlets Sold to Franchisee	Outlets At End Of The Year
	2020	5	3	0	1	5	2
Florida	2021	2	3	2	0	1	6
	2022	6	0	5	0	1	10
	2020	0	0	1	0	0	1
Georgia	2021	1	0	0	0	0	1
	2022	1	0	0	0	1	0
	2020	0	0	0	0	0	0
Maryland	2021	0	1	0	0	0	1
	2022	1	0	0	0	1	0
	2020	5	3	1	1	5	3
Total*	2021	3	4	2	0	1	8
	2022	8	0	4	0	3	10

* There are no company stores but our affiliate, DrPhoneFix, Inc., operates ten locations and previously entered into two (2) licenses to operate a business using our trademarked name outlined in Exhibit "B".

Table 5Projected Openings
As of December 31, 2022

State	Franchise Agreements Signed But Outlet Not Opened	Projected New Franchised Outlets in the Next Fiscal Year	Projected New Company-Owned Outlets in the Next Fiscal Year
California	1	0	0
Florida	3	5	0
Georgia	3	1	0
Indiana	1	0	0
Ohio	1	0	0
TOTAL	9	6	0

Exhibit "B" lists the name of our current franchisees and the addresses and telephone number of each of their outlets as of date of issuance.

Exhibit "C" lists the name, city and state, and current business telephone number, or if unknown, the last known home telephone number for each franchisee who had an outlet terminated,

canceled, not renewed, or otherwise voluntarily or involuntarily ceased to do business under the franchise agreement during the most recently completed fiscal year or who did not communicate with us within 10 weeks of the disclosure document issuance date.

If you buy this franchise, your contact information may be disclosed to other buyers if you leave the franchise system. No franchisees signed any confidentiality clauses during the last three fiscal years.

There are no trademark specific franchisee organizations associated with the franchise system being offered.

We have not signed any confidentiality clauses with any current or former franchisees in the last three years that would restrict their ability to speak openly about their experience with us.

ITEM 21. FINANCIAL STATEMENTS

Attached as Exhibit "D" is our Audited Financial Statement for the period ending December 31, 2022, December 31, 2021 and December 31, 2020.

ITEM 22. CONTRACTS

The following agreements are attached as exhibits to this disclosure document:

Exhibit E	Franchise Agreement
Exhibit E-1	In-Store (Walmart) Addendum to Franchise Agreement
Exhibit E-2	Start-Up Package Addendum to Franchise Agreement
Exhibit E-3	Techy Café Addendum to Franchise Agreement
Exhibit E-4	Wrapped Techy Van Addendum to Franchise Agreement
Exhibit F	Area Development Agreement
Exhibit F-1	In-Store (Walmart) Addendum to Area Development Agreement

ITEM 23. RECEIPTS

See Exhibit "M" for detachable receipts.

EXHIBIT "A" TO THE DISCLOSURE DOCUMENT

CONFIDENTIAL OPERATING MANUAL TABLE OF CONTENTS

$T \ A \ B \ L \ E \quad O \ F \quad C \ O \ N \ T \ E \ N \ T \ S$

<u>Chapter</u>	<u>Page</u>
Chapter 1 – Finding and Leasing a Location	2
Chapter 2 – Accounting and Insurance	13
Chapter 3 – Store Design and Layout	22
Chapter 4 – Décor	30
Chapter 5 – Signage	36
Chapter 6 – Construction and Equipment	40
Chapter 7 – Preparing to Open	50
Chapter 8 – Personnel and Training	58
Exhibits to Chapter 8	77
Chapter 9 – Services, Products, Purchasing	79
Exhibits to Chapter 9 – Inventory and Pricing Guide	86
Chapter 10 – Paperwork and Controls	89
Chapter 12 – Evaluations and Compliance	104

Total number of Pages: 110

EXHIBIT "B" TO THE DISCLOSURE DOCUMENT

LIST OF FRANCHISEES	

List of Current Franchises As of December 31, 2022

LOCATION	FRANCHISEE	ADDRESS	PHONE		
		Delaware			
Techy Wilmington	CELLBIZ LLC	1251 Centerville Road Wilmington, DE 19808	(302) 691-9288		
Florida					
Techy Boca Raton	Decaro International Corp.	9858 Glades Rd d-1 Boca Raton, FL 33434	(561) 756-8427		
Techy Boynton	Cibele Dorigo	3200 Old Boynton Rd Boynton Beach, FL 33436	(561) 244-2377		
Techy Cape Coral	The Glass Labs Recycling LLC	1619 Del Prado Blvd Cape Coral FL 33990	(239) 672-9399		
Techy Clermont	Techy Central Florida LLC	1450 Johns Lake Rd Clermont, FL 33403	(352) 717-8742		
Techy Cooper City	G & N Phone Repair	8787 Stirling Road, Suite B4 Cooper City, FL 33328	(954) 368-6830		
Techy Coral Springs	IW Florida Business LLC	3801 Turtle Creek Dr. Coral Springs, FL 33067	(954) 392-9633		
Techy Davie	Patterson Income Holdings LLC	13658 W State Road 84 Davie, FL, 33325	(416) 998-7560		
Techy Weston	TECH-F4U LLC	4442 Weston Rd. Davie, FL 33331	(786) 973-3886		
Techy Delray	IW Investment LLC	14851 Lyons Rd. #120 Delray Beach, FL 33446	(561) 908-0719		
Techy Las Olas	JLP Systems LLC	1261 E Las Olas Blvd Fort Lauderdale, FL 33301	(561) 315-4010		
Techy North Fort Myers	JL Phones LLC	545 Pine Island Rd Fort Myers, FL 33903	(239) 672-9399		
Techy Fort Meyers	Luajo LLC	14821 Ben C Pratt/6 Mile Cypress Pkwy Fort Myers, FL 33912	(239) 596-0518		
Techy Jupiter	Playback Productions	660 US Hwy One North Jupiter FL 33469	(561) 725-9194		
Techy Hialeah Gardens	Hoyos Brother Corp.	18518 NW 67 Avenue Hialeah, FL 33015	(305) 905-1521		
Techy Miami Lakes	KYS Services LLC	9300 NW 77 th Ave. Hialeah, FL 33016	(305) 818-9566		
Techy Kissimmee	BETA TRO CORP	904 Cypress Pkwy. Kissimmee, FL 34759	(407) 749-6866		
Techy Lake Park	Vegon Services LLC	101 N Congress Ave Lake Park, FL 33403	(561) 370-7478		

LOCATION	FRANCHISEE	ADDRESS	PHONE
Techy Marco Island	Novic Investments	6650 Collier Blvd. Naples, FL 34114	(239) 513-1452
Techy Margate	WI Florida Business LLC	5555 W Atlantic Blvd Margate, FL 33063	(954) 975-8682
Techy Wynwood	Chance308 LLC	3110 N Miami Ave. Miami, FL 33127	(786) 431-2197
Techy Tamiami	NBE2020 LLC	11461 SW 40th Street Miami, FL 33165	(305) 553-2500
Techy Doral	Honest Abe Florida LLC	1850 NW 117 th Place, Suite 302 Miami, FL 33182	(305) 666-2660
Techy Miramar	KYS Services LLC	16263 Miramar Pkwy Miramar, FL 33027	(954) 392-5520
Techy Naples South	Gallorth LLC	9885 Collier Blvd Naples, FL 34114	(239) 352-0053
Techy Aventura	NIBECA TECH, LLC	18557-B West Dixie Highway North Miami Beach, FL 33180	(303) 356-4005
Techy Ocoee Walmart	BUCAM LLC	10500 W Colonial Dr. Ocoee, FL 34761	(407) 789-2887
Techy Pembroke Pines Walmart	Zaks Telecom LP	12800 Pines Blvd. Pembroke Pines, FL 33331	(954) 681-0579
Techy Sunrise	Flagus LLC	8359 West Sunrise Blvd. Plantation, FL 33322	(954) 446-9011
Techy Port Charlotte	Florida Free Phones	19100 Murdock Cir. Port Charlotte, FL 33948	(941) 257-5505
Techy Port St. Lucie	Chaconia Holdings, LLC	1675 NW St. Lucie Blvd. Port St Lucie, FL 34986	(772) 905-4384
Techy Port St. Lucie South Walmart	Chaconia Holdings LLC	10855 S U.S. Hwy #1 Port St. Lucie, FL 34952	(772) 905-4387
Techy Tallahassee	LifeLine Repairs, Inc.	1005 N. Monroe Street Tallahassee, FL 32303	(850) 210-2517
Techy Tavares	Lake Drpfx Inc.	438 E Burleigh Blvd Tavares, FL 32778	(352) 462-1577
Techy West Palm	Vegon Services LLC	4225 45 th St. West Palm Beach, FL 33407	(561) 246-3349
		Georgia	
Techy Athens	TECHMATE LLC	1911 Epps Bridge Pkwy Athens, GA 30606	(706) 608-4061
Techy Atlanta	Lifeline Repairs, Incorporated	2 Briarcliff Road Atlanta, GA 30329	(404) 343-2252

LOCATION	FRANCHISEE	ADDRESS	PHONE		
Techy Ashford Dunwoody	Lifeline Repairs, Incorporated	4505 Ashford Dunwoody Road NE #10 Atlanta, GA 30346	(404) 989-4588		
Techy Peachtree	LifeLine Repairs, Inc.	1007 Georgian Park Peachtree City, GA 30269	(770) 487-3222		
Techy Norcross	Techmate Norcross LLC	4975 Jimmy Carter Blvd Norcross, GA 30093	(770) 282-8980		
	N	Maryland			
Techy Frederick	VolGun LLC	2480 Osprey Way S, Unit B Frederick MD, 21701	(240) 422-8330		
	N	Minnesota			
Techy Waterville	Redwin Ventures LLC	11 Elm Plz. Waterville, MN 04901	(240) 422-8330		
		Ohio			
Techy Lancaster	DPF Columbus LLC	1622 N Memorial Dr. Lancaster, OH 43130	(740) 688-3504		
	7	Tennessee			
Techy Nashville	LifeLine Repairs, Inc.	4109 B Hillsboro Pike Nashville, TN 37215	(615) 712-8915		
	Texas				
Techy Plano	KKMG LLC	425 Coit Rd. Plano, Texas 75075	(972) 905-9645		
Techy Woodlands	Hyotem Integrated Services USA LLC	10001 Woodlands Pkwy Woodlands, TX 77382	(281) 843-3457		

Note:

* There are no Area Developers.

List of Franchises Sold But Not Open As of December 31, 2022

LOCATION	FRANCHISEE	ADDRESS	PHONE
California			
Techy Chula Vista	Robert Perez	875 E H St Chula Vista, CA 91910	(619) 494-2725
Florida			
Techy Parkland	Andres Del Toro	6059 Coral Ridge Dr. Coral Springs, FL 33076	(954) 507-0510
Techy Hilaleah Café	T & H Franchising LLC	3311 W 80 th St, Unit 10A Hialeah, FL 33018	(305) 547-8668
Techy Leesburg	Techy Central Florida LLC	2501 Citrus Blvd Leesburg, FL 34748	(352) 504-0545

Georgia				
Chamblee, GA	Techmate Chamblee	1871 Chamblee Tucker Rd	(470) 460 6062	
Chamblee, GA	LLC	Chamblee, GA 30341	(470) 460-6062	
Techy Lilburn	Dharti Niril Patel	4004 Lawrenceville Highway	(770) 999-0886	
Techy Libum	Dilatti Nitti Fatei	Lilburn, GA 30047	(770) 999-0880	
Techy Stone	Dharti Niril Patel	1825 Rockbridge Road SW	(770) 501-7848	
Mountain	Dianti Iviini I dici	Stone Mountain, GA 30087	(770) 301-70-0	
	Ind	iana		
Techy Noblesville	CEEBEE Investments	16865 Clover Road	(317) 779-2134	
Techy Noblesville	LLC	Noblesville, IN 46060	(317) 779-2134	
	Ol	nio		
		8934 Darrow Road, Unit		
Techy Twinsburg	Techy Ohio LLC	C103	(330) 964-6858	
		Twinsburg, OH 44087		

LIST OF AFFILIATE AND LICENSEES

List of Affiliate Locations As of December 31, 2022

LOCATION	AFFILIATE	ADDRESS	PHONE
	Flor	rida	
Techy Parkland	DrPhoneFix, Inc.	6059 Coral Ridge Dr. Coral Springs, FL 33076	(954) 507-0510
Techy Deerfield	Techy, LLC	1335 S Military Trail Deerfield Beach, FL 33442	(954) 573-6929
Techy Ft. Lauderdale	DrPhoneFix, Inc.	3000 SW 4th Ave Fort Lauderdale, FL 33315	(754) 200-8917
Techy Oakland Park	DrPhoneFix, Inc.	371 E Commercial Blvd Oakland Park, FL 33334	(954) 653-2482
Techy Oldsmar	DrPhoneFix, Inc.	3801 Tampa Rd Oldsmar, FL 34677	(203) 577-8567
Techy Leesburg	DrPhoneFix, Inc.	2501 Citrus Blvd Leesburg FL, 34748	(352) 504-0545
Techy Pinellas	DrPhoneFix, Inc.	5075 Park Blvd N Pinellas Park, FL 33781	(813) 333-7254
Techy Port St. Lucie	DrPhoneFix, Inc.	10855 S U.S. Hwy 1 Port St. Lucie, FL 34952	(772) 905-4387
Techy Sarasota	Techy, LLC	4070 Cattleman Rd. Sarasota, FL, 34233	(941) 296 6980
Techy Wellington	DrPhoneFix, Inc.	11924 Forrest Hill Blvd. Suite 38A Wellington, FL 33414	(561) 909-2906

Licensees

- 1. Dr Phone Fix Plantations, FL G & N Phone Repair Corp 37 South State Road 7 Plantation Florida 33317 (954) 680-1823
- Dr Phone Fix Kendall, FL PhonoWiz, LLC 8813 SW 107th Ave Suite 110 Miami Florida 33176 (305) 275-2870

EXHIBIT "C" TO THE DISCLOSURE DOCUMENT

FRANCHISEES WHO HAVE LEFT THE SYSTEM

List of Former Franchisees

LOCATION	FRANCHISEE	ADDRESS	PHONE
	De	laware	
Techy Wilmington	PDR LLC	1251 Centerville Road Wilmington, DE 19808	(302) 691-9288
	F	lorida	
Techy Boynton	Pieter M Soeters	3200 Old Boynton Rd. Boynton Beach, FL 33436	(561) 244-2377
Techy Cape Coral	JL Phones LLC	1619 Del Prado Blvd Cape Coral FL 33990	(239) 672-9399
Techy Clermont	LBR Partners LLC	1450 Johns Lake Rd Clermont, FL 33403	(352) 717-8742
Techy Parkland	The Fix Company	6059 Coral Ridge Dr Coral Springs, FL 33076	(954) 507-0510
Techy Deerfield	PP Chuela LLC	1335 S Military Trail Deerfield Beach, FL 33442	(954) 573-6929
Techy Miami Lakes	Argin Tech Support Inc.	9300 NW 77 th Ave. Hialeah, FL 33016	(305) 818-9566
Techy Lake Park	Brindesi & Silez Investment Group LLC	101 N Congress Ave Lake Park, FL 33403	(561) 370-7478
Techy Leesburg	LBR Partners	2501 Citrus Blvd Leesburg FL, 34748	(352) 504-0545
Techy Mount Dora	Lake Drpfx Inc.	1800 Edgewater Dr Mount Dora, FL 32757	(354) 636-3226
Techy North Miami	Brimax Tech LLC	12253 Biscayne Blvd. North Miami, FL 33181	(303) 356-4005
Techy Pembroke Pines	La Manchis LLC	400 N. University Dr. Pembroke Pines, FL 33024	(786) 252-0286
Techy Pembroke Pines Walmart	Vandur	12800 Pines Blvd. Pembroke Pines, FL 33331	(954) 681-0579
Techy Pompano Beach	Saint Soul Technologies LLC	2300 W Atlantic Blvd Pompano Beach, FL 33069	(954) 960-6220
Techy Sarasota	Tedeschi LLC	4070 Cattleman Rd. Sarasota, FL, 34233	(941) 296 6980

Techy West Palm	Brindesi & Silicz Investment Group LLC	4225 45 th St. West Palm Beach, FL 33407	(561) 246-3349			
Minnesota						
Techy Waterville	Redwin Ventures LLC	11 Elm Plz. Waterville, MN 04901	(240) 422-8330			
	Pennsylvania					
Techy Pittsburg	Mohammad Norman	250 Summit Park Dr Pittsburgh, PA 15275	(412) 275-0468			
Techy Pittsburg	PDR LLC	250 Summit Park Dr Pittsburgh, PA 15275	(412) 275-0468			

EXHIBIT "D" TO THE DISCLOSURE DOCUMENT

FINANCIAL STATEMENTS

OF

TECHY, LLC

Techy, LLC

Audited Financial Statements For the year ended

December 31, 2022

Kevin Norton, P.A.

Certified Public Accounting
1451 W. Cypress Creek Road, Suite 300
Ft. Lauderdale, Florida 33309
(954) 822-1223

February 3, 2023

Techy, LLC

Independent Auditor's Report

Report on the Financial Statements

We have audited the accompanying financial statements of Techy, LLC

(A Development Stage Company) which comprise the consolidated balance sheet as of December 31, 2022 and the related consolidated statements of income, comprehensive income, changes in members' equity and cash flows for the year then ended, and the related notes to the financial statements.

Management's Responsibility for the Financial Statements

Management is responsible for the preparation and fair presentation of these financial statements in accordance with accounting principles generally accepted in the United States of America; this includes the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

Auditor's Responsibility

Our responsibility is to express an opinion on these financial statements based on our audit. We conducted our audit in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free from material misstatement.

An audit involves performing procedures to obtain audit evidence about the amounts and disclosures in the financial statements. The procedures selected depend on the auditor's judgment, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error. In making those risk assessments, the auditor considers internal control relevant to the entity's preparation and fair presentation of the financial statements in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the entity's internal control. Accordingly, we express no such opinion. An audit also includes evaluating the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluating the overall presentation of the financial statements.

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Opinion

In our opinion, the financial statements referred to above, present fairly, in all material respects, the consolidated financial position of Techy, LLC. as of December 31, 2022 and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Respectfully submitted,

Kevin J Norton

Kevin Norton, C.P.A.

Techy, LLC

TABLE of CONTENTS

Independent Auditor's Report	1
Financial Statements:	
Balance Sheet	2
Statement of Income and Members Equity	3
Statement of Cash Flows	4
Notes to Financial Statements	5

<u>TECHY, LLC</u> CONSOLIDATED STATEMENT OF INCOME AND MEMBERS EQUITY For the period ended December 31,

BEVENUE	-	2022	2021	2020
REVENUE: Franchise Revenues Royalties Other	\$	230,753 960,285 92,551	275,100 880,978 53,440	379,360 679,419 -
Total Revenues	\$	1,283,589	\$ 1,209,518	\$ 1,058,779
Cost of Sales				
Purchase Labor	\$	(461,794) (101,400)	(233,225) (101,400)	(102,010) (135,200)
	\$	(563,194)	\$ (334,625)	\$ (237,210)
Gross Profit	\$	720,395	\$ 874,893	\$ 821,569
EXPENSES:				
Advertising	\$	49,447	\$ 94,397	\$ 41,214
Auto		1,155	1,223	=
Bank & Credit Cards		11,720	7,825	3,472
Commissions - related party		61,500	15,250	6,000
Office Expenses & Other		11,096	2,804	37,569
Travel & Entertainment		1,512	7,086	5,829
Professional fees		67,059	73,393	43,918
Insurance Rent		2,500 95,607	49,983	23,619
Total Expenses	\$	301,596	\$ 251,961	\$ 161,621
Net Income from Operations	\$	418,799	\$ 622,932	\$ 659,948
Gain on Sale of Reclaimed Franchises	\$	170,000	\$ *	\$ •
Net Income	\$	588,799	\$ 622,932	\$ 659,948
MEMBERS EQUITY - beginning	\$	109,275	\$ 236,127	\$ 80,398
Net Income from Operations Prior Period Adjustment	\$	588,799 30,000	\$ 622,932	\$ 659,948
Distributions	<	(315,641)	 (749,784)	(504,219)
MEMBERS EQUITY - ending	\$	412,433	\$ 109,275	\$ 236,127

<u>TECHY, LLC</u> CONSOLIDATED STATEMENT OF CASH FLOWS For the period ended December 31,

	 2022	2021	2020
CASH FLOWS FROM OPERATING ACTIVITIES Net Income	\$ 588,799 \$	622,932 \$	659,948
Depreciation and amortization	-	-	
(Increase) decrease in Accounts Receivable (Increase) decrease in Due to Related Party Increase (decrease) in Loans Payable	- (17,279) (2,070)	300	11,779 (2,600) 142,400
Increase (decrease) in Accrued Expenses	 -	(3,000)	2,150
Cash Received from Operating Activities	 569,450	620,232	813,677
CASH FLOWS FROM FINANCING ACTIVITIES			
Distributions - Members	(315,641)	(749,784)	(504,219)
	(315,641)	(749,784)	(504,219)
CASH FLOWS FROM INVESTING ACTIVITIES Prior Period Adjustments	\$ 30,000 \$	- \$	-
	30,000		
INCREASE IN CASH	283,809	(129,552)	309,458
CASH - Beginning	243,925	373,477	64,019
CASH - Ending	\$ 527,734 \$	243,925 \$	373,477

<u>TECHY, LLC</u> CONSOLIDATED BALANCE SHEET At December 31,

ASSETS		2022	2021	2020
CURRENT ASSETS: Cash Accounts Total	\$	527,734 \$	243,925 \$	373,477
Accounts Receivable		-	-	-
Total Current Assets		527,734	243,925	373,477
FIXED ASSETS Fixed Assets - Net		-	-1	
OTHER ASSETS Security Deposits		- 24.770	7.500	7,000
Trademark		24,779 250 25,029	7,500 250 7,750	7,800 250 8,050
Total Assets	\$	552,763 \$	251,675 \$	381,527
LIABILITIES AND MEMBERS EQUITY (LOSS)				
LIABILITIES:	•	<u> </u>		
Accrued Expenses Due to Related Party	\$	- \$	- \$	3,000
Loan Payable - SBA	:	140,330	142,400	142,400
Total Liabilities		140,330	142,400	145,400
MEMBERS' EQUITY:				
Capital		-	-	-
Retained Earnings		412,433	109,275	236,127
Members Equity		412,433	109,275	236,127
Total Liabilities and Members Equity	\$	552,763 \$	251,675 \$	381,527

Techy, LLC

Notes to The Consolidated Financial Statements December 31, 2022

Note 1 – Summary of Significant Accounting Policies

Nature of Business...

Techy, LLC (The Company) is a Florida based company engaged in the business of selling franchises to the general public. The main headquarters is located in Florida.

Basis of Presentation...

The financial statements have been presented on the accrual basis of accounting and in accordance with United States generally accepted accounting principles.

Use of Estimates...

The preparation of the Company's financial statements are made in conformity with generally accepted accounting principles that require estimates and assumptions that affect revenues and expenses, assets and liabilities. Actual results could differ from those estimates, and such estimates could be material.

Revenue Recognition...

The Company recognizes revenue from various sources of revenue including service fees, contracts, advertising, franchise fees and royalties. Revenues from such sources are recognized as earned upon the completion of work performed. Initial franchise fees are recognized upon the completion of all contractual requirements, with subsequent revenues being predominantly classified as royalty fees. Additionally revenues are earned via product sales and rental income. All franchises are independent franchise outlets.

Customer Deposits...

The Company will be maintaining an account called Customer Deposits for the collection of funds from pending franchisees that are in the process of completing all pertinent requirements prior to the finalization of franchise acceptance by the Company. These funds are ultimately transferred to Revenue upon satisfactory completion of the franchise process. ASC 606 has been considered regarding revenue recognition that there is no material impact. Management has determined all responsibilities have been contractually satisfied, with no pending additional requirements.

Inventory...

The company maintains no physical inventory.

Cash...

The company maintains its cash balances at banks that participate with the Federal Deposit Insurance Corporation with guarantees insuring accounts with balances up to \$250,000.

Techy, LLC

Notes to Financial Statements December 31, 2022

Fixed Assets...

The Company will be maintaining property, plant and equipment, and depreciates them using the straight line basis over their useful lives of 5 to 28 years.

Note 2 – Provision for Federal Income Tax:

The company has elected under the Internal Revenue Service Code to be taxed as an LLC as a "pass through" entity. In lieu of this election, all shareholders will pay tax on their proportionate share of profits / losses via a K1. Therefore, no provision for income tax has been included in these financial statements.

Note 3 – Commitments and Contingencies:

The Company has no material commitments or legal or financial contingencies to note in 2022

Note 4 - Concentration:

The Company had no significant concentrations during 2020 that represented any material impact on its revenues.

Note 5 - Related Party Transactions:

The Company has no material related party transactions to report in 2022

Note 6 - Financial Instruments:

Cash, money market investments, current loans payable, accounts receivable, accounts payable and accrued liabilities are all short term in nature and as such, their carrying values approximate fair values.

EXHIBIT "E" TO THE DISCLOSURE DOCUMENT

FORM OF FRANCHISE AGREEMENT	



FRANCHISE AGREEMENT

Table of Contents

1.	INTRODUC 1.1	TION, DEFINITIONS, AND PRELIMINARY AGREEMENTSIntroduction	
2.	GRANT AN	D TERM	2
	2.1	Term	
	2.2	Grant.	2
	2.3	Performance.	2
	2.4	Rights We Reserve	2
3.	DEVELOPM	MENT AND OPENING OF YOUR STORE	
	3.1	Site Selection.	
	3.2	Lease of Premises.	
	3.3	Store Design Standards.	
	3.4	Development of Your Store.	6
	3.5	Equipment, Furniture, Fixtures and Signs.	
	3.6	Store Opening.	7
	3.7	Grand Opening Program - Marketing.	7
	3.8	Relocation of Store Premises.	7
4.	COMPUTER	R HARDWARE AND SOFTWARE SYSTEMS	7
5.	TRAINING	AND GUIDANCE	7
	5.1	Initial Orientation	
	5.2	Training	
	5.3	Guidance and Assistance.	
	5.4	Manuals	9
6.			
	6.1	Goodwill and Ownership of Marks	
	6.2	Limitations and Use of Marks	
	6.3	Notification of Infringements and Claims.	
	6.4	Discontinuance of Use of Marks	
7.		SHIP OF THE PARTIES; INDEMNIFICATION.	
	7.1 7.2	Independent Contractor	
		No Liability for Acts of Other Party.	
	7.3	Taxes	
	7.4	Responsibility, Indemnity, etc.	
	7.5	Disclosure.	
8.	CONFIDEN' 8.1	TIAL INFORMATION; EXCLUSIVE RELATIONSHIP	
	8.2	Exclusive Relationship, Restrictions on Similar Businesses During Franchi	
	0.2	Term and After Transfer, Termination, Expiration, Repurchase, etc	

9.	FEES		13
	9.1	Initial Franchise Fee, Releases, etc.	13
	9.2	Continuing Weekly Royalty Fee and Payment Dates	14
	9.3	Electronic Funds Transfer.	15
	9.4	Interest and Late Fees on Late Payments and/or Reports.	15
	9.5	Application of Payments, Set-Offs etc	15
	9.6	Inflation Adjustments	16
	9.7	Mandatory Convention Attendance, Possible Fee.	16
	9.8	Start-Up Package	16
	9.9	Discontinuance of Services	16
	9.10	Cellphone Accessory Support Fee	16
	9.11	Call Log PBX Software Fee	16
10.	YOUR STOR	E — IMAGE AND OPERATION.	
	10.1	System Compliance, Regular Upgrading	
	10.2	Designated Equipment, Products, Services and/or Suppliers	
	10.3	Purchasing Cooperative.	
	10.4	Compliance with Laws and Ethical Business Practices.	18
	10.5	Management and Personnel of Your Techy Store, Training	19
	10.6	Insurance.	20
	10.7	Program Participation.	20
	10.8	Continued Payment of Continuing License Fees, etc. During Closure	20
	10.9	Customer Satisfaction, Quality Controls, etc	20
	10.10	Coupons, Certificates and Vouchers	21
	10.11	Hours of Operation	21
11.	MARKETING	G	
	11.1	Marketing and Technology Fund.	
	11.2	Your Participation in the Marketing and Technology Fund	
	11.3	Your Local Store Marketing Activities	
	11.4	Marketing Cooperative and Groups.	24
12.		ND REPORTING.	25
	12.1	Bookkeeping, Accounting and Records, Cash Register, Computer and Other Systems	25
	12.2	Reports, Financial Statements and Tax Returns.	
13.	INSPECTION	VS AND AUDITS	
	13.1	Our Inspections, etc.	
	13.2	Audit.	26
	13.3	Gross Revenues Understatements.	26
14.	TRANSFER		26

	14.1	Transfers by Us.	26	
	14.2	Transfers by You	27	
	14.3	Conditions for Approval of Any Transfer	27	
	14.4	Additional Conditions for Transfer to a Business Entity	29	
	14.5	Death or Disability of Franchisee.	29	
	14.6	Effect of Consent to Transfer.	30	
	14.7	Our Right of First Refusal	30	
15.	SUCCESSOR 15.1	FRANCHISE. Your Right to Acquire a Successor Franchise.		
	15.2	Grant of a Successor Franchise.	31	
	15.3	Agreements/Releases.	32	
	15.4	Training and Refresher Programs.	32	
	15.5	Fees and Expenses.	32	
16.	TERMINATION	ON OF THE FRANCHISE		
	16.1	Defaults with No Right to Cure.		
	16.2	Defaults with Right to Cure.		
	16.3	Repeated Defaults.	35	
	16.4	Cross-Defaults.	35	
17.	RIGHTS AND OBLIGATIONS ON TRANSFER, REPURCHASE, TERMINATION AND/OR			
	17.1	NOF THE FRANCHISE Payments of All Amounts Owed, etc		
	17.2	Intellectual Property, Confidential Information, Trade Dress, etc		
	17.3	Telephone and Other Directory Listings, Internet Sites		
	17.4	Continuing Obligations.		
18.	GRANT OF S	ECURITY INTEREST.		
19.		OIDANCE AND RESOLUTION–MANDATORY BINDING ARBITRATIO Fundamental Business Intention To Mediate And/Or Arbitrate, Severability C Provisions, Modification Of Agreement To Comply With Applicable Law, Federal Arbitration Act Governs.	N . 37 Of	
	19.2	Face-To-Face Meeting, Mediation And Mandatory Binding Arbitration		
	19.3	Exclusive Jurisdiction and Venue.		
	19.4	Non-Retention of Funds		
	19.5	Waiver of Punitive Damages, etc	42	
	19.6	Survival of Obligations.		
	19.7	Attorneys' Fees and Costs.		
	19.8	Enforcement and Construction.		
	19.9	Our Exercise of "Sole Discretion" and Other Choices; Express Agreement	44	
	19 10	Miscellaneous	44	

	1	9.11 Intention to Arbitrate, Severability of Dispute Resolution Provisions, etc	44
20.	NOTICE	S AND PAYMENTS	45
21.	ACKNOWLEDGMENTS AND REPRESENTATIONS, ENTIRE AGREEMENT, NO FIDUCIARY RELATIONSHIP, ETC45		
22.	DEFINITIONS47		47
23.	3. SPOUSAL CONSENT		51
EXHII	BITS:		
EXHIBIT 1		OWNER'S GUARANTY AND ASSUMPTION OF OBLIGATIONS	
EXHIBIT 1.2		CURRENT FORM OF RELEASING LANGUAGE	
EXHIBIT 2.2		PROTECTED TERRITORY & SITE SELECTION	
EXHIBIT 3.2 B		COLLATERAL ASSIGNMENT OF LEASE	
EXHIBIT 3.3		SITE SELECTION ACKNOWLEDGMENT AND ADA CERTIFICATION FORM	
EXHIB	BIT 8.1	EMPLOYEE CONFIDENTIALITY AGREEMENT	
EXHIB	BIT 10	EXECUTIVE ORDER 13224 AND RELATED CERTIFICATIONS	
EXHIB	BIT 23	SPOUSAL CONSENT	



Agreement Number:	
-------------------	--

FRANCHISE AGREEMENT

Date of this Agreement:	("Agreement Date")
Expiration Date:	
Franchisor: Techy, LLC	
Franchisee:	

1. INTRODUCTION, DEFINITIONS, AND PRELIMINARY AGREEMENTS.

1.1 Introduction.

- A. We have developed methods of operating Franchised Stores that provide, at a retail level, the operation of businesses for the installation, maintenance and repair of cell phones, tablets, computers, game consoles, smart home related hardware and other electronic devices, and the sale of specified electronic devices, accessories and other supplies and items that we may designate from time to time under the trade name "Techy Repairs & Smart Home Installs Powered by DrPhoneFix" hereinafter sometimes referred to as "Techy". We refer to these businesses as "Techy Stores."
- B. To simplify this Agreement and make it easier to read and understand, we have defined certain terms used in this Agreement in Article 22. When you see a capitalized word, or if you do not understand the meaning of a particular pronoun reference, look at Article 22 to see whether the term has been defined. Capitalized words that are not defined in Article 22 are defined in the section where they first appear.
- C. You applied for a franchise to own a Techy Store and your application has been approved by us in reliance on the information you gave us.
- D. Your Techy franchise is a license arrangement, awarded under specific terms and conditions. You must comply fully with this Agreement and the Manuals in order to use the Marks, System and other Intellectual Property.
- E. You agree that it is critical to you, us and each Franchisee for the System to be flexible to respond to commercial opportunities and challenges. An inability to change the System could adversely affect all franchisees. You, therefore, agree and anticipate that the Manual and the System may be changed by us, from time to time in our Business Judgment. You agree to comply with the Manuals and the System as they are changed by us.

- F. Every detail of your Techy Franchised Business is important not only to you, but to us and to all franchisees to increase and maintain the value of the Marks and the businesses associated with them. Therefore, during the term of this Agreement, you must at all times develop, maintain and operate your Techy franchised business in accordance with each Techy System Standard, as modified and supplemented by us from time to time in our Business Judgment, and understand that such changes may require additional investments and/or changes by you in operations and other areas of your Franchised Business.
- G. Without your commitment to the System and to fulfill each of the obligations detailed in this Agreement, we would not form this franchise relationship with you.

2. GRANT AND TERM.

2.1 Term.

The term of the Franchise and this Agreement begins on the Agreement Date and expires 10 years from the Agreement Date. This Agreement may be terminated before it expires in accordance with its terms.

2.2 Grant.

Subject to the terms of and upon the conditions contained in this Agreement, we grant you a franchise (the "Franchise") to: (a) operate one Techy Store at the Site, and at no other location (temporary or permanent); (b) use the Marks solely in connection with operating the Techy Store; and (c) use the System in its operation. Except as set forth in paragraph 2.4 and its sub-paragraphs below, as long as you are in compliance with this Agreement, we will not grant a franchise to anyone else to operate, or ourselves operate, a Techy Store within the Protected Territory.

2.3 Performance.

You agree that you will at all times faithfully, honestly and diligently perform your obligations, continuously exert your best efforts to promote and enhance the Techy Store and not engage in any other business or activity that conflicts with your obligations to operate or with our written permission, for a manager who has received our training certification to operate, the Techy Store in compliance with this Agreement. You may not operate the Techy Store from any location other than the Site without our prior written consent. At all times, either you or one of your Principal Owners must meet our qualifications for responsibility and authority over the operations of the Techy Store. In addition, at all times the Techy Store must be managed by a general manager who has satisfactorily completed our Initial Training program.

2.4 Rights We Reserve.

Notwithstanding any of the foregoing, we (and our affiliates) retain the right in our sole discretion to:

- a. establish, and grant to other franchisees the right to establish, Techy Stores anywhere outside the Protected Territory, on such terms and conditions as we deem appropriate (even immediately outside the border of the Protected Territory), but not within the Protected Territory of your Techy Store you open under this Agreement and continue to operate under it;
- b. operate, and grant franchises to others to operate businesses, whether inside or outside the Protected Territory, specializing in the sale of products or provisions of services, other than a

Competitive Business or a Techy Store, using certain of the Marks and pursuant to such terms and conditions as we deem appropriate;

- c. operate, and grant franchises to others to operate businesses or provide other services, whether inside or outside the Protected Territory, that do not use any of the Marks;
- d. market and sell, inside and outside of the Protected Territory, through channels of distribution or franchise, (like mail order, Internet or Intranet, website (including our National Web site [TechyShopp.com] through our Techy Shop program) or other forms of e-commerce or grocery, retail or convenience Stores or kiosks or full service franchises or stores located within Walmart Stores), or through special purpose sites including military bases, public transportation facilities, sport facilities (including race tracks) Walmart Stores, airports, student unions or similar buildings on college or university campuses, amusement and theme parks, special events, and through other sites which are covered or closed traffic malls, all of which are designated ("Sites We Reserve"), such Sites We Reserve are not protected and are not part of your Protected Territory; and
- e. you agree that we have the right, now, or in the future, to purchase, merge, acquire, or affiliate with an existing competitive or non-competitive franchise network, chain, or any other business regardless of the location of that chain's or business' facilities, and to operate, franchise or license those businesses and/or facilities as "Techy" or "Techy Store" operating under the Marks or any proprietary marks or any of their marks following our purchase, merger, acquisition or affiliation, regardless of the location of these facilities (which you acknowledge may be within the "Protected Territory" or proximate thereto, or proximate to any of the Franchisee's locations).

If we sell any item from our National Website under our Techy Shop program and the customer resides in your Protected Territory, we will pay you a commission of twenty five percent (25%) of the gross revenue received from the sale. We may in our sole discretion discontinue this program at any time.

We will have the right to assign this agreement, and all of its rights and privileges under this agreement, to any person, firm, corporation or other entity.

You agree and affirm that we may sell ourselves, our assets, Marks or other proprietary marks and/or our System to a third party; may go public; may engage in a private placement of some or all of our securities; may merge, acquire other corporations, or be acquired by another corporation; and/or may undertake a refinancing, recapitalization, leveraged buyout or other economic or financial restructuring. With regard to any of the above sales, assignments and dispositions, you expressly and specifically waive any claims, demands or damages arising from or related to the loss of Franchisor's name(s), Marks, proprietary marks (or any variation thereof) and system and/or the loss of association with or identification of "Techy" or "Techy Store" as Franchisor under this Agreement. You specifically release any and all other claims, demands or damages arising from or related to the foregoing merger, acquisition and other business combination activities including, without limitation, any claim of divided loyalty, breach of fiduciary duty, fraud, breach of contract or breach of the implied covenant of good faith and fair dealing.

If we assign our rights in this Agreement, nothing herein shall be deemed to require us to remain in the Techy Store business or any business which we now conduct or to offer to sell any products or services to you.

3. DEVELOPMENT AND OPENING OF YOUR STORE.

3.1 Site Selection.

- A. If you have applied for a Inline Franchise you applied to own and operate one Techy Store at a location we have accepted (the "Site"). During the period ending on the 90th day following the date of this Agreement (the "Site Selection Period"), you must identify, submit to us for acceptance, and obtain our acceptance of the Site. During the Site Selection Period, we will not ourselves, nor grant a franchise to someone else to, open and operate a Techy Store at a fixed location inside the Site Selection Area. During the Site Selection Period, you must adhere to the following time schedule:
- (1) <u>Site Selection Area</u>: During the first 45 days of the Site Selection Period, you must obtain our acceptance of an intersection of streets or other landmark that will form the center (the "Center") of the Site Selection Area (the "Site Selection Area"). When the Center and Site Selection Area is determined, we will complete Exhibit "2.2" and provide a copy of it to you.
- (2) <u>Site Identification</u>: Prior to the 90th day following the Agreement Date, you must identify your proposed Site (which must be located within the Site Selection Area) and submit it to us for our acceptance. We may propose a Site to you, which you must independently investigate. You are solely responsible for your Site Selection, development and operation. If we notify you that we will not accept that proposed Site, you must, within the next 30 days of our notice rejecting that proposed Site (but prior to the expiration of the Site Selection Period), identify and submit to us an alternative proposed Site, for our review and acceptance within the Site Selection Area.
- (3) <u>Site Acceptance</u>: We are not obligated to evaluate or accept any proposed Site submitted to us for acceptance after the expiration of the Site Selection Period or outside the Site Selection Area. Once you have identified the Site and we have accepted it, and the lease has been reviewed and is acceptable to us, we will designate your Protected Territory and complete Exhibit 2.2 and provide it to you. If you have not identified and obtained our acceptance of a Site prior to signing this Agreement, you (with or without our assistance) must, within the Site Selection Period, identify, submit to us and obtain our acceptance (in our sole judgment) of a Site for your Techy Store located within the Site Selection Area. However, if as of the expiration of the Site Selection Period we have yet to notify you whether we will accept or reject a Site that was submitted to us for review during the Site Selection Period, we will have 30 days following the end of the Site Selection Period to notify you of our decision to accept or reject that proposed Site. If we do not accept a Site during that 30 day period following the Site Selection Period, we will not be obligated to permit you to submit an alternative proposed Site to us.

Acceptance by us of any site is not a recommendation, approval or endorsement of such Site. We make no representations or warranties as to the success of any Site or as to any other matter relating to the Site.

- (4) <u>Protected Territory</u>: Upon our acceptance of the Site, we will designate the "Protected Territory". After the lease has been reviewed and is acceptable to us, we will insert that Site and the Protected Territory into Exhibit "2.2" and send a copy to you.
- B. Each proposed Site must be evaluated by us or by a professional site analyst that we have approved. The Site must meet our criteria, with which we will provide you, for demographic characteristics, traffic patterns, parking, competition from and proximity to other businesses and other Techy Store, the nature of other business in proximity to the site and other commercial characteristics and the size, appearance and other physical characteristics of the proposed site. We retain the right to accept or reject the

floor plan for your site. We will accept or reject a Site you propose for a Techy Store within 30 days. We may rely entirely on the site analysis in doing so. You acknowledge and agree that:

- (1) our recommendation of the Site or acceptance of the Site, the Site Selection Area, or the Protected Territory does not imply, guaranty, assure, warrant or predict profitability or success, express or implied;
- (2) our recommendation or acceptance of the Site indicates only that we believe that the Site, the Site Selection Area and the Protected Territory fall within the acceptable demographic and other criteria for site selection areas, sites and premises or protected territories that we have established as of the time of our recommendation or acceptance of the Site, Site Selection Area or Protected Territory;
- (3) application of criteria that we have developed for our System or have appeared effective with respect to other sites and premises may not accurately reflect the potential for all sites and premises, and, after our acceptance of a site, demographic and/or other factors included in or excluded from our criteria could change to alter the potential of a site and premises; and
- (4) the uncertainty and instability of such criteria are beyond our control, and we will not be responsible for the failure of a premises, Site and Protected Territory we accept.
- C. All matters related in any way to your site are your sole responsibility, regardless of any assistance or recommendation we may choose to provide. You are responsible for obtaining any architectural and engineering services required for your facility and for ensuring its compliance with local law. Neither we, nor any Franchisor-Related Persons/Entity, nor any other person or company associated with us will have any liability for any site-related matter. You agree not to make any claims against us and/or any of the Franchisor-Related Persons/Entities with regard to such matters.
- D. If you are unable to acquire an approved site within the time provided in 3.1 (A), and are not making diligent efforts or adequate progress to do so in our Business Judgment, then we may Terminate this Agreement without refund to you of any fees paid to us or expenses incurred by you.
- E. If you applied for an In-Store (Walmart) franchise to operate one Techy Store there are certain terms and provisions of this Agreement which have been modified in the Addendum attached hereto as Exhibit "E-1" and made a part hereof.

3.2 Lease of Premises.

- A. Upon approval of your site, you agree to submit any lease and all site-related documents to us for our review and negotiation prior to execution by you. You will use commercially reasonable efforts to arrange for the inclusion of the provisions of the Lease Addendum attached hereto as Exhibit 3.2 B or other appropriate site-related documents which will:
- (1) Obligate the lessor to provide us, upon request, with sales and other operations information related to your Techy Store;
- (2) Permit you to operate your Techy Store in accordance this Agreement and the Manuals;
- (3) Provide that the Premises will be used only for the operation of a Techy Store, and prohibit you from assigning or modifying any of your lease rights, or extending the term without our prior written consent:

- (4) Require the lessor to concurrently provide us with a copy of any written notices (whether of default or otherwise) to you under the lease and give us the right to cure any default if we so choose;
- (5) Provide us with a right to take assignment and possession of your Techy Store without the lessor's consent or any additional consideration. If we exercise this right, we will not have any liability for any obligations incurred prior to our occupancy. You agree to take whatever actions are necessary to accomplish such assignment and will, when you sign this Agreement, also sign the Collateral Assignment of Lease attached as Exhibit 3.2 B. If you lose your lease rights to the Premises in connection with any bankruptcy, the lessor will, on our request, enter into a new lease with us on essentially the same terms as the terminated lease;
- (6) Provide that the lessor consents to the use of the Marks, Trade Dress and other aspects of the System, as modified from time to time, and give us the right to enter the premises during normal business hours for purposes of inspection, to take steps to protect the Marks and Trade Dress and/or prevent/cure any default.

You will not execute a lease or sublease, or any modification or amendment, without our prior written consent, which we may grant, condition or withhold in our Business Judgment. We may, in our sole Business Judgment, elect to require you to enter into either (i) the negotiated lease directly with the landlord of the Premises or with us or our designee as sub-tenant of the Premises, or (ii) an operating or license agreement with us or our designee permitting you to operate the Franchised Business on the Premises (the "Operating Agreement"), or (iii) we or our designee may assign the lease with the landlord for your Premises to you. You will deliver the relevant executed original of the signed lease, sublease, license, Operating Agreement or lease assignment, as applicable within five (5) days after it is signed.

(7) In all future negotiations for amendments, modifications or renewals of the lease, sublease or Operating Agreement for the Premises, you will notify us in writing and we reserve the right to retain an attorney to negotiate with the landlord. We have no obligation to provide any legal services or to retain an attorney to represent you if you are in breach or default of the lease, or there is an alleged violation of the lease or the Operating Agreement. You have no right to negotiation or otherwise on or with respect to a lease if you have an Operating Agreement for the Premises.

3.3 Store Design Standards.

You agree to comply with any standards, specifications and other requirements (the "Design Standards") that we furnish you for design, decoration, layout, equipment, furniture, fixtures, signs and other items for your Techy Store and if applicable, Techy Café. Any changes from plans provided by us must be submitted to us for our consent, which may be provided in our Business Judgment. Your compliance with the Design Standards does not release you from your obligations to ensure that your Techy Store and if applicable Techy Café is designed, constructed and operated in compliance with all local, state, and federal laws, including (without limitation) the Americans with Disabilities Act ("ADA"). You agree to execute and deliver to us an ADA Certification in the form attached to this Agreement as Exhibit 3.3 before you open your Techy Store to confirm and certify that your Techy Store and any proposed renovations comply with the ADA and other requirements.

3.4 Development of Your Store/Techy Café.

You must select a Start-Up Package (see Paragraph 9.8) and are solely responsible for all work performed at the Premises, even if referred by us.

3.5 Equipment, Furniture, Fixtures and Signs.

You will use only Designated Equipment and suppliers approved by us in the development and operation of your Techy Store as we may require. We and/or our Affiliates may be such approved suppliers.

3.6 Store Opening.

You will open your Techy Store for business within 180 days from signing your franchise agreement and immediately upon our notice to you that: (a) all of your pre-opening obligations have been fulfilled; (b) pre-opening training has been completed; (c) all amounts due us (and/or any Affiliate) have been paid; and (d) copies of all insurance policies (and payment of premiums), leases/subleases and other required documents have been received.

3.7 Opening Program - Marketing.

You agree to spend at least Three Thousand Dollars (\$3,000) on a opening marketing for your Techy Store. We will furnish advice and guidance to you with respect to our Opening Program that you agree to follow.

3.8 Relocation of Store Premises.

If you are in Good Standing, you may relocate your Store within your existing Territory with our prior written consent (unless we waive this provision), which we may condition or withhold in our Business Judgment, including the requirement that you sign a General Release. Relocation costs and expenses will be borne solely by you. If your Techy Store is damaged, condemned or otherwise rendered unusable, or if, there is a change in the character of the location of your Techy Store sufficiently detrimental to its business potential to warrant relocation, you agree to relocate your Techy Store.

4. COMPUTER HARDWARE AND SOFTWARE SYSTEMS.

- A. You must purchase, use, maintain and update at your expense the software, computer and other systems (including point-of-sale and back-office systems) meeting our specifications, as we may modify at our sole discretion. You agree to maintain your systems online to allow us access to system data and information. You agree to comply with our then-current Terms of Use and Privacy Policies and any other requirements regarding all computer and other systems, including Internet usage. Supplier and/or licensor charges for use, maintenance, support and/or updates of and to the required systems are payable by you upon receipt.
- B. Neither we nor any of the Franchisor-Related Persons/Entities will have any liability and/or obligation (and neither you, nor any Affiliate of yours, will make any claims) about any failures, errors or any other occurrences relating to any computer or system hardware or software without an express written warranty from us, even if recommended or specified by us.

5. TRAINING AND GUIDANCE.

5.1 Initial Orientation.

Prior to selecting and obtaining our approval for your site, we will schedule an orientation with you to provide you with materials and guidance related to selecting a site for your Techy Store, as well as leasing requirements and guidelines. We may also propose a site to you, which you must independently investigate as you are solely responsible for your selection, development and operation of the site.

5.2 Training.

- A. You must select and have approved a site for your Techy Store and if applicable, Techy Café before you begin our Techy Training Program (the "Training Program"). You or if we approve a manager of your store and if applicable, Techy Café, then your initial Techy manager must successfully complete our Training Program before operating your Techy Store. The cost of our Training Program for you and your initial Techy manager is \$7,500, which you must pay us upon execution of this Franchise Agreement. You and your initial Store manager should, but need not attend our Training Program simultaneously. We may charge a reasonable fee for training of additional and/or subsequent managers. We can choose to eliminate or shorten training for persons previously trained or with comparable experience. The Training Fee is non refundable and deemed fully earned upon execution of this Franchise Agreement.
- B. The initial Training Program will be at a time and place, and for such period, as we specify. You will be responsible for all travel, living, incidental and other expenses for you and your personnel attending the Training Program and any other voluntary or mandatory training programs, seminars or meetings, unless otherwise agreed to in advance in writing. We may charge a fee for any optional training programs.
- C. You or the manager we have approved, must complete our Training Program before operating your franchised Techy Store Business and if applicable, Techy Café. The Initial Franchise Fee covers a Training Program for you and your initial Techy manager and one other person. We may charge a reasonable fee for training of additional and/or subsequent managers. If in our judgment any trainee fails to successfully complete any part of the training program the trainee must repeat that portion of the training program until successful completion of all portions, at your expense, including the payment to us of \$400 per day of such additional training.
- D. You and your manager must attend additional and/or refresher training programs, including national and regional conferences, conventions and meetings, as we may reasonably require to correct, improve and/or enhance your operations, the System and its members. You must obtain certification from a state approved provider as required by State agencies before opening, and you must maintain such certification at all times that your Techy Store is operating and if applicable, that your Techy Café is operating. In addition, we can require successful completion of training by your personnel as specified by us from time-to-time.

5.3 Guidance and Assistance.

We will provide guidance in the operation of your Techy Store and if applicable, Techy Café or Wrapped Techy Van. This guidance can be furnished in whatever manner we consider appropriate in our Business Judgment, including electronically, in writing or telephonically, through training programs and/or on-site consultations, among other methods. We may provide at your request on-site consultations at your Techy Store and if applicable, Techy Café, based on notice, availability of personnel and your payment of reasonable travel, food, incidental and lodging expenses. We may elect to charge a reasonable fee for any such on-site consultations. If we believe in our Business Judgment that your operations warrant it, we can require that a manager or other person designated by us (and compensated by you) be placed in your Techy Store and if applicable, Techy Café to supervise its day-to-day operations until operations meet System standards.

5.4 Manuals.

During the term of the Franchise, we will loan you (or allow you electronic or other access to) one copy of the Manuals. If we advise you that all or part of the Manual or other specifications, standards and operating procedures are posted on a Website, you agree that it is your responsibility to monitor the Website for any changes, additions or deletions in the information provided. You will continuously comply, at your sole expense, with all provisions of, and additions/deletions/changes to, the Manuals. Any such additions/deletions/changes will take precedence over all prior communications. Mandatory specifications, standards and operating procedures prescribed from time to time by us in the Manuals, or otherwise communicated to you electronically or otherwise, are a part of this Agreement. In the event of a dispute, the master Manuals maintained at our office will control. The Manuals and the information and data that they contain will at all times remain our sole and exclusive property. It is your sole responsibility to establish, with respect to your employees, appropriate personnel and security-related policies and procedures (provided that we always have the right to terminate your rights by declaring a breach under this Agreement for conduct by you which threatens the goodwill associated with the Marks.) You and we acknowledge and agree that we neither dictate nor control labor or employment matters for you and your employees, including (but not limited to) hiring, firing and/or discipline of employees, nor control the manner and means by which they carry out their duties. You and we agree that neither of us are, or shall be deemed to be, a joint employer with the other and you will indemnify us with respect to any such or similar claims against us. You may not at any time copy, duplicate, record or otherwise reproduce any part of the Manual(s). If your copy of the Manuals is lost, destroyed or significantly damaged, you agree to obtain a replacement copy at our then applicable charge which is currently \$250 (unless we have made our on-line Manual(s) accessible to you. If so, you may utilize the on-line Manual(s) instead of purchasing other protected Manuals).

6. MARKS.

6.1 Goodwill and Ownership of Marks.

You have a non-exclusive right to use the Marks and only as expressly authorized by us under this Agreement. We have all rights in and to the Marks. All goodwill belongs exclusively to us, and you will not obtain any goodwill in the Marks as a result of this Agreement, your operation of the Franchise or for any other reason. Any unauthorized use of the Marks is a breach of this Agreement and an infringement of our proprietary rights. You agree that if you breach any obligation regarding the Marks, we would have no adequate remedy at law and that we will be entitled to equitable relief. You will not oppose, or engage in any acts or omissions inconsistent with, our rights in and to the Marks. This Agreement applies to all trademarks, service marks and other commercial symbols that we may authorize you to use throughout its term.

6.2 Limitations and Use of Marks.

You will use the Marks as the sole identification for your Techy Store. You will not use any Mark, or modified version or derivative of a Mark, or any other mark or form of commercial identification confusingly similar to the Marks or Trade Dress, as part of any business or trade name or in any other manner not expressly authorized by us in advance and in writing. Prior to adoption and/or use, any proposed corporate and/or trade name must be approved by us in our Business Judgment. You will give such trademark and other notices (including notices of independent ownership) as we direct and will, at your expense, obtain fictitious or assumed name registrations as may be required under law. You will display the Marks as required by us and will not use the Marks so as to negatively affect their goodwill. You will not use any Mark in connection with the performance or sale of any unauthorized services or products or at any location or in any other manner not expressly authorized in writing by us.

6.3 Notification of Infringements and Claims.

You will take such actions as we consider important in our Business Judgment to protect the Marks. You will not take any action that jeopardizes our interests in, or the validity or enforceability of, the Marks. You agree to immediately notify us of any apparent or actual infringement of, or of any challenge to your use of, the Marks. You will not communicate with any third party with respect to such a claim. We will take such action as we deem appropriate in our Business Judgment. As owner of the Marks, we have the exclusive right to control any settlement, litigation or proceeding arising out of or related to any such matters.

6.4 Discontinuance of Use of Marks.

You agree to comply at your expense with any directions from us to discontinue, modify, substitute or add Marks. We cannot and do not make any guaranty that a modification, discontinuance or otherwise may not be required for any reason. In such event, we will have no liability or obligation to you. You agree to make no claim in connection with any modification, discontinuance or other action, and/or with any dispute regarding the Marks. There is always a possibility that there might be one or more businesses using a name and/or marks similar to ours and with superior rights. We urge you to research this possibility, using telephone directories, local filings and other means, prior to signing any documents or making any payments or commitments

7. RELATIONSHIP OF THE PARTIES; INDEMNIFICATION.

7.1 Independent Contractor.

You will always identify yourself to all persons and in all dealings of your Techy Store as an independent owner under a Techy franchise, clearly indicating that your Franchised Business is separate and distinct from our business. You will include notices of independent ownership on such forms, business cards, stationery, advertising, signs and other materials as we require from time to time. Subject to the requirements of this Agreement and the Manuals, you will have complete operational control of your business, including the right to hire and fire each employee.

7.2 No Liability for Acts of Other Party.

You will not represent that your and our relationship is other than that of independent Franchisor and Franchisee. Neither you nor we will have any liability under any acts, omissions, agreements or representations made by the other that are not expressly authorized in writing.

7.3 Taxes.

Payment of all taxes related to your Franchised Business is your sole responsibility. We have no liability for any taxes on the sales made and/or business conducted by you (except for any taxes we are required by law to collect from you with respect to purchases from us).

7.4 Responsibility, Indemnity, etc.

A. You will indemnify and hold us and all of the Franchisor-Related Persons/Entities harmless from all fines, suits, proceedings, claims, demands, actions, losses, damages, costs, fees (including attorneys' fees and related expenses) and/or any other liability of any kind or nature, however arising, growing out of or otherwise connected with and/or related to any act, error and/or omission of yours

(including, but not limited to, your ownership and/or management of your Techy Store and/or any transfer of any interest in this Agreement or your Techy Store). We will have the right to control all litigation, and defend and/or settle any claim, against and/or including us and/or the Franchisor-Related Persons/Entities, or affecting our and/or their interests, in such manner as we deem appropriate in our Business Judgment, without affecting our rights under this indemnity.

B. Any goods and/or services provided by us, the Franchisor-Related Persons/Entities and/or any "approved" person/company/referral are provided without any warranties, express or implied, THE WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE BEING EXPRESSLY DISCLAIMED, absent a specific written warranty expressly provided in connection with a particular item or service.

7.5 Disclosure.

We can disclose, in Franchise Disclosure Documents and other places we designate, and/or as required by law, any information relating to your Techy Store, including your name, any address and/or phone number(s), revenues, expenses, results of operations and/or other information. Any disclosure by us shall be for reasonable business purposes and/or as required by law.

8. CONFIDENTIAL INFORMATION; EXCLUSIVE RELATIONSHIP.

8.1 Confidential Information - Non-Disclosure and Non-Use.

A. "Confidential Information" includes all information (current and future) relating to the operation of a Techy Store or the System, including, among other things, all: i) Manuals, training, techniques, processes, policies, procedures, systems, data and know how regarding the development, marketing, operation and franchising of a Techy Store; ii) designs, specifications, procedures and information about Products and Services used in the operation of a Techy Store and iii) all information regarding customers and suppliers, including any statistical and/or financial information and all lists.

"Confidential Information" is not intended to include any information that: (1) is or subsequently becomes publicly available other than by breach of any legal obligation; (2) was known by you prior to you becoming a Franchisee; or (3) became known to you other than through a breach by you of a legal obligation.

You agree that we own and control all domain names and URLs ("Uniform Resource Locator") relating to any Techy Business, as well as all information, lists and data related to past, present and future customers of your Techy Business. Your only interest in any of this proprietary and/or Confidential Information is the right to use it pursuant to this Agreement. You have the burden of proof and of going forward in any dispute between you and us involving the proprietary or confidential nature of any information.

B. You agree as follows: (1) to use the Confidential Information only for the operation of your Techy Business under a Franchise Agreement; (2) to maintain the confidentiality of the Confidential Information; (3) not to make or distribute, or permit to be made or distributed, any unauthorized copies of any portion of the Confidential Information; (4) not to alter, appropriate, use or distribute any Techy Equipment designs or specifications, or any substantially similar designs or specifications; and (5) to implement all prescribed procedures for prevention of unauthorized use or disclosure of the Confidential Information.

- C. You agree to disclose to us all ideas, techniques, methods and processes relating to a Techy Store that are conceived or developed by you and/or your employees. We shall have the perpetual right to use, and to authorize others to use, such ideas, etc., without payment to you.
- D. You will cause each of your employees, agents, principals and Affiliates to sign a form of confidentiality agreement containing substantially the same provisions as are set forth in this Section and as may be approved by us. You will provide us copies of the same upon request. A sample copy of an employee confidentiality agreement is attached as Exhibit 8.1.
- E. If your Techy Store is to be located and/or operated within, in conjunction with or as part of another business, you will first arrange for the other business and its personnel (as specified by us) to enter into appropriate arrangements to protect our Intellectual Property and other interests, including (but not limited to) signing of agreements with us regarding non-competition, confidentiality, non-solicitation of employees and customers and indemnity/insurance arrangements.

8.2 Exclusive Relationship, Restrictions on Similar Businesses During Franchise Term and After Transfer, Termination, Expiration, Repurchase, etc.

- A. <u>In Term Restrictions</u>: During the term of this Agreement and any successor franchise, neither you, nor any Affiliate of yours, nor any shareholder, member or partner of yours (if a business entity), nor any Immediate Family member of any of the foregoing, will, without our prior written consent, which may be withheld or conditioned in our Business Judgment: (1) have any direct or indirect interest anywhere in any Similar Business, or in any entity awarding franchises or licenses or establishing joint ventures or other business enterprises for the operation of Similar Businesses; or (2)perform any services anywhere as an employee, agent, representative or in any capacity of any kind for any Similar Business, or for any entity awarding franchises or licenses or establishing joint ventures to operate Similar Businesses.
- B. <u>Post Term Restrictions</u>: For two (2) years after the later of any one of the following terminating events: i) any transfer, Repurchase and/or Termination of this Agreement; ii) the expiration of this Agreement; and/or the date on which you stop operating your final Techy Store or using the Marks and/or System; all of the persons and entities named in Section 8.2 (A), above:
- (1) shall not accept or solicit any person, firm or company that has been a Techy customer during the period twelve (12) months prior to termination, nor try to divert any such customers from any Techy Store or Techy enterprise of any kind (including any operations owned by any Franchisor-Related Persons/Entity); and
- (2) shall be subject to all of the restrictions stated in Section 8.2 (A), above, with respect to Similar Businesses located, and/or services to be performed, in the Protected Territory nor within 10 miles of the Protected Territory sometimes hereinafter referred to as ("Non Compete area"). However, the two (2) year period following such terminating event shall be extended for an additional one (1) year if, during the first two (2) years a Techy Store is located in a portion of your former Techy Store Marketing Area. You and we intend to provide any such newly established Techy Store a reasonable period of time in which to launch its operations in a new market without unfairly being competitively disadvantaged by having a party familiar with and experienced in the Techy System operating in the same Marketing Area.
- (3) You are responsible for learning whether or not a particular location is within a Techy Store Non Compete area by providing us a written request for such information. Any and all determinations that we make regarding the Non Compete area will be final and binding on you.

- (4) You and we have expressly bargained and agreed that it is your obligation under this Agreement to ensure the compliance of each of the persons identified in Section 8.2 (A), with the restrictions described in this Section 8.2. The foregoing notwithstanding, we shall use reasonable judgment in evaluating whether or not the conduct of an Immediate Family member warrants our exercising any rights under this provision, considering your actual relationship to such member and his/her activities, among other factors. The restrictions of this Section do not apply to the ownership of shares of a Similar Business (of a class of securities listed on a stock exchange or traded on the over-the-counter market) which represent less than three percent (3%) of the number of shares of that class issued and outstanding.
- (5) You and we share a mutual interest in ensuring compliance with the limitations on competition described in this Section 8.2. A Franchisee's non-compliance with these restrictions would damage you, us and other Franchisees and unfairly limit reasonable expansion alternatives open to us and Techy System members. You acknowledge and agree that such protections can enhance the value of the Techy System to you as a Franchisee, and represent a reasonable balancing of your and our respective interests for which you and we have expressly bargained. You confirm that you possess valuable skills unrelated to the Franchised Business and have the ability to be self-supporting and employed regardless of the competitive restrictions described in this Section 8.2. You and we also acknowledge that the restrictions of this Article 8 will not generally prevent you from practicing a lawful profession, trade, or business and are limited to the express restrictions on solicitation of customers and operation of a Similar Business in certain limited geographical areas as detailed herein.
- (but not be limited to) the right to obtain equitable relief including a temporary restraining order and preliminary injunction without the requirement of a bond. All competitive restrictions will be extended for the length of time that any breach of the Post Termination Obligations is ongoing. You expressly acknowledge that you possess skills and abilities of a general nature and have opportunities for exploiting such skills. Consequently, enforcement of the covenants made herein will not deprive you or your owners of your or their goodwill or ability to earn a living. If any of the restrictions of this Section are determined to be unenforceable to an extent because of excessive duration, geographic scope, business coverage or otherwise, they will be reduced to the level that provides the greatest protection to us and the Techy System, but which is still enforceable, notwithstanding any choice-of-law or other provisions in this Agreement to the contrary.

9. FEES.

9.1 Initial Franchise Fee, Releases, etc.

- A. You will pay us the Initial Franchise Fee in the amount set forth in Paragraph 9.1 B hereinafter. You acknowledge and agree that the Initial Franchise Fee is deemed fully earned and nonrefundable. You further agree that the grant of this franchise constitutes the sole consideration for payment of the Initial Franchise Fee.
- B. In consideration for our granting this Franchise, you agree to pay us the sum of Fifteen Thousand dollars (\$15,000) as the Initial Franchise Fee.
- C. The language of the General Release, attached as Exhibit 1.2 is incorporated in and effective upon the signing of this Agreement, excepting only those claims solely related to the offer and sale of this Franchise where such releases are expressly prohibited by applicable law.

- D. As a condition to the occurrence of any of the following events (the "Events"), you and/or any affiliate/owner of yours will sign a General Release, excepting only (where such releases are expressly prohibited by applicable law) those claims solely related to the offer and sale of the new Franchise:
 - (1) the awarding of any future, additional or other franchise;
- (2) the renewal of this franchise and/or awarding of a successor franchise; any assignment or transfer (as defined in this Agreement) by you and/or any affiliate/owner of you; and/or
- (3) any other event described in this Agreement as being conditioned in whole or in part upon such a General Release (as defined in Article 22).

If we fail to request a General Release(s), as applicable, at the occurrence of any of the foregoing Events, then the occurrence of the Event itself shall constitute the grant of such General Release.

E. You and we have agreed on these provisions considering that: (1) the releases to be provided in the future will be effective as of future dates only, (2) the release requirement generally is triggered by a discretionary choice made by you to receive various future benefits (e.g. an award of an additional, successor, assignment franchise, etc.), (3) you providing a release to us (and we informing you of possible known claims by us) is a practical business approach if you and/or we propose to change, extend, expand or otherwise modify our relationship at a future date. You and we agree that setting mutual expectations for the receipt of such future releases and assenting to grant them now is more productive than being surprised by such requirements at a later point in our relationship.

9.2 Continuing Weekly Royalty Fee and Payment Dates.

- A. You shall pay us a Continuing Weekly Royalty Fee of 7% of Gross Revenues received by you from the previous week's operations payable on Monday at 3 p.m. of each week together with a report detailing all Gross Revenues received during the previous 7-day week (Monday through Sunday) and the amount of fee you are paying. In the event your Gross Revenues received by you for any previous 7-day week (Monday through Sunday) is \$7,500 or greater, then instead of paying us 7% of your Gross Revenues for the previous 7-day week (Monday through Sunday) your Royalty Fee of 7% for the week will be reduced to 6% of Gross Revenue for that previous 7-day week (Monday through Sunday). Your Continuing weekly royalty will remain at 7% for any 7 day week (Monday through Sunday) that you do not reach Gross Revenue of \$7,500.
- B. We can require that various Designated Equipment, Products and/or Services only be supplied by us, a Franchisor-Related Person/Entity and/or a designee of ours and we and/or one or more Franchisor-Related Person/Entities may derive additional revenues (and possibly profits) as a result of your purchases of such Designated Equipment, Products/Services. You and we have agreed on the foregoing continuing license fee rates based, in part, on your commitment not to use and/or obtain any such Designated Equipment, Products and/or Services from any source other than the source we require. The possibility of such arrangements, and your and our mutual expectations that you will faithfully observe your obligations to purchase such items as provided herein, and that the provisions of this subsection will be fully enforceable, form part of the underlying financial and business model on which your relationship with us, and the economic provisions of this Agreement, including Continuing license fee rates, are based.

Therefore, if you use and/or obtain such Designated Equipment, Products and/or Services from any source other than the source we require, (or if such restrictions are unenforceable for any reason), you will pay us a weekly continuing license fee of (a) twenty percent (20%) of the Gross Revenues attributable to

sales related to your purchase, use or otherwise of items, or from suppliers, in violation of the provisions of this subsection, or (b) if your violation of the provisions of this subsection cannot be easily related to sales of specific items by you, ten percent (10%) of the Gross Revenues of your Techy Store, in each case during any period in which you are in violation of the provisions of this subsection, subject to the continuing license fee described in Section 9.2 A., which will be regarded as a minimum against such percentage. Such increased weekly continuing license fee is intended by you and us to preserve a basic economic premise of this Agreement and is in addition to all other remedies available to us under this Agreement, at law and in equity for a breach of obligation. If an audit determines that you have used and/or obtained such Designated Equipment, Products and/or Services from any other source, you will pay the costs of such audit.

9.3 Electronic Funds Transfer.

You must participate in our then-current electronic funds transfer and reporting program(s). All continuing license fees owed and any other amounts designated by us must be received or credited to our account by pre-authorized bank debit by end of business on the Monday after a continuing license fee period. You authorize us to debit your account by an amount equal to the minimum continuing license fee if a continuing license fee payment is not received when due, and to collect the balance of any amounts owed in accordance with this Agreement. Any such non-payment or late payment of the actual amount due is a breach of this Agreement. We may also require that any amounts owed or to be owed to us and/or any of the Franchisor-Related Persons/Entities be paid by charges against your credit card, and you will provide us with appropriate authorizations on our request.

9.4 Interest and Late Fees on Late Payments and/or Reports.

All amounts you owe us and/or our Affiliates bear interest at the highest applicable legal rate for open account business credit, but not to exceed one and one-half percent (1.5%) per month. Additionally, we may require you to pay an administrative late fee of Fifty Dollars (\$50) for each late report and/or late payment. The foregoing amount is subject to inflation adjustment under Section 9.6, but will not exceed any applicable legal restrictions. If we experience repeated late payments by you, then we may require you to pay all amounts by cashier's check.

9.5 Application of Payments, Set-Offs etc.

As to you and/or any Affiliate of yours, we can:

- A. apply any payments received to any past due, current, future or other indebtedness of any kind in our Business Judgment, no matter how payment is designated by you, except that Marketing and Technology Fund contributions may only be credited to the Marketing and Technology Fund;
- B. set off, from any amounts that may be owed by us, any amount owed to us or any marketing fund; and
- C. retain any amounts received for your account (and/or that of any Affiliate of yours), whether rebates from suppliers or otherwise, as a payment against any amounts owed to us.

We can exercise any of the foregoing rights in connection with amounts owed to or from us and/or any Franchisor-Related Person/Entity.

9.6 Inflation Adjustments.

Amounts specified as being subject to inflation adjustment may be adjusted by us annually in our Business Judgment in proportion to the changes in the Consumer Price Index (U.S. Average, all items) maintained by the U.S. Department of Labor (or any successor index) as compared to the previous year. We will notify you of any such percentage adjustment.

9.7 Mandatory Convention Attendance, Possible Fee.

You are required to attend all meetings designated by us as mandatory (including without limitation the Techy annual convention), unless otherwise excused by us. One management-level individual shall attend on behalf of each of your Techy Stores. You will bear all costs of attendance.

9.8 Start-Up Package.

You agree to purchase from our affiliate, DrPhoneFix, Inc. a Start-Up Package set forth in the Start-Up Package Addendum attached hereto as Exhibit E-2. The Start-Up package provides Franchisee with the following items designated by Franchisor: limited start-up inventory, signage, furnishings, equipment, computer, Point of Sale system and software, equipment, cabinetry and preparation of the leased space to a condition acceptable by Franchisor for your commencement of business.

9.9 Discontinuance of Service.

If you do not pay amounts due to us timely under this Agreement, we may discontinue services to you, without limiting any of our other rights in this Agreement.

9.10 Cell Phone Accessory Support Fee.

Franchisor currently requires Franchisee to pay Franchisor a fee of \$100.00 per month for cell phone accessory support provided to Franchisee. This fee shall be paid to Franchisor by "Electronic Funds Transfer" as described hereinabove and will be payable on the 10th day of the month. If Franchisee purchases a minimum of \$300.00 of cell phone accessories during any month, Franchisor agrees to waive the \$100.00 Cell Phone Accessory Support Fee for that month.

9.11 Call Log PBX Software Fee.

Franchisor requires Franchise to pay Franchisor a fee of \$50.00 per month for its Call Log PBX Software. This fee is paid to Franchisor by "Electronic Funds Transfer" as described hereinabove and is payable on the 10th day of each month. This monthly fee may be increased at any time upon a thirty (30) day written notice to you.

10. YOUR STORE — IMAGE AND OPERATION.

10.1 System Compliance, Regular Upgrading.

A. You agree to operate your Techy Store in full compliance with the then-current Techy System and the Manuals. You agree to promptly comply at your expense with all then-current requirements, standards and operating procedures relating to every aspect of a Techy Store and its operations (including without limitation use of specified equipment, Products and Services, computer hardware and software; supplier programs and operating systems; signs, logos, designs and advertising/marketing materials and forms; website designs and formats).

- B. You will maintain your Techy Store at your expense according to all our standards for new Stores and promptly undertake all changes as are required by us from time to time in our Business Judgment. If you fail to do so, we may do so on your behalf. You agree to reimburse us within ten (10) days of our delivery of an account statement. You will not make any alterations to your Techy Store or its appearance as originally approved by us without our prior written approval.
- C. You agree at your sole expense that you and your employees will wear then-current Techy career apparel.

10.2 Designated Equipment, Products, Services and/or Suppliers.

- A. Your Techy Store will purchase, use and offer such Designated Equipment, Products and Services, as are specified by us from time to time. We may designate single or multiple suppliers for any given item or service and may concentrate purchases with one or more suppliers in our Business Judgment. Such suppliers may include, and may be limited to, us and/or companies affiliated with us. You will not offer or deal with any products or services not approved by us. If we disapprove a particular item, you will not use it.
- B. Designation of a supplier may be conditioned upon factors established by us in our Business Judgment, including without limitation performance relating to frequency of delivery, standards of service, and payment or other consideration to us or parties designated by us. We may approve, or revoke or deny approval, of particular items or suppliers in our Business Judgment.
- C. If you want to propose a new supplier of any product, item or service, you agree to submit to us, on our "Supplier Approval Form" and pay us a Supplier Approval Fee of \$1,000 at the time you submit the "Supplier Approval Form", sufficient written information about the proposed new supplier to enable us to approve or reject either the supplier or the particular product, item or service and be responsible for any expenses incurred in the process by us or you. We will have the 30 days from receipt of the information to approve or reject the proposed new supplier or product, item or service. We may consider in providing such approval not just the quality standards of the products or services, but their capabilities, financing terms and ability to service our franchise system as a whole. We may terminate or withhold approval of any Techy Store Materials or Operating Assets, or any supplier of such item, product, or service that does not meet our quality standards by giving you written notice. If we do so, you agree to immediately stop purchasing from such supplier or using such product, item or service in your Techy Store until we notify you that such supplier or such product, item or service meets our quality standards. At our request, you agree to submit to us sufficient information about a proposed supplier and samples of the proposed product, item or service for our examination so that we can determine whether they meet our quality standards. We also have the right to require our representatives to be permitted to inspect the proposed supplier's facilities at your expense. We reserve the right, at our option, to re-inspect or re-test from time to time the facilities and products, items or services of any approved supplier and to revoke approval upon a supplier's failure to meet any of our then-current criteria. Nothing hereinabove shall be construed to require us to approve any particular supplier. Your failure to comply with the provisions of Section 10.2 shall be deemed a material breach under this Agreement.
- D. You will not make any claims against us with respect to any supplier and/or related Products/Services (and/or our designation of, or our relationship with, any supplier/Products/Services). Claims with respect to any supplier-related and/or similar matters shall be made only against the supplier in question. You will provide us with written notice prior to taking any action in connection with such a claim. We will use diligent efforts to assist you in resolving any disputes with suppliers approved and/or designated by us.

10.3 Purchasing Cooperative.

We may require that you join and make required purchases/leases through the Techy purchasing cooperative or other entity designated by us. Such entity may adopt its own bylaws, rules, regulations and procedures, subject to our consent in our Business Judgment. We can require each such entity to submit monthly and annual financial statements, and can require that the annual financial statements be audited, all at the expense of such cooperative. Your failure to timely pay amounts due to, or comply with the bylaws, rules, regulations and procedures of such cooperative is a breach of this Agreement. We may offset against amounts we owe to you the amount of your unpaid cooperative obligations.

10.4 Compliance with Laws and Ethical Business Practices.

- A. You will operate your Techy Store in full compliance with all applicable laws, ordinances and regulations. We make no representations or assurances as to what (if any) licenses, permits, authorizations or otherwise may be required in connection with your Techy Store. It is your sole responsibility to identify and obtain all authorizations necessary to your operation. You will maintain high standards of honesty, integrity, fair dealing and ethical conduct in your business activities. You will notify us in writing within five (5) days of the commencement of any proceeding and/or of the issuance of any governmental order or action impacting you and/or your Techy Store.
- B. You agree to comply and/or assist us in our compliance efforts, as applicable, with any and all laws, regulations, Executive Orders or otherwise relating to antiterrorist activities, including without limitation the U.S. Patriot Act, Executive Order 13224, and related U.S. Treasury and/or other regulations. In connection with such compliance efforts, you agree not to enter into any prohibited transactions and to properly perform any currency reporting and other activities relating to your Franchised Business as may be required by us or by law. You are solely responsible for ascertaining what actions must be taken by you to comply with all such laws, orders and/or regulations, and specifically acknowledge and agree that your indemnification responsibilities as provided in Section 7.4 pertain to your obligations hereunder. You agree to sign and deliver to us, along with a signed copy of this Agreement, the attached Exhibit 10, "Executive Order 13224 and Related Certifications".
- C. In circumstances designated by us in the Manuals, or where otherwise reasonably required by us, you will give those customers who have made complaints appropriate refunds or otherwise deal with such complaints as we reasonably direct, you and we agreeing that such responses to customer complaints are a vital element in maintaining and enhancing the goodwill associated with the Marks. Policies and procedures (including, but not limited to, refunds and credits) adopted by us for Techy Stores owned and/or operated by us and/or our Affiliates will be conclusively presumed to be reasonable.
- D. In the marketing and operation of your Techy Store you will use each of, and only, the contracts, waivers and/or other forms and/or materials as are designated by us from time to time. We may provide you with template or sample forms of such items but it is your responsibility to have all such items which are to be used with prospective and/or actual clients/customers reviewed, at your expense, by an attorney licensed to practice law in the state(s) where your Techy Store will be located and/or operate, for compliance with all applicable state legal requirements. We make no warranty or representation that any contracts, waivers and/or other forms and/or materials, whether supplied by us or otherwise, are in compliance with the laws of any particular state(s). Prior to opening, and prior to use of any such items to be used with prospective and/or actual clients/customers, you will provide us, at your expense, with a letter from such attorney to us indicating that he/she has completed such review and that such items to be used with prospective and/or actual clients/customers meet, or have been modified to comply with, all applicable state legal requirements.

10.5 Management and Personnel of Your Store, Training.

- A. You will keep us advised, in writing, of all management and non-management personnel involved in the operation or otherwise of your Techy Store. Your Techy Store must be personally managed on a full-time basis by a person who has successfully completed mandatory training and met then-current standards as specified by us. We require that you must have management responsibility and authority over the Techy Store on a day-to-day basis and oversee all operations. You and your initial manager must attend our training. At your request, we will make additional or refresher training, in form and content as we deem appropriate, available at your Techy Store or other locations we designate for an additional charge to you at our then current rate, which at the time of the Franchise Agreement is \$400 per day plus travel and living expense, minimum of 1 day charge.
- B. You are solely responsible for the hiring and management of your Techy Store employees, for the terms of their employment and for ensuring their compliance with any training or other employment related requirements established by us from time to time in our Business Judgment.
- C. We have the right to deal with the manager regarding routine operations and reporting requirements. You will ensure that our records for your Techy Store managers/supervisors are kept current.
- D. If you have obtained our prior approval and Franchisee is a corporation, partnership, limited liability company or other form of legal entity, Franchisee and the owners agree, represent, and warrant and covenant that:
- (1) you have the authority to execute, deliver and perform your obligations under this Agreement and are duly organized or formed and validly existing in good standing under the laws of the state of your incorporation or formation;
- (2) your organizational or governing documents will recite that the issuance and transfer of any ownership interests by you are restricted by the terms of this Agreement, and all certificates and other documents representing ownership interests in you will bear a legend referring to the restrictions of this Agreement;
- (3) Franchisee shall provide to Franchisor prior to the execution of this Agreement, true and correct copies, as applicable, of Franchisee's articles of incorporation, by laws, partnership agreement, articles of organization and limited liability company operating agreement and other governing documents and amendments thereto, as well as resolutions of the Board of Directors, partners or members authorizing entry into and performance of this Agreement. During the term of this Agreement, Franchisee shall promptly provide to Franchisor true and correct copies of any amendments or other changes to such governing documents. No such documents shall contain any provision that is contrary to or inconsistent with any provision of this Agreement;
- (4) you and your owners agree to revise the Principal Owners Statement as may be necessary to reflect any ownership changes and to furnish such other information about your organization or formation as we may request (no ownership changes may be made without our approval, which shall not be unreasonably withheld);
- (5) a Principal Owner of the Business Entity (with ownership of at least 10% of its voting securities) must: (i) have management responsibility and authority over the Techy Store on a day-to-day basis; (ii) oversee the Techy Store's operations; (iii) be bound by our then-current form of

Confidentiality Agreement (or other form satisfactory to us); and (iv) satisfactorily complete our initial training program and any other training programs we request during the term.

(6) each of your owners during the term of this Agreement will sign and deliver to us our standard form of Principal Owner's Guaranty undertaking to be bound jointly and severally by all provisions of this Agreement and any other agreements between you and us. A copy of our current form of Owner's Guaranty and Assumption of Business Entity Franchisee's Obligations agreement is attached to this Franchise Agreement as Exhibit 1. The spouse of each of your owners will execute a spousal consent in the form attached hereto as Exhibit 23.

10.6 Insurance.

- A. You will maintain in force policies of insurance issued by carriers approved by us covering various risks, as specified by us from time to time. We may specify the types and amounts of coverage required under such policies and require different and/or additional kinds of insurance at any time, including excess liability insurance. Each insurance policy must: (1) name us and our Affiliates as additional named insureds; (2) contain a waiver of all subrogation rights against us, our Affiliates and any successors and assigns; (3) and provide thirty (30) days prior written notice to us of any material modifications, cancellation, or expiration of such policies.
- B. If you fail to maintain required insurance coverage, we may obtain such insurance coverage on your behalf. You will pay us on demand any costs and premiums incurred by us plus a 15% service fee.
- C. Current insurance requirements include the following and are subject to change by us: i) comprehensive general liability insurance against claims for bodily and personal injury, death and property damage caused by, or occurring in conjunction with, your Techy Store; ii) all risk property and casualty insurance for the replacement value of your Techy Store and all associated items; and iii) business interruption insurance providing for continued payment of all amounts due us and/or any Affiliate of ours under this Agreement.

10.7 Program Participation.

We may condition your participation in any program, or your receipt of any Techy System benefits, on your being in Good Standing.

10.8 Continued Payment of Continuing License Fees, etc. During Closure.

You will immediately notify us of any closure of your Techy Store for any reason and submit a plan for re-opening. All financial obligations of yours to us or to any Franchisor-Related Person/Entity (including continuing license fees) will remain in effect during such closure period. Any such closure not authorized and/or excused by us shall be a default of this Agreement, entitling us to all remedies available hereunder, at law and in equity.

10.9 Customer Satisfaction, Quality Controls, etc.

We may institute various programs for auditing customer satisfaction and/or other quality control measures. We may require you to pay for such program costs. You agree to request your customers to participate in any surveys performed by or on behalf of us, using forms prescribed by us from time to time.

10.10 Coupons, Certificates and Vouchers.

Except as other sites permitted by Franchisor in writing, Franchisee shall honor all coupons, gift certificates and vouchers sold by Franchisor or other Franchisees in the franchise System and upon redemption thereof shall be entitled to credit the retail price of the item provided against Gross Revenue. Any coupons offered or proposed by Franchisee must be approved in writing by Franchisor prior to being extended.

10.11 Hours of Operation.

You agree to comply with all operating procedures relating to hours of operation and days you are required to be open for business to the public (subject to applicable law or terms of you lease) as may be set by us or changed by us from time to time. You acknowledge and agree that all mandatory operating procedures prescribed from time to time by us in the Manuals or otherwise shall constitute binding obligations on your part, and any failure by you to adhere to such mandatory operating procedures shall constitute grounds for termination of this Agreement by us as provided for herein.

You further acknowledge that due to peculiarities of particular market areas and circumstances, complete and detailed uniformity may not be practical or in the best interests of all franchisees. We reserve the absolute right to vary hours of operation and/or days you are required to be open for business to the public, as they relate to a particular franchise or group of franchisees.

For any day your Techy Store is not open, except as permitted hereinabove, you will be charged \$250 per day unless you have obtained, in writing, permission from us to close for the day(s) not permitted hereinabove. Any amount owed hereinabove will be due and payable upon demand by us and such amount due may be collected by us through electronic funds transfer.

11. MARKETING.

11.1 Marketing and Technology Fund.

- A. You agree to pay us a Marketing and Technology Fund ("Fund") fee of \$785 per month, or two percent (2%) of Gross Revenues, whichever is greater, which fee may be increased up to 10% per year upon a thirty (30) day written notice. We must receive the Fund fee by the fifth day of each month. We require you to pay the Fund fee by electronic funds transfer. You agree to comply with the procedures we specify in our Confidential Operations Manual and perform such acts and sign and deliver such documents as may be necessary to accomplish payment by this method. You will give us authorization in a form we designate to initiate debit entries or credit correction entries to your bank account for each month's Fund fee and any interest charges due.
- B. We will direct all marketing and technology programs, with sole control over the creative concepts, materials and media used in such programs, and the placement and allocation of Fund advertising. You acknowledge that the Fund is intended to further general public recognition and acceptance of the Proprietary Marks for the benefit of the Techy Store System and other benefits derived from web based media Manager/Consumer website we develop or utilize. You further acknowledge that we and our designees undertake no obligation in administering the Fund to make expenditures for you which are equivalent or proportionate to your contributions, to ensure that any particular franchisee benefits directly or pro rata from the placement of advertising or to ensure that any advertising impacts or penetrates your Protected Territory. The Fund is not a trust and we are not a fiduciary with respect to the Fund.

- C. The Fund may be used to meet any and all costs of administering, directing, preparing, placing and paying for national, regional or local advertising, including (without limitation): television, radio, magazine, newspaper and worldwide web/internet advertising campaigns; other advertising, marketing and public relations materials; point-of-purchase materials; paper goods; consumer research, interviews and related activities; the creation, maintenance and periodic modification of the Techy Store website; reviewing any advertising material you propose to use (as provided below); search engine optimization; establishing a third party facility for customizing local advertising materials; the creative development of signage, message boards, posters and unit décor items; the development and creative activity associated with loyalty programs, promotions and public relations events; accounting for Techy Store Fund Revenues and expenditures; attendance at industry related conventions, shows or seminars; other activities that we in our business judgment believe are appropriate to enhance, promote and/or protect the Techy System or any component thereof; and, engaging advertising agencies to assist in any or all of the foregoing activities, including fees to have print, broadcast and/or internet advertising placed by an agency, and all other advertising agency and public relations fees.
- D. We need not maintain the sums paid by franchisees to the Fund, or income earned from the Fund, in a separate account from our other funds, but we may not use these amounts under any circumstance for any purposes other than those provided for in this Agreement. We may, however, expend monies from the Fund for any reasonable administrative costs and overhead that we may incur in activities reasonably related to the administration or direction of the Fund and marketing and technology programs for franchisees including, without limitation, preparing marketing and advertising materials; working with advertising agencies, advertising placement services and creative talent; preparing an accounting of contributions to the Fund and the annual statement of Fund contributions and expenditures provided for below; and, otherwise devoting our personnel, resources and/or funds for the benefit of the Fund. Our right to expend monies from the Fund to reimburse us for such activities is exclusive of any advertising agency or public relations fees which the Fund must expend to secure the services of an advertising agency or public relations firm or to have print, broadcast or internet advertising placed by an agency.
- E. Within 60 days following the close of our fiscal year, we will prepare (but not audit) a statement detailing Fund income and expenses for the calendar year just ended a copy of which statement will be sent to you upon written request.
- F. We expect to expend most contributions to the Fund during the calendar year when the contributions are made. If we expend less than the total sum available in the Fund during any fiscal year, the excess amount will be carried forward to the following fiscal year to be used as provided for in subsection g. If we advance and expend an amount greater than the amount available in the Fund in any calendar year (in addition to any sum required to be expended because we did not expend all the sums in the Fund during the preceding year), we will be entitled to reimburse ourselves from the Fund during the following fiscal year for all such advanced sums, with interest payable on such advanced sums at the greater rate of 1.5% per month or the maximum commercial contract interest rate permitted by law (with interest accruing the first calendar day following the day on which we advance and expend any such sum).
- G. We reserve the right to use any media, create any programs and allocate advertising funds to any regions or localities in any manner we consider appropriate in our business judgment. The allocation may include rebates to individual franchisees of some or all of their Fund contributions for local advertising expenditures if, in our judgment, our national or regional advertising program or campaign cannot effectively advertise or promote in certain regions or communities. If we determine that the total Fund contributions collected from all franchisees is insufficient to sustain a meaningful regional or national advertising campaign, we may rebate all or a portion of the Fund contributions to franchisees on a pro rata basis. Franchisees must expend any rebate on the types of local advertising and media that we determine.

All rebate advertising expenditures must be documented to us in a monthly rebate advertising expenditure report form which we will furnish.

- H. Although the Fund is intended to be a perpetual duration, we maintain the right to terminate the Fund, but will not do so until all of the monies in the Fund have been expended for advertising and promotional purposes.
- I. The Fund will not be used for any activity whose sole purpose is the sale of franchises; provided, however, that the design and maintenance of our Web site (for which Fund monies may be used) may include information and solicitations for prospective franchisees and public relations and community involvement activities which may result in greater awareness of the Techy Store brand and the franchise opportunity.
- J. No Techy Store which we or our affiliates have any interest in or operate will be required to participate in or contribute to the Marketing and Technology Fund or other advertising programs provided for in this Section, unless it determines to participate in a regional or joint franchise advertisement setting forth the names, addresses and/or telephone numbers of individual Techy Stores, including those owned and operated by us or our affiliates. If we or our affiliate decide to participate in any joint or regional advertising of this type, then we or our affiliate will contribute our or its proportionate share of the cost of the advertisement to the franchisee group sponsoring the advertisement.
- We or our designee will direct all programs financed by the Fund, with sole control K. over the allocation and any Internet or Intranet websites, networks or communities it operates or participate in, or which requires your participation. You agree that the Fund may be used to pay the costs associated directly or indirectly with the operation, maintenance, hosting or development of website bearing our marks; or establishing Internet, Intranet, website or other forms of e-commerce communities, networks, systems, methods, processes, databases or monitoring systems, which may include our establishing one or more Internet or Intranet websites for purposes of: linking this Agreement; our sharing or selling information to third parties; our establishing business to business or business to customer e-commerce; promoting the development and growth of franchises or soliciting franchisees; or your reporting of License Fees, Gross Revenues or other information as we designate from time to time. The Fund may be used for defraying administrative hosting, development maintenance costs and overhead incurred by us or our designees in connection with the Fund. The Fund may periodically furnish you with samples of advertising, marketing and promotional formats and materials at no cost. Multiple copies of such materials will be furnished to you at our direct cost of producing them, plus any related shipping, handling and storage charges.
- L. Neither we (nor any of the Franchisor-Related Persons/Entities, including the FAC) will be liable for any act or omission in connection with the Marketing and Technology Fund which is consistent with this Agreement or which is done in subjective good faith. You and we expressly agree that none of the relationships with you in connection with the Marketing and Technology Fund are in the nature of a "trust," "fiduciary" or similar special arrangement.
- M. Subject to the express requirements of this Agreement that your contributions will only be spent as authorized herein, you agree that we may deny access to any and all programs and/or materials created by, and benefits of, the Marketing and Technology Fund to you and to any Franchisees who are not in Good Standing.

11.2 Your Participation in the Marketing and Technology Fund.

You agree to participate in all Marketing and Technology Fund programs. You have the right to set your own prices, except that we may specify maximum prices for goods or services to the greatest degree permitted by law. You will fully honor all coupons, price reduction and other promotions/programs as directed by us. The Marketing and Technology Fund may furnish you with marketing, advertising and promotional materials; however, we may require that you pay the cost of producing, shipping and handling for such materials.

11.3 Your Local Store Marketing Activities.

- A. You must spend for local advertising and promotion of your Techy Store each month the greater of 1.5% of the Gross Revenues received during the previous month or \$500, subject to inflation adjustment as set forth in Section 9.6. If we request it, you will submit verification of your expenditures in a form prescribed by us in our Business Judgment. Appropriate local advertising expenditures may include, but are not limited to, classified telephone directory listings and advertising. The value of discounts, coupon redemptions and/or products or services given without charge shall not be applied toward satisfaction of your local advertising obligation under this Section.
- B. Your advertising will be in good taste and conform to ethical and legal standards and our requirements. Samples of all advertising and promotional materials (and any use of the Marks and/or other forms of commercial identification) for any media, including the Internet, World Wide Web or otherwise, must be submitted to us for our review and consent prior to use, which approval we may condition or withhold solely in our Business Judgment. You agree not to use any materials or programs disapproved by us at any time in our Business Judgment and will use all materials and programs designated by us as mandatory. We can require that a brief statement regarding the availability of Techy franchises be included in advertising used by you and/or that brochures regarding purchase of Techy franchises be displayed in your Techy Store.
- C. We reserve the rights to all Internet and other electronic marketing and operational usage related to Techy and do not currently permit franchisees to operate websites or internet sites in connection with the operation or marketing of Techy products and/or services. Any future use of the Internet, World Wide Web or other electronic media by you in connection with your Techy Store will be as specified in advance and in writing by us in our Business Judgment from time to time, whether in the Manuals or otherwise.

11.4 Marketing Cooperative and Groups.

We may decide to form one or more associations and/or sub-associations of Techy Stores to conduct various marketing-related activities on a cooperative basis (a "Franchise Marketing Group" or sometimes "FMG"). If one or more FMGs (local, regional and/or national) are formed covering your area, then you must join and actively participate. Each Techy Store, whether franchised or non-franchised, will be entitled to one (1) vote, but in order to vote franchised Techy Stores must be in Good Standing. You may be required to contribute such amounts as are determined from time-to-time by such FMGs. Each FMG may adopt its own bylaws, rules, regulations and procedures, subject to our consent in our Business Judgment. Any failure to timely pay amounts due to, or to comply with the bylaws, rules, regulations and procedures of an FMG, is a breach of this Agreement. We may offset against amounts we or any Affiliate owe to you the amount of your unpaid FMG obligations. We will have the right to have a representative participate in all FMG meetings as a non-voting participant. While we are not required to do so, if we submit any matters for approval to an FMG of which you are a member, and approval is granted, the approval will be binding on you.

12. STORE RECORDS AND REPORTING.

12.1 Bookkeeping, Accounting and Records, Cash Register, Computer and Other Systems.

You must obtain and maintain at your sole expense accounting, sales, reporting and records retention systems conforming to any requirements prescribed by us from time to time, including electronic systems with online access for us. Such systems may include, but are not limited to, electronic cash register, computer and point-of-sale (POS) systems, and software programs, and may have components only available from us, a Franchisor-Related Person/Entity and/or designated suppliers. We reserve the right to use, and to have full access to, all cash register, computer and any other systems, and the information and data they contain. We may charge a reasonable fee for the license, modification, maintenance or support of software or any other goods and/or services that we furnish to you in connection with any of the systems.

12.2 Reports, Financial Statements and Tax Returns.

- A. You will provide to us such information regarding the sales and operation of your Techy Store, and in such form and format, as we specify from time to time in our Business Judgment. We may elect to obtain such information through a variety of methods, including direct online access, facsimile transmissions and written copies. Current information requirements include, but are not limited to, the following, and are subject to change by us:
- (1) Sales and operations reports for each Continuing license fee Period, due at the same time as the corresponding continuing license fee payment; and
- (2) within forty-five (45) days after the end of each fiscal year, an unaudited fiscal year-end balance sheet and income statement for your Techy Store, prepared in accordance with generally accepted accounting principles, and verified and signed by you;
- (3) retention of all records of or relating to your Techy Store, including all income, sales and other tax returns, for the term of this Agreement and one (1) year thereafter.
- B. You agree to provide such other data, information and supporting records for your Techy Store as we reasonably may request from time to time, including without limitation copies of your sales tax returns and those portions of your income tax returns relating to your Techy Store. We can require you to provide us, at your expense, with an annual audited financial statement prepared by a certified public accountant.

13. INSPECTIONS AND AUDITS.

13.1 Our Inspections, etc.

We and/or our agents will have the right, at any time during business hours, and without prior notice to you, to: i) inspect your Techy Store and related activities and items and record the same; ii) remove samples for testing and analysis; iii) interview personnel; iv) interview customers; and v) conduct inventories. You will cooperate fully in connection with such matters. We may require you or an individual designated by us to meet at our headquarters or other location designated by us, for the purpose of discussing and reviewing your Techy Store's operations, financial performance and other matters. If we inspect your Techy Store and find an unapproved supplier item(s), we may elect to charge you a non-compliance fee of \$250 per day until you have cured your default. Notice of non-compliance hereunder is not required and

the non-compliance fee shall be due from the date of inspection until you have cured the violation or Franchisor elects to terminate this Franchise Agreement.

13.2 **Audit.**

We and/or our agents will have the right at any time during business hours, and without prior notice to you, to inspect and/or audit business records relating in any way to your Techy Store and the books and records of any person(s), corporation or partnership which holds, or does business with, the Franchise. Such business records may include, but are not limited to, bookkeeping and accounting records, sales and income tax records and returns, cash register tapes, invoices, and deposit Revenues. Our right to audit includes the right to access all cash registers, computers and other equipment by electronic means. You will cooperate fully with such an audit.

13.3 Gross Revenues Understatements.

If any inspection or audit discloses an understatement of Gross Revenues, you will pay to us the continuing license fees and marketing contributions due on the understated amount, plus interest at 1.5% per month, from the date originally due until the date of payment. We may require you to reimburse us for the cost of the inspection or audit, including, without limitation, the charges of any independent accountants, and related travel and per diem charges for our and their employees, if:

- A. any inspection or audit is necessary because of your failure to timely furnish required information/reports; or
 - B. Gross Revenues is understated for any period by more than two percent (2%).

In addition to all other remedies and rights of ours hereunder or under applicable law, we may Terminate this Agreement if:

- A. Gross Revenues is understated for any period by more than five percent (5%); or
- B. any understatement is determined by us to be intentional.

14. TRANSFER.

14.1 Transfers by Us.

This Agreement, and any or all of our rights and/or obligations under it, are fully transferable by us in our Business Judgment, in whole or in part, without your consent; provided that any such transferee shall appear at the time of the transfer to have financial resources reasonably appropriate to fulfill its obligations under this Agreement. For the purposes of this Section 14.1, we shall be entitled to rely upon financial statements provided to us by the transferee. If we transfer this Agreement, only the transferee will have obligations to you and our obligations (and those of any of the Franchisor-Related Persons/Entities) will be extinguished. You specifically acknowledge and agree that we may:(1) be sold and/or we may sell any or all of our intellectual property and/or other assets (including the Marks); (2) go public; (3) engage in a private or other placement of some or all of our securities; (4) merge, acquire other entities and/or assets (competitive or not); (5) be acquired by a competitive or other entity; (6)) and/or undertake any refinancing, leveraged buy-out and/or other transaction. You agree that we will have no liability to you resulting from our entering into any transactions permitted hereunder. We also may, on a permanent or temporary basis,

delegate any or all of our duties to another company to perform. In such event, you will look only to such other company for the performance of such duties.

14.2 Transfers by You.

- A. The rights and duties created by this Agreement are personal to you (or your owners, if the Franchisee is a Business Entity). We have awarded the Franchise relying on the individual integrity, ability, experience and financial resources of you or such owners. Therefore, neither this Agreement, the Franchise, the Franchisee nor your Techy Store, Techy Café or Wrapped Techy Van, (or any interest in, or the assets of, any of the foregoing) may be transferred without our prior written approval. Any transfer or attempted transfer without our approval is null and void.
- B. The term "transfer" includes (but is not limited to) any voluntary or involuntary assignment, sale, gift, pledge or any grant of any security or other interest (whether partial or whole, or direct or indirect), by you (or your owners, if the Franchisee is a Business Entity). A transfer also includes the following events: (1) any transfer of ownership of capital stock or any partnership or similar interest; (2) any merger, consolidation or issuance of additional securities representing an ownership interest in the Franchisee; (3) any sale of voting stock of the Franchisee or of any security convertible to voting stock; (4) any transfer in a corporate or partnership dissolution, divorce, insolvency proceeding or otherwise by operation of law; (5) any transfer of any interest in any revenues, profits, or assets of your Techy Store and which is not in the ordinary course of business; or vi) any transfer to a business entity and/or a trust or similar entity. A transfer of ownership, possession or control of your Techy Store, or of its assets, may only be made with a transfer of the Franchise. Any transfer in the event of death or disability will be governed by Section 14.5.

14.3 Conditions for Approval of Any Transfer.

- A. All of the following conditions must be met prior to, or concurrently with, the effective date of any transfer. We may waive any condition in our sole and absolute discretion.
- (1) You must be in compliance with this Agreement, the Manuals, all other agreements between you and us (including any of our respective Affiliates), and all leases/subleases or the Operating Agreement with any party, and the transferee must expressly assume all obligations under all such agreements; and
- (2) You must have continuously operated and been in sole possession of your Franchised Business for at least five (5) full years from the effective date of this Agreement. We will not authorize or approve a transfer of your Franchised Business prior to five (5) full years of continuously uninterrupted operation absent extremely urgent and unavoidable circumstances, the sufficiency of which we will determine in our sole and absolute discretion; and
- (3) The transferee and its owners must meet our then-current requirements for new franchisees, including but not limited to business experience, aptitude and financial resources; and
- (4) You must meet all payment and reporting obligations under the Franchise Agreement and any other agreements between you and us (and any of our respective Affiliates). Promissory notes shall be accelerated and paid in full; and
- (5) All obligations to third parties in connection with your Techy Store must be satisfied or assumed by the transferee; and

- (6) Your Techy Store and its operations must have been brought into full compliance with the Manuals and specifications and standards then-applicable for new Techy Stores; and
- (7) The transferee must pay us a Seven Thousand Five Hundred Dollars (\$7,500) training fee prior to attending training (which will not be refundable once training commences, whether or not the transfer is consummated) and successfully complete our Training Program for new franchisees; and
- (8) The transferee must, execute our then-current form of Franchise Agreement and ancillary documents (including guarantees) as are then customarily used by us in the grant of franchises; the term of such new Franchise Agreement shall, at our option, be either for the balance of the term of this Agreement or for the full term generally awarded to new franchisees as of the time of the transfer.
- (9) The transfer must be completed in compliance with the terms of any applicable leases or the Operating Agreement and other agreements and with all applicable laws, including, but not limited to, licensing and operations-related laws and/or laws governing franchise sales; and
- (10) You or the transferee must pay us, with your application for a transfer, a non-refundable transfer fee of Ten Thousand Dollars (\$10,000).
- (11) You and each of your owners and/or Affiliates, and the transferee (and each owner and/or Affiliate of the transferee), must sign a General Release; and
- (12) Any grant of a security or similar interest in connection with a transfer (which grant may or may not be permitted by us in our Business Judgment), will be subordinated to our rights and the rights of any Franchisor Related Person/Entity under the Franchise Agreement or any other agreement; provided that we may refuse to allow you or anyone else to grant or receive a pledge, mortgage, lien or any security or similar interest in and/or to the Franchise or the Franchised Business (or any of its assets) if, after having expended commercially reasonable efforts in discussions with lenders or other applicable parties, we are unable in our Business Judgment to obtain appropriate protections for our rights under this Agreement and/or for Techy System interests; and
- (13) You will agree with the transferee not to compete after the transfer in accordance with restrictions acceptable to us and substantially similar to those described in Section 8.2 (B), above, to the maximum extent permitted by law. We shall be named a third party beneficiary of such agreement; and
- (14) We may (but are not required to) withhold or condition our consent to any transfer in our Business Judgment, particularly if we believe that the terms of transfer jeopardize the economic viability of the franchise or based on other circumstances of the transfer, and/or if we would not normally directly award a franchise in such a situation.
- B. You agree that we may (but are not required to) discuss with you and/or the proposed transferee any matters related to any transfer and/or proposed transfer at any time which we consider to be appropriate in our Business Judgment without liability (including our opinion of the terms of sale, performance of your franchise, etc.). You expressly consent to any such discussions by us.
- C. Neither you nor any transferee shall rely on us to assist in the evaluation of the terms of any proposed transfer. You acknowledge and agree that an approval of a proposed transfer shall

not be deemed to be an approval of the terms, nor any indication as to any likelihood of success or economic viability.

14.4 Additional Conditions for Transfer to a Business Entity.

We will consent to a transfer from you to a Business Entity entirely owned by you and formed for the sole purpose of operating the Techy Store if the conditions described in 14.3, above, and the following conditions are met. Such a transfer will not relieve you of your obligations under this Agreement. You will remain jointly and severally liable to us for your and the Business Entity's obligations.

- (1) The Business Entity's stock certificates (and/or other applicable evidences of ownership and all documents of formation/governance) must recite that any ownership interest in the Business Entity is restricted by the terms of this Agreement; and
- (2) You must have (and continue to maintain) management control and ownership of at least fifty-one percent (51%) of the Business Entity and personally manage its affairs; and
- (3) The individual Franchisee (or, if the Franchisee is a partnership, at least one of the partners) must be and remain the chief executive officer, chief operating officer or chief financial officer and meet our then-current training requirements. If the Franchisee is or becomes a corporation, LLC, partnership or other business entity, the chief executive officer, chief operating officer or chief financial officer of such entity must always meet all of our then-current training and other standards; and
- (4) The transferee must enter into an approved form of assignment in which the Business Entity assumes all of the Franchisee's obligations under this Agreement and any other agreements with us and/or a Franchisor-Related Person/Entity, and any other documents we may require as provided in 14.3 A. (7), above; and
- (5) All current and future owners of the Business Entity must agree in writing to comply with this Agreement and any other agreements with us and/or any Franchisor-Related Persons/Entities. We may at our option and in our Business Judgment require any and all owners to jointly and severally guarantee (in a written form approved by us) any such obligations of the Business Entity under any such agreements. The current approved form of Owner's Guaranty is attached as Exhibit 1 to this Agreement; and
- (6) No public offerings of debt or equity ownership in the transferee entity may be conducted, and no shares of any type issued without obtaining our prior written consent; and
- (7) We may require that each of the present and/or future shareholders, directors, and/or officers execute confidentiality and non-competition agreements with terms substantially similar to those described in Sections 8.1 and 8.2, respectively.
- (8) In any event, we may withhold or condition our consent to any transfer as we deem appropriate in our Business Judgment, based on the circumstances of the transfer or otherwise.

14.5 Death or Disability of Franchisee.

A. If the Franchisee, or if the owner of the Franchisee with a controlling interest, dies or is permanently disabled, then his or her interest in this Agreement, the Franchise and/or the Franchisee shall be transferred to a third party subject to all of the provisions of this Article 14. A "permanent disability" occurs if you are not able to personally, actively participate in the management of your Techy

Store for six (6) consecutive months. Any transfer under this Section shall be completed within six (6) months from the date of death or permanent disability. If no transfer occurs, the Franchise will automatically terminate at the end of such period, unless a written extension is granted by us in our Business Judgment.

B. We can (but are not required to) operate the Franchised Business on your behalf and at your expense in the event of your death, disability or absence. We can pay ourselves a reasonable amount for our management services and other costs. We will use reasonable efforts and business judgment in managing the business, but will in all cases be indemnified by you (and/or your estate) against any costs and/or liabilities related in any way to our management and the operation of the Franchised Business. We are expressly authorized by you to manage in good faith and on terms that we consider appropriate in our Business Judgment, including payment of any past, current and/or future obligations to us or to any other creditor out of assets and/or revenues of the Franchised Business.

14.6 Effect of Consent to Transfer.

Our consent to a transfer is not a waiver of any claims we may have against you, and you are not relieved of any obligations to us or any Franchisor-Related Persons/Entities (including any defaults by any transferee.) In any case, your obligations under the Post Termination Provisions will survive any transfer of this Agreement. Any dispute regarding any proposed or completed transfer will be resolved through the dispute resolution provisions of this Agreement. Neither we nor any Franchisor-Related Persons/Entities will have any liability to you or any proposed or actual transferee in connection with our examination and/or possible consent or withholding of consent involving any transfer or proposed transfer, or our exercise of any right of ours, which is consistent with this Agreement. You agree to indemnify and hold us harmless from any liability to you, the proposed transferee or otherwise.

14.7 Our Right of First Refusal.

- A. We have a right of first refusal regarding any proposed transfer subject to this Agreement, excluding only those transfers which are subject to Section 14.4. For each non-excluded proposed transfer, you will provide us with a true and complete copy of the offer received by you (and any ancillary agreements), and the conditions to transfer described in Sections 14.3 and 14.4, as applicable, will be met. The offer and the price and terms of purchase must apply only to an interest in this Agreement, the Franchise, your Techy Store or the Franchisee. Any value attributable to the goodwill of the Marks, Techy System elements, Confidential Information or any other assets, tangible or intangible, related to the Techy Brand and System will be excluded from the purchase price, but goodwill related solely to the value of your Techy Store as a going business may be included.
- B. We will give you written notice of our decision to exercise our right of first refusal within thirty (30) days from the date of our receipt of the offer and ancillary documents. If any of the assets to be purchased do not meet the standards we then apply to new Techy Stores, or if you are in default, we can require that the Store be brought into compliance and any defaults cured before the thirty (30) day period begins. We may substitute cash for any form of payment proposed in such offer and will have a reasonable period of time in which to prepare for the close of the transaction, generally sixty (60) days. The purchase price to be paid by us will be the price specified in the proposed transfer, less the value attributable to the goodwill of the Marks, Techy System elements, Confidential Information or any other assets, tangible or intangible, related to the Techy Brand and System. We will be entitled to purchase any interest subject to all Customary Representations, Warranties and Agreements. We can require that the closing of the sale be through an escrow. You and we will comply with any applicable bulk sales and/or similar laws, and you will maintain all insurance policies until the date of closing. We will have the right to set off against any amount of money payable by us all amounts due from you and/or your Affiliates to us and/or our Affiliates.

We will also have the right, in our Business Judgment, to pay any amount otherwise payable to you directly to your creditors in satisfaction of your obligations. If you violate any of your obligations that expressly or by their nature survive this Agreement, we will not be obligated to pay any amount otherwise due or payable to you thereafter. In connection with such purchase, you and each transferor (and your respective Affiliates) will sign a General Release.

C. If we do not exercise our right of first refusal, you or your owner may complete the sale to such purchaser on the exact terms of such offer, subject to the conditions of this Article 14. If there is a material change in the terms of the sale, we will have an additional right of first refusal on the same terms and conditions as are applicable to the initial right of first refusal. Our rights under this or any other Section are fully assignable.

15. SUCCESSOR FRANCHISE.

15.1 Your Right to Acquire a Successor Franchise.

This Agreement expires 10 years from the Agreement Date. Upon expiration, if you (and each of your owners) have substantially complied with this Agreement during its term, and provided that:

- A. you maintain possession of and agree to remodel and/or expand the Techy Store, add or replace improvements, equipment and signs and otherwise modify the Techy Store as we require to bring it into compliance with specifications and standards then applicable for Techy Store; or
- B. if you are unable to maintain possession of the Site, or if in our judgment the Techy Store should be relocated, you secure substitute premises accepted by us, you develop such premises in compliance with specifications and standards then applicable for Techy Stores and continue to operate the Techy Store at the Site until operations are transferred to the substitute premises, then, subject to the terms and conditions set forth in this Article 15, you will have the privilege to apply for successor franchises to operate the then existing Techy Store as a Techy Store (each a "Successor Franchise"), for one additional 10-year period on the terms and conditions of the franchise agreement we are then using in granting Successor Franchises for Techy Stores, which may contain materially different terms and conditions than this Agreement.

15.2 Grant of a Successor Franchise.

- A. <u>Your Election</u>: You agree to give us written notice of your election to acquire a Successor Franchise during the first 90 days of the 9th year of this Agreement. We agree to give you written notice ("Response Notice"), not more than 90 days after we receive your notice, of our decision:
 - (1) to grant you a Successor Franchise;
- (2) to grant you a Successor Franchise on the condition that deficiencies of the Techy Store, or in your operation of Techy Stores are corrected; or
- (3) not grant you a Successor Franchise based on our determination that you and your owners have not substantially complied with this Agreement during its term.
 - B. Response Notice: If applicable, our Response Notice will:

- (1) describe the remodeling and/or expansion of the Techy Store and other improvements or modifications required to bring the Techy Store into compliance with the applicable specifications and standards for Techy Store; and
- (2) state the actions you have to take to correct operating deficiencies and the time period in which such deficiencies must be corrected.

If we elect not to grant a Successor Franchise, the Response Notice will describe the reasons for our decision. Your right to acquire a Successor Franchise is subject to your continued compliance with all of your terms and conditions of this Agreement through the date of its expiration, in addition to your compliance with the obligations described in the Response Notice.

- C. Deficiencies: If our Notice states that you must cure certain deficiencies of the Techy Store or its operation as a condition to the grant of a Successor Franchise, we will give you written notice of a decision not to grant a Successor Franchise unless you cure such deficiencies, not less than 90 days prior to the expiration of this Agreement. However, we will not be required to give you such notice if we decide not to grant you a Successor Franchise due to your breach of this Agreement during the 90 day period prior to its expiration. If we fail to give you:
- (1) notice of deficiencies in the Techy Store, or in your operation of the Techy Store, within 90 days after we receive your timely election to acquire a Successor Franchise; or
- (2) notice of our decision not to grant a Successor Franchise at least 90 days prior to the expiration of this Agreement, if such notice is required.

We may extend the term of this Agreement for such period of time as is necessary in order to provide you with either reasonable time to correct deficiencies or 90 days' notice of our refusal to grant a Successor Franchise.

15.3 Agreements/Releases.

If you satisfy all of the other conditions to the grant of a Successor Franchise, you and your owners agree to execute the form of franchise agreement and any ancillary agreements we are then customarily using in connection with the grant of Successor Franchises for Techy Stores. You and your owners further agree to execute general releases, in the form attached to this Franchise Agreement as Exhibit 1.2, of any and all claims against us and our shareholder, officers, directors, employees, agents, successors and assigns. Failure by you or your owners to sign such agreements and releases and deliver them to us for acceptance and execution within 60 days after their delivery to you will be deemed an election not to acquire a Successor Franchise.

15.4 Training and Refresher Programs.

Our grant of a Successor Franchise is also conditioned on the satisfactory completion by you (or and Operating Partner of yours approved by us) of any new training and refresher programs as we may reasonably/require. You are responsible for travel, living and compensation costs of attendees.

15.5 Fees and Expenses.

Our grant of a Successor Franchise is contingent on your payment to us of a Successor Franchise fee equal to twenty percent (20%) of our then current Initial Franchise Fee for an initial franchise. We must receive the fee from you when you sign the Successor Agreement.

16. TERMINATION OF THE FRANCHISE.

16.1 Defaults with No Right to Cure.

This Agreement will automatically Terminate upon delivery of our written notice of Termination to you in compliance with Article 20 (without further action by us and without opportunity to cure) if you (or any of your owners):

- A. fail to timely meet the site selection, development, opening and other requirements provided in Sections 3.1 (A) and 3.6, above;
- B. abandon or fail to operate your Techy Store for more than seven (7) consecutive calendar days, or lose the right to possession of the premises and do not relocate your Techy Store in accordance with this Agreement. If you have a sublease or an Operating Agreement with us or an Affiliate for the Premises, and you have been in full compliance with this Agreement and the Operating Agreement, we will use commercially reasonable efforts to identify an alternative location for you to operate a Techy Store. However, you must understand that neither we nor any Affiliate of ours represent or warrant to you that your ability to operate on the Premises or at any other location will be ongoing or for any specific duration;
- C. make any material misrepresentation or omission in your application for the Franchise, including (but not limited to) failure to disclose any prior litigation or criminal convictions (other than minor traffic offenses);
- D. are judged bankrupt, become insolvent, make an assignment for the benefit of creditors, are unable to pay your debts as they become due, or a petition under any bankruptcy law is filed by or against you (or any of your owners) or a receiver or other custodian is appointed for a substantial part of the assets of your Techy Store;
- E. are convicted of, or plead no contest to, a felony, or to any crime or offense that is likely to adversely affect the reputation of the Franchisee or any owner, your Techy Store, us or the goodwill associated with the Marks:
- F. engage in any misconduct which unfavorably affects the reputation of the Franchisee or any owner, your Techy Store, us or the goodwill associated with the Marks (including, but not limited to, child abuse, health or safety hazards, drug or alcohol problems, or permitting unlawful activities at your Store);
- G. make, or attempt to make, an unauthorized "transfer" as defined in this Agreement or surrender control without our prior written approval;
- H. make an unauthorized use of the Marks or any unauthorized copy, use or disclosure of any Confidential Information;
- I. violate any of the In Term or Post Term Restrictions against competition provided in Section 8.2, above (or any other person identified therein commits such a violation);
- J. commit any act or omission of fraud or misrepresentation, whether with respect to us, any of the Franchisor-Related Persons/Entities and/or any third party, including (but not limited to) any misrepresentation of Gross Revenues;

- K. have five (5) or more material customer complaints with respect to your Techy Store in any twelve (12) month period, whether or not resolved;
- L. fail to permit or cooperate with us or our designee in any audit or inspection or fail to retain (or to produce on request) any records required to be maintained by you;

16.2 Defaults with Right to Cure.

This Agreement will automatically Terminate on delivery of our written notice of Termination to you in compliance with Article 20 (without further action by us and without further opportunity to cure beyond that set forth in this Section):

- A. <u>10 Day Cure</u>. If within ten (10) calendar days after delivery of our written notice to you, you (or any of your owners) do not cure any:
 - (1) failure to maintain required insurance;
- (2) failure to correct any condition that, in our reasonable judgment, might pose a danger to public health and/or safety;
- (3) failure to report accurately Gross Revenues or fail to submit any other report due under this Agreement or any lease/sublease or Operating Agreement in accurate and complete form and when required;
- (4) failure to make payments of any amounts due us, any Franchisor-Related Person/Entity, any designee of ours and/or any supplier/creditor of yours and do not correct such failure(s);
- (5) failure to comply with any of the dispute resolution provisions of this Agreement, including (but not limited to) failure to pay/deposit any amounts or otherwise and/or unexcused failure to appear or respond to any dispute resolution proceedings.

With respect to items (A) 1 and/or (A) 2 above, we may require you to immediately cease all operations until such defaults are fully cured.

- B. <u>30 Day Cure</u>. If i) within ten (10) calendar days after delivery of our written notice to you, you have failed to commence steps to cure the following defaults and/or ii) within thirty (30) calendar days after delivery of our written notice to you, you (or any of your owners) do not cure any:
- (1) default under the lease or sublease or the Operating Agreement for your Techy Store within the applicable cure period set forth in the lease or sublease or the Operating Agreement. If such applicable cure period is less than thirty (30) days, then such applicable cure period shall apply, notwithstanding any cure period provided in this Article;
- (2) delinquency in your obligations to taxing authorities, landlords, equipment lessors, suppliers or others;
- (3) failure to comply with any other provision of this Agreement, any other agreement with us and/or any Affiliate of ours, or any specification, standard or operating procedure or rule prescribed by us in the Manuals or by other writing which does not provide for a shorter notice period.

If any such default under this Section 16.2 (B) cannot reasonably be corrected within such thirty (30) day period, then you must undertake diligent efforts within such thirty (30) day period to come into full compliance. You must furnish, at our request, proof acceptable to us of such efforts and the date full compliance will be achieved. In any event, all such defaults must be fully cured within ninety (90) days after delivery of the initial written notice to you of Termination.

16.3 Repeated Defaults.

This Agreement will automatically Terminate upon delivery of our written notice of Termination to you in compliance with Article 20 (without further action by us and without opportunity to cure) if you or any Affiliate has committed two (2) or more applicable defaults within any twelve (12) consecutive months, or three (3) or more applicable defaults within any twenty-four (24) consecutive months. An "applicable default" is a single breach of any obligation under this Agreement and/or the Manuals, or under any other agreement with us and/or any of our Affiliates, whether or not such default is cured, or is the same as or similar to a prior event of default.

16.4 Cross-Defaults.

Any default by you (or any owner or Affiliate of yours) under this Agreement may be regarded by us as a default under any other agreement between us (or any Franchisor-Related Persons/Entities) and you (or any owner or Affiliate of yours). Any default by you (or any owner or Affiliate of yours) under any other agreement or any other obligation between us (or any Franchisor-Related Persons/Entities) and you (or any owner or Affiliate of yours) may be regarded as a default under this Agreement. Any default by you (or any owner or Affiliate of yours) under any lease, sublease, Operating Agreement, loan agreement, or security interest relating to the Franchised Business may be regarded as a default under this Agreement, regardless of whether or not any such agreements are between you (or any owner or Affiliate of yours) and us (or any Franchisor-Related Persons/Entities.

17. RIGHTS AND OBLIGATIONS ON TRANSFER, REPURCHASE, TERMINATION AND/OR EXPIRATION OF THE FRANCHISE.

17.1 Payments of All Amounts Owed, etc.

You must pay all continuing license fees, marketing contributions and all amounts of any kind owed to us and/or any Franchisor-Related Persons/Entities within ten (10) days after the Repurchase, Termination or expiration of the Franchise, or from a later date when the amounts due can be determined.

17.2 Intellectual Property, Confidential Information, Trade Dress, etc.

After any Transfer, Repurchase, Termination or expiration of the Franchise:

- A. You agree to immediately and permanently discontinue your Techy business and any use of the Intellectual Property and/or the Confidential Information, as defined in Article 22, and will not use any similar or derivative marks, or materials, or colorable imitations of any of the Intellectual Property in any medium or manner or for any purpose;
- B. You will return to us or (at our option) destroy all software, Manuals, forms, materials, signage and any other items containing any Intellectual Property or Marks, or otherwise identifying or relating to a Techy Store (to the extent they have not been assigned in connection with an authorized Transfer or a Repurchase);

- C. You will take such actions as may be required to cancel all fictitious or assumed name or equivalent registrations relating to your use of any Mark which have not been assigned in connection with an authorized Transfer or a Repurchase;
- D. You will remove from the Premises any distinctive signage, physical and/or structural features associated with the Trade Dress of Techy Stores, so that the Premises are clearly distinguished from other Techy Stores and do not create any public confusion (to the extent they have not been assigned in connection with an authorized Transfer or a Repurchase);
- E. You agree not to identify yourself, or any business you may operate or in which you may become involved, or to advertise or promote yourself in any manner, as a present or former franchisee:
- F. You will furnish to us within thirty (30) days satisfactory evidence of your compliance with the obligations described in this Section 17.2 and in Section 17.3, below. If you operate any business using any of the Intellectual Property, Marks, Confidential Information or any aspect of the System, our remedies will include (but will not be limited to) recovery of the greater of (a) all profits earned by you in the operation of such business, or (b) all continuing license fees, advertising contributions and other amounts which would have been due if this Agreement remained in effect with you.

17.3 Telephone and Other Directory Listings, Internet Sites.

- A. You understand and agree that we own all telephone numbers, domain names, Internet addresses/sites and/or other communications services links (collectively, the "Numbers"), and any related directory listings/advertising, used in connection with the operation of Techy Store. We may, in our Business Judgment, require you to sign an assignment of such Numbers prior to training or at another time. After any Termination, Repurchase and/or expiration of the Franchise, you will promptly transfer, call-forward, discontinue or otherwise deal with the Numbers and any related directory listings/advertising as we direct. You agree to sign any documents and/or pay any amounts required by a telephone/communication services provider as a condition to our exercising any rights under this Section. By signing this Agreement, you irrevocably appoint us your attorney in fact to take any such actions regarding the Numbers and any related directory listings/advertising if you do not do so yourself within ten (10) days after the Termination, Repurchase or expiration of this Franchise. Such companies may accept this Agreement as conclusive evidence of our exclusive rights in such Numbers and related directory listings, web pages and advertising/marketing.
- B. If we choose at any time to be direct billed by a provider for any account for the Numbers and/or directory listings/advertising, you agree to pay us all amounts due such providers within ten (10) days of our written notice to you. If you fail on two or more occasions to pay any such amounts to us when due, then we may require you to maintain a deposit with us in an amount reasonably determined by us based upon usage history and other relevant factors.

17.4 Continuing Obligations.

A. All obligations and rights which expressly or by their nature survive the Transfer, Repurchase, Expiration or Termination of this Agreement will continue in full force and effect until they are satisfied or by their nature expire (including but not limited to indemnity, non-competition and confidentiality rights and obligations; obligations to pay and the provisions of Articles 19 and 21). These obligations continue notwithstanding any rejection of this Agreement in a bankruptcy proceeding or otherwise.

B. If this Agreement is Terminated because of a default of yours, you will not be released or discharged from your obligations, including payment of all amounts then due and other amounts which would have become due under this Agreement if you had continued in operation as a Franchisee for the full term. Our remedies will include (but are not limited to) the right to collect the present value of these amounts and to receive the benefit of our bargain with you, as well as to accelerate the balances of any promissory notes owed and to receive any other unpaid amounts owed to us or any Affiliates of ours. You and we agree that it would be commercially unreasonable and damaging to the integrity of the Techy system if a Franchisee could default and then escape the financial consequences of his contractual commitment to meet payment obligations for the term of the Agreement. You (and each of your owners/Affiliates) agree to sign a General Release if we choose in our Business Judgment to waive our rights to collect any amounts that would have become due if you had continued in operation as a Franchisee. This option of ours may be exercised at any time.

18. GRANT OF SECURITY INTEREST.

For valuable consideration, as security for the payment of all amounts owing or to be owed by you (and/or any Affiliate of yours) to us (and/or any Affiliate of ours) under this Agreement or any other agreements, and your performance of all obligations thereunder, you hereby grant to us a security interest in all proceeds of your Techy Store and in all of the assets, including equipment, furniture, fixtures and signs, used by, at or in connection with, your Techy Store and its related business and (the "Collateral"). You will not remove the Collateral or any portion thereof without our prior written consent. You represent and warrant that the security interest granted is prior to all other security interests in the Collateral except for (a) bona fide purchase money security interests and (b) the security interest granted to a third party in connection with your original financing for your Techy Store, if any. In connection with any request for our approval of a security interest, we will make commercially reasonable efforts to accommodate reasonable lender's requirements, including the subordination of our interests to the lender's and/or lessor's, as applicable, in our Business Judgment, bearing in mind the interests of the borrower, lender, ourselves and the System. On the occurrence of any event entitling us to Terminate this Agreement or any other agreement between the parties, or if we reasonably determine that we are not assured that your (and/or any Affiliates') obligations will be timely and fully paid and/or performed, we will have all the rights and remedies of a secured party under the Uniform Commercial Code of the State in which your Techy Store is located, including, without limitation, the right to take possession of the Collateral. You will execute and deliver to us financing statements and/or such other documents as we reasonably deem necessary to perfect our interest in the Collateral within ten (10) days of your receipt of such documents from us.

19. DISPUTE AVOIDANCE AND RESOLUTION – MANDATORY BINDING ARBITRATION.

For purposes of this Article 19, "you" includes all of your owners, Affiliates and their respective employees, and "we" includes all of the "Franchisor-Related Persons/Entities."

19.1 FUNDAMENTAL BUSINESS INTENTION TO MEDIATE AND/OR ARBITRATE, SEVERABILITY OF PROVISIONS, MODIFICATION OF AGREEMENT TO COMPLY WITH APPLICABLE LAW, FEDERAL ARBITRATION ACT GOVERNS.

A. It is your and our fundamental agreement not to engage in court proceedings, except in the rare instances specified in this Agreement, but rather to have disputes resolved through face-to-face meetings, mediation and binding arbitration. You and we agree our activities relating to the franchise relationship are in interstate commerce and that this Agreement is governed by the Federal Arbitration Act.

B. You and we view mediation and binding arbitration to be generally superior to court proceedings, as often being more time and cost efficient and allowing you and us to use mediators/arbitrators with substantial franchising experience, while understanding that costs to you and/or us of face-to-face meetings, mediation, arbitration and/or appeal of arbitration, as provided in this Agreement, may sometimes be greater than in civil litigation or may make it more difficult to proceed.

THEREFORE, IF ANY PROVISION OF THIS AGREEMENT IS DEEMED BY A COURT TO BE UNCONSCIONABLE, VOID, VOIDABLE, AGAINST PUBLIC POLICY, RESTRICTIVE OF CONSTITUTIONALLY OR STATUTORILY PROVIDED RIGHTS OR OTHERWISE UNENFORCEABLE, YOU AND WE INTEND AND AGREE (I) THAT SUCH PROVISION BE MODIFIED OR RESTRICTED BY THE ARBITRATOR OR COURT SO AS TO BE ENFORCEABLE OR (II), IF SUCH PROVISION CANNOT BE MODIFIED OR RESTRICTED, IT WILL BE SEVERED AND, TO THE MAXIMUM DEGREE POSSIBLE, THE REMAINING PORTIONS KEPT INTACT, VALID AND IN FULL FORCE AND EFFECT, SO AS TO ACCOMMODATE OUR FUNDAMENTAL AGREEMENT TO ENGAGE IN A FACE-TO-FACE MEETING, MEDIATION AND BINDING ARBITRATION/ARBITRATION APPEAL, AS PROVIDED IN THIS AGREEMENT.

- C. You and we reserve the right to challenge any law, rule or judicial or other construction which effectively varies, or renders without effect, any provision of this Agreement. If any limitation of rights is held unenforceable with respect to one party, then such limitation shall also be unenforceable as to other party.
- D. This Agreement will be deemed automatically modified to comply with governing law if such law requires a greater time period for notice of Termination of, or refusal to renew, this Agreement or otherwise.

19.2 FACE-TO-FACE MEETING, MEDIATION AND MANDATORY BINDING ARBITRATION.

You and we believe that it is important to resolve all disputes effectively and professionally and to return to business as soon as possible. You and we agree that the provisions of this Article 19 support these business objectives and, therefore, agree as follows:

- A. <u>Dispute Resolution Process</u>: Any litigation, claim, dispute, suit, action, controversy, or proceeding of any type whatsoever, including any claim for equitable relief and/or in which a party is acting as a "private attorney general," suing pursuant to a statutory claim or otherwise, between or involving you and us (and/or any Affiliate of either), on whatever theory and/or facts based, and whether or not arising out of this Agreement ("Claim"), will be exclusively processed in the following manner, except as expressly provided at Section 19.2 F., to the extent it is applicable and not contrary to governing law:
- (1) First, discussed in a face-to-face meeting held at a neutral location reasonably near your franchised location within thirty (30) days after either you or we give written notice to the other proposing such a meeting;
- (2) Second, if unresolved, submitted to non-binding mediation for a minimum of four (4) hours before (a) the American Arbitration Association ("AAA") or its successor or (b) Judicial Arbitration and Mediation Service ("JAMS") or its successor, if the AAA cannot conduct such mediation.

The mediator shall be a neutral person experienced in franchising. We will pay the costs of the first four (4) hours of any mediation, and no mediation is required to extend beyond such four (4) hour period. Any party may be represented by counsel and may, with permission of the mediator, bring persons appropriate to the proceeding.

- (3) Third, if unresolved, submitted to binding arbitration before and in accordance with the arbitration rules of (a) the American Arbitration Association ("AAA") or its successor or (b) Judicial Arbitration and Mediation Service ("JAMS") or its successor, if the AAA cannot conduct such arbitration. Arbitration may be filed prior to a face-to-face meeting and/or mediation, with such face-to-face meeting and/or mediation to follow as quickly thereafter as possible. Arbitration and mediation/ meeting may proceed concurrently. Franchisee agrees that any arbitration between Franchisee and Franchisor will be Franchisee's individual claim and that the claim or claims subject to arbitration shall not be arbitrated on a class-wide basis. The arbitrator must be a lawyer substantially experienced in franchising. Judgment on any arbitration award may be entered in any court having jurisdiction, thereof, or of Franchisor or of Franchisee, subject to the opportunity for appeal to a panel, as contemplated below. The arbitrator's award will be in writing.
- (4) Fourth, a final award by an arbitrator (there will be no appeal of interim awards or other interim relief) may be appealed within thirty (30) days of such final award. Appeals will be conducted before a three (3)-arbitrator panel appointed by the same organization as conducted the arbitration, each member of which must be a lawyer substantially experienced in franchising. The arbitration panel will not conduct any trial de novo or other fact-finding function. Such panel's decision will be in writing, may be entered in any court having jurisdiction and will be BINDING, FINAL AND NON-APPEALABLE. On request by either party, the arbitration panel will provide to all disputants a reasoned opinion with findings of fact and conclusions of law.
- (5) If required by applicable law for any arbitration provision (including the appeal process) to be enforceable (for example, to preserve constitutionally or statutorily provided rights), the arbitrator or a court will, as soon as possible, appropriately allocate between you and us the fees of the arbitrator(s) and/or his/her related organization, or require us to advance a portion of such fees subject to possible reimbursement, or otherwise address such issues so as to allow the mediation/arbitration to proceed and may adjust such allocations appropriately during the arbitration process for such purpose. In addition, you or we can choose to deposit all fees required by any arbitration organization, subject to later allocation between you and us, as provided above.
- (6) If the organizations specified by this Agreement to conduct any mediation and/or arbitration are unable or unwilling to conduct such proceeding(s), and the parties to the dispute cannot agree on an appropriate substitute organization or person to conduct such proceeding(s), then a court of competent jurisdiction shall designate an appropriate organization or person to conduct such proceeding(s).
- B. <u>Location</u>: Any mediation/arbitration (and any arbitration appeal) will be conducted exclusively at a neutral location in the county and state in which our then-current headquarters is located, which may change from time to time, and be attended by you and us, and/or designees authorized to make binding commitments on each of our respective behalf. You and we agree that the provisions of this Article will control, notwithstanding any language included in our franchise disclosure documents due to state requirements suggesting that the provisions of any section might be unenforceable due to a failure to have a meeting of the minds or otherwise. Neither you nor we have any expectation that the provisions of this (or any other) section will be unenforceable or that they will not be enforced. You understand and agree that one effect of this paragraph may be that mediation, arbitration and other related costs may be greater, and it may be more difficult for you to proceed, than if those proceedings took place in a location

near your residence or business. If any court determines that this provision is unenforceable for any reason, mediation/arbitration (and any arbitration appeal) will be conducted at a neutral location reasonably near your franchised location.

- C. Arbitration Authority: Arbitrators shall apply all applicable law, and a failure to apply the applicable law in accord with Section 19.8 C. below shall be deemed an act in excess of authority. The arbitrator shall be appointed within thirty (30) days of the filing of any demand for arbitration. Any arbitration hearing shall be held within ninety (90) days of the applicable appointment. The arbitrator shall decide all questions relating in any way to the parties' agreement to arbitrate, including but not limited to arbitrability, applicability, subject matter, timeliness, scope, remedies, and any alleged fraud in the inducement or claimed unconscionability. The arbitrator can issue summary orders disposing of all or part of a claim and provide for temporary restraining orders, preliminary injunctions, injunctions, attachments, claim and delivery proceedings, temporary protective orders, receiverships and other equitable and/or interim/final relief. Each party consents to the enforcement of such orders, injunctions, etc. by any court having jurisdiction. The subpoena powers of the arbitrator with respect to witnesses to appear at the arbitration proceeding shall not be subject to any geographical limitation. No award in arbitration will have any effect of preclusion or collateral estoppel in any other adjudication or arbitration.
- D. <u>Discovery, Effect of Awards, Compulsory Counter-Claims</u>: The disputants will have the same discovery rights as are available in civil actions under the state law selected in Section 19.8 C., except as modified by this Section. Discovery will relate exclusively to the relationship/agreements between you and us and not to our relationship/agreements with other Franchisees, depositions will be limited to key persons and either party can depose a maximum of three (3) such persons, including any experts, for a maximum period of seven (7) hours. On request by either party, the arbitrator shall provide a reasoned opinion with findings of fact and conclusions of law and the party so requesting shall pay the arbitrator's fees and costs connected therewith. Each disputant must submit or file any claim which would constitute a compulsory counter-claim within the same proceeding as the claim to which it relates as if the laws of the state law selected in Section 19.8 C. relating to such matters in litigation also governed such matters in arbitration. Subject to applicable law, any such Claim which is not submitted or filed in such proceeding will be forever barred. If any portion of this subsection is deemed to be unenforceable for any reason, it shall be modified or restricted, or severed, so as to comply with applicable law.
 - E. Fees and Costs: The parties will bear their own attorneys' fees and costs.
 - F. Claims Not Required to be Processed Under 19.2.A.:
- (1) Any party to this Agreement may forego the requirements of 19.2.A(1) and (2) above and seek interim, injunctive or expedited remedies (including temporary restraining orders) regarding the following Claims (or portions thereof), in either a judicial or arbitration proceeding, without regard to the adequacy of other potential legal remedies: (a) Enforceability (including specific performance) of rights and/or obligations with respect to the Marks and/or any of the Intellectual Property licensed to you and/or the ownership or proper use thereof; (b) Specific performance of any other rights and/or obligations under or relating to this Agreement including but not limited to obligations of Franchisee upon termination or expiration of the Franchise, and assignments of the Franchise and ownership of Franchisee; and/or (c) Matters concerning rights to real property including (but not limited to) any or all eviction and/or unlawful detainer actions, enforcement of any Collateral Assignment of Lease; (d) disputes relating to monies payable under the Franchise and related agreements, including, without limitation, any asset purchase agreement, area development agreement, operating agreement, and/or sublease, and
- (2) Claims (or portions thereof) asserted by any party to this Agreement relating primarily to enforceability of rights in and to, the Marks and/or any of the Intellectual Property

licensed to you and/or the ownership or proper use thereof[and not covered by (1) (a) above; i e. not seeking such remedies] will be exclusively subject to court action, you and we agreeing that since (a) the legal status/protection of the Marks and/or any Intellectual Property licensed to you and our other Franchisees is a core element of our System and of critical importance to all of our Franchisees and the future of our System, and (b) an adverse decision might affect all our Franchisees, it is critical that such matters should be only decided by means of a court trial and be subject to possible appellate court review. If the provisions of this subsection are unenforceable or would result in rendering this Article 19 (or any portion) unenforceable for any reason, the entire dispute shall be subject to mediation, arbitration and appeal of any arbitration decision, as provided in Section 19.2. Any action to compel a party's compliance with Section 19.2 must be consistent with Section 19.3.

- G. <u>Your and Our Intentions and Agreements</u>: You and we mutually agree and have expressly had a meeting of the minds that, notwithstanding any contrary provisions of state, provincial or other law:
- (1) all disputes relating to arbitrability will be governed only by the Federal Arbitration Act and the federal common law of arbitration and exclusive of state statutes and/or common law. including disputes about arbitration procedures, whether or not any particular claim, issue or otherwise is to be submitted to arbitration, and/or enforcement of any of the dispute resolution-related or other provisions of this Agreement; all such disputes will be exclusively decided by the arbitrator, including any assertion that this Agreement as a whole, or the arbitration or any other provision, of this Agreement, is unreasonable, oppressive, unlawful, invalid, void or voidable, procured by fraud or uneven bargaining power, unconscionable, part of a contract of adhesion, not subject to negotiation or not enforceable for any reason or similar claims;
- (2) all provisions of this Agreement can be fully enforced, including but not limited to those provisions relating to arbitration, venue, and choice of law; and
- Arbitration Act (9 U.S.C. § 1 et seq. and each knowingly waive all rights to a trial and select arbitration as the sole means to resolve disputes, except as expressly provided otherwise in this Agreement, understanding that arbitration may be less formal than a court or jury trial, may use different rules of procedure and evidence and that appeal is generally less available, and that the fees and costs associated with a face-to-face meeting, mediation and/or arbitration may be greater than in civil litigation, but still strongly preferring a face-to-face meeting, mediation and/or arbitration as provided in this Agreement believing that arbitration has overriding advantages from a business standpoint, such as an arbitrator being particularly familiar with franchising, as well as offering the possibility of lower costs and more rapid disposition of disputes.

19.3 Exclusive Jurisdiction and Venue.

Without in any way limiting or otherwise affecting your and our obligations under Section 19.2, above, you and we agree that any court proceedings (including any action to compel a party's compliance with Article 19) will be exclusively held in the United States District Court encompassing our then-current headquarters (the "Proper Federal Court"), which may change from time-to-time, subject to the following exceptions:

A. If a basis for federal jurisdiction does not exist, then any such proceeding will be brought exclusively before a court in the most immediate state judicial district encompassing our thencurrent headquarters and having subject matter jurisdiction (the "Proper State Court");

- B. Proceedings to remove or transfer a matter to the Proper Federal Court and/or to compel arbitration as contemplated by this Agreement may be brought in the court where the applicable action is pending and/or the Proper State or Federal Court; and
- C. Any action primarily with respect to any real and/or personal property (including any action in unlawful detainer, ejectment or otherwise) may be brought in any court of competent jurisdiction and/or the Proper State or Federal Court.

You and we consent to jurisdiction of the courts as specified above.

These and similar provisions are agreed on due to practical business concerns, such as (1) the fact that relevant business records and personnel key to a resolution of the dispute will generally be located at our then-current headquarters and (2) you and we have a shared interest in avoiding inconsistent legal resolutions in multiple jurisdictions, which could adversely affect the franchise system and its day-to-day management. You understand and agree that the effect of the above (and other) provisions of this Agreement may be, among other things, that your costs of litigation or otherwise in a location away from your residence or business may be greater, and it may be more difficult for you to proceed, than if those proceedings took place in a location nearer your residence or business. However, if a court or arbitrator determines that these provisions are unenforceable for any reason, they shall be severed and the venue for litigation will be as determined under applicable law.

19.4 Non-Retention of Funds.

Neither party has the right to offset or withhold payments of any kind owed or to be owed to the other against amounts purportedly due as a result of a dispute or otherwise, except as authorized by an arbitration award or as expressly provided in this Agreement.

19.5 Waiver of Punitive Damages, etc.

To the fullest extent permitted by law, you and we (and your and our Affiliates) mutually and knowingly waive all rights to or claim for punitive, exemplary, multiple or similar damages and agree that, in the event of any dispute, all recovery shall be limited to actual damages sustained by the injured party; provided that no such waiver or limitation shall apply to amounts owed under any indemnification obligation; and provided further that, to the extent required by applicable law, if you or we have any unwaivable rights to punitive, exemplary, multiple or similar damages under any statute or regulation, such rights will remain in effect. In any event, any constitutional and/or statutory limitations on punitive, exemplary, multiple or similar damages will apply, and any award by an arbitrator or court in excess of such limitations will be in excess of legal authority and void.

19.6 Survival of Obligations.

Articles 19 and 21 will continue in full force and effect after and notwithstanding the expiration, Termination, rescission, or cancellation of this Agreement for any reason, will survive and will govern any Claim for rescission or otherwise, as well as any Claim against, or with respect to, you, us, any Marketing and Technology Fund and/or any of the Franchisor-Related Persons/Entities and the non-solicitation, confidentiality, protection of the Marks, indemnity/hold harmless obligations, ongoing audit rights and all other Post-Termination Provisions provided in this Agreement shall also survive. Notwithstanding any bankruptcy or other proceeding, you and we wish to have the dispute avoidance and resolution provisions of this Agreement strictly enforced according to their terms, to the maximum extent allowable by law.

19.7 Attorneys' Fees and Costs.

The parties will each bear their own costs of enforcement and/or defense (including but not limited to attorneys' costs and fees), including those matters resolved pursuant to a settlement agreement between the parties.

19.8 Enforcement and Construction.

- A. Each portion of this Agreement is severable, and the remainder of the Agreement is fully enforceable. This Agreement is binding on the parties and their respective executors, administrators, heirs, assigns, and successors in interest, and will not be modified or supplemented except by means of a written agreement signed by both you and our President. Changes to the Manuals do not require any acceptance by you. No other officer, field representative, salesperson or other person has the right or authority modify this Agreement, or to make any representations or agreements on our behalf, and any such modifications, representations and/or agreements shall not be binding. Notwithstanding the fact that a party to this Agreement is or may become a party to a court action or special proceeding with a third party or otherwise, and whether or not such pending court action or special proceeding (1) may include common or other issues of law, fact or otherwise arises out of the same transaction or occurrence, or series of related transactions or occurrences, as any arbitration between or involving the parties to this Agreement, (2) involves a possibility of conflicting rulings on common issues of law, fact or otherwise, and/or (3) such pending court action or special proceeding may involve a third party who cannot be compelled to arbitrate, the agreement of the parties to this Agreement shall be enforced according to its terms and any party to this Agreement may bring an action to compel a face-to-face meeting, mediation and/or arbitration, you and we strongly preferring arbitration to court actions and wishing to have a single entity (the arbitrator) determine all issues of fact and law between or involving us, except as expressly provided otherwise in this Agreement.
- B. No waiver shall be effective unless in writing and signed by an authorized representative of the signing party. The rights and remedies provided in this Agreement are cumulative. Except as expressly provided in this Agreement, no party will be prohibited from exercising any rights or remedies provided under this Agreement or permitted under law or equity. If two (2) or more persons are at any time the Franchisee or the Franchisee owners, all of their obligations and liabilities under this or any other agreement with us and/or any Franchisor Associates will be joint and several.
- C. This Agreement and all matters relating to the relationship hereby created will be governed by, construed and enforced under the laws of the state in which your Techy Store is located, without regard to any conflict of laws, except to the extent of the applicability of the Federal Arbitration Act and related federal preemption and of the United States Trademark Act and other federal laws, and except that the provisions of any applicable state law regarding franchises, including any franchise or non-competition law, shall not apply unless jurisdictional requirements are met independently of this Section. This Agreement will be deemed automatically modified to comply with governing law if such law requires (1) a greater time period for notice of the Termination of this Agreement; or (2) the taking of some other action not described in this Agreement. Such modifications to this Agreement shall be effective only in such jurisdiction.
- D. You and we agree that none of the relationships formed in connection with the Agreement are in the nature of a "trust," "fiduciary" or similar special arrangement and that this is an ordinary commercial relationship.
- E. We have the right to elect in our sole discretion to not enforce (or to selectively enforce) any provision of this or any Agreement, standard or policy, whether with respect to you and/or any other franchisee or other person, in a lawful manner without liability.

19.9 Our Exercise of "Sole Discretion" and Other Choices; Express Agreement.

When we use the phrases "sole and absolute discretion" and/or "sole discretion" and whenever we exercise a right, prescribe an act or thing, or otherwise make a choice or use discretion, you and we agree that we have the express, unrestricted right to make decisions and/or take (or refrain from taking) actions, as we deem appropriate. We shall use our judgment in exercising such discretion based on our assessment of the interests we consider appropriate and will not be required to consider your individual interests or the interests of any other particular franchisee(s). We have this right even if a particular decision/action may have negative consequences for you, a particular franchisee or group of franchisees. You and we shall execute this Agreement in the belief that it is the basis for a long-term business relationship and should be enforced according to its express provisions. Neither you nor we have any expectation that the rights and obligations described herein will be defined or determined to be other than as expressly written. You and we agree that is not your and our intention or expectation that a third party use any doctrine and/or rule of interpretation to impose additional obligations on you and/or us.

19.10 Miscellaneous

- A. Section and Article headings are for convenience only and do not define, limit, or construe such provisions.
- B. References to a "controlling interest" are to a shareholder, membership or partnership interest, as applicable, which enables the holder(s) of such interest to determine the outcome of a decision making process for the applicable entity.
- C. This Agreement will be executed in multiple copies, each of which will be deemed an original.
- D. You and we have carefully reviewed and thought about each provision of this Agreement. Provisions shall be construed simply according to their fair meaning, without presumptions or inferences concerning terms or interpretation and not strictly against you or us.
- E. The rights and obligations of this Agreement run directly between you and us and are not intended to create any third-party beneficiary or similar rights or obligations, unless specifically expressed in this Agreement; except that the protections which apply to us relating to indemnification and/or releases will also benefit any past, current and/or future Franchisor-Related Persons/Entities, the Marketing and Technology Fund and the FAC and that the protections which apply to you will also apply to your Affiliates, unless your Affiliate is under a separate Franchise Agreement with us, in which case their protections under that Franchise Agreement will apply to them.

19.11 Intention to Arbitrate, Severability of Dispute Resolution Provisions, etc.

Irrespective of any statute, regulation, decisional law or otherwise, it is your and our fundamental agreement not to engage in any court proceedings (except as expressly provided for in the rare instances specified in this Agreement), viewing the dispute resolution mechanism established by this Agreement to be superior from a business standpoint and more likely to generate creative business-oriented solutions and compromise, and more accommodating to our business relationship and the needs of an evolving and diverse franchise system. Therefore, if any provisions of this Article 19 are deemed by a court to be unenforceable for any reason, you and we agree and intend that such provisions will be (a) modified so as to be enforceable or (b), if that cannot be done, severed and, in any event, any remaining portions of this

Article 19 shall remain in full force and effect. You and we agree that such remaining portions will still form an appropriate and complete dispute resolution mechanism.

20. NOTICES AND PAYMENTS.

All written notices and reports to be delivered by the provisions of this Agreement or of the Manuals will be deemed so delivered when delivered by hand, immediately on transmission by facsimile transmission or other electronic system, including e-mail or any similar means, one (1) business day after being placed in the hands of a commercial courier service for overnight delivery, or three (3) business days after placement in the United States Mail by Registered or Certified Mail, Return Receipt Requested, postage prepaid and addressed to us at Techy, LLC, 3000 South West 4th Avenue, Fort Lauderdale, FL 33315 (or our then-current headquarters), to the attention of the President, or Managing Partner and/or to you, at your Techy Store. Until your Techy Store has opened for business, we may send you notices at any address appearing in your application for a franchise or in our records. Any required payment or report not actually received by us during regular business hours on the date due will be deemed delinquent. Notice to any one Franchisee, or owner of the Franchisee, shall be deemed effective as to all Franchisees under this Agreement and all owners of the Franchisee(s). Any party may change its address for receipt of notices by providing prior written notice of such change to the other party.

21. ACKNOWLEDGMENTS AND REPRESENTATIONS, ENTIRE AGREEMENT, NO FIDUCIARY RELATIONSHIP, ETC.

- A. You and we agree that your and our relationship is not a fiduciary or similar special relationship, but rather is an ordinary commercial relationship between independent business people with arm's length dealings.
- B. You acknowledge that you (and each of your owners, if you are a Business Entity) have had the opportunity and been advised by us to have this Agreement and all other documents reviewed by your own attorney, and that you have read, understood, had an opportunity to discuss and agreed to each provision of this Agreement. You agree that you have been under no compulsion to sign this Agreement.
- C. You and we expressly acknowledge and agree that the provisions of Article 19, above, (whether relating to arbitration, mediation, venue, limitations on types of damages, and/or otherwise) may require you to travel to a distant location to resolve a dispute, expend additional funds, and/or raise challenges for you and/or us in prosecution of claims/actions. You and we view these provisions in the context of a diverse franchise system with both large and small, sophisticated and unsophisticated participants, and that requires uniformity and predictability. As such, you and we knowingly accept such provisions and limitations as justified by business necessities and representative of a reasonable balancing of your and our interests, and those of the System as a whole, and not as unfair or burdensome.
- D. You and we agree that this Agreement contains the final, complete and exclusive expression of the terms of your and our agreement [along with concurrently signed writings, such as but not limited to personal guarantees, Statement of Prospective Franchisee, addenda, exhibits, releases and any other related documents (collectively, the Related Documents)] and supersedes all other agreements and/or representations of any kind or nature; provided that nothing herein is intended to disclaim or require you to waive reliance on any representation made in the Disclosure Document. Any understandings, agreements, representations, or otherwise (whether oral or written) which are not fully expressed in this Agreement and the Related Documents are expressly disclaimed by you and us, including but not limited to any promises, options, rights of first refusal, guarantees, and/or warranties of any nature (excepting only the written representations made by you in connection with your application for this franchise). Neither

you nor we believe it to be fair or reasonable for the other party to have to deal with allegations about understandings, representations, etc. not fully expressed in writing in this Agreement.

- E. You specifically acknowledge that you have not received or relied on (nor have we or anyone else provided) any statements, promises or representations that you will succeed in the franchised business or at any location; achieve any particular sales, income or other levels of performance; earn any particular amount, including any amount in excess of your Initial Franchise Fee or other payments to us; or receive any rights, goods, or services not expressly set forth in this Agreement.
- F. You represent, warrant and agree that no contingency, prior requirement, or otherwise (including but not limited to obtaining financing) exists with respect to you fully performing any or all of your obligations under this Agreement. You further represent to us, as an inducement to our entering into this franchise relationship, that you have made no misrepresentations or material omissions in obtaining the Franchise.
- G. You acknowledge that you have not received or relied on (nor have we or anyone else provided, except as may have been contained in the Franchise Disclosure Document received by you):
 - (1) any sales, income or other projections of any kind or nature; or
- (2) any statements, representations, charts, calculations or other materials which stated or suggested any level or range of sales, income, profits or cash flow; or
- (3) any representations as to any profits you may realize in the operations of the Franchised Business or any working capital or other funds necessary to reach any 'break-even' or any other financial level.

If any such information, promises, representations and/or warranties have been provided to you, they are unauthorized and inherently unreliable. You agree to advise us of the delivery of any such information. You must not rely upon any such information, nor shall we be bound by it. We do not, nor do we attempt to, predict, forecast or project future performance, revenues or profits of any you or any franchisee. We are unable to reliably predict the performance of a Techy Store even operated by us, and certainly cannot predict results for your Techy Store.

You understand and agree that Franchisees are separate and distinct from us and are independently owned and operated and that while we may encourage you to speak with such Franchisees in connection with your evaluation of this franchise opportunity, they do not act as our agents or representatives in providing any information to you and we will have no obligations or liabilities with respect to (and you should not rely on) any information, opinions or otherwise they may provide to you.

- H. You acknowledge and agree that the success of the business venture contemplated to be undertaken by you is speculative and will be dependent on your personal efforts, and success is not guaranteed; and you further acknowledge that we have just recently begun franchising, we are not an experienced franchisor, our franchise system is relatively unproven and our business model is still under development. You acknowledge and represent that you have entered into this Agreement and made an investment only after making an independent investigation of the opportunity, including having received a list with your Franchise Disclosure Document of others currently operating, or who have operated, our franchises.
- I. You acknowledge that you (and each of your owners) has received, fully read and understood, and all questions have been answered regarding, i) a copy of our Franchise Disclosure Document with all exhibits at least fourteen (14) calendar days prior to signing any binding documents or paying any sums (whichever occurred first), and ii) a copy of this Agreement and all other agreements

complete and in form ready to sign at least seven (7) calendar days prior signing any binding documents or paying any sums (whichever occurred first).

- J. You understand, acknowledge and agree that i) we may have offered franchises in the past, may currently be offering franchises and/or may offer franchises in the future, on economic and/or other terms, conditions and provisions which may significantly differ from those offered by this Agreement and any related documents and ii) we may, from time to time, deal with our Franchisees on differing economic and/or other terms (including making special arrangements for payments of amounts due, waiving defaults and/or otherwise) to suit individual business circumstances, in each case in our Business Judgment and without being required to offer similar terms to other Franchisees, such flexibility being a practical necessity to respond to distinct business situations.
- K. You understand that we are relying on you to bring forward in writing at this time any matters inconsistent the representations contained in this Article 21. You agree that if any of the statements or matters set forth in this Article 21 are not true, correct and complete that you will make a written statement regarding such next to your signature below so that we may address and resolve any such issue(s) at this time.
- L. You acknowledge and agree that the officers, directors, employees, and agents of the Franchisor act only in a representative capacity and not in an individual capacity, and that no other persons and/or entities other than the Franchisor has or will have any duties or obligations to you.
- M. You acknowledge and agree that in fulfilling its obligations to negotiate any lease or Operating Agreement for the Premises, the Franchisor will negotiate certain terms that will be protective of Franchisor and grant to Franchisor certain rights and remedies against you and that some of those terms and conditions are set out in the Franchise Lease Addendum attached hereto as Exhibit 3.2. If you have a sublease or an Operating Agreement with us or an Affiliate for the Premises, you further acknowledge that we/the applicable Affiliate have the unrestricted right to exercise any and all rights under the lease for the Premises, including any rights to termination, assignment, non-renewal and otherwise. You also acknowledge that we do not control any actions taken by the landlord in connection with the Premises or otherwise. You acknowledge and agree that neither we nor any Affiliate of ours shall have any liability or obligation to you if such lease is cancelled, transferred, terminated or not renewed, with or without cause, unless solely resulting from our/our Affiliate's exclusive breach of the lease. You further acknowledge that you (and each of your owners, if you are a Business Entity) have had the opportunity and been advised by us to have this Agreement and all Exhibits reviewed by your own attorney, and that you have read, understood, had an opportunity to discuss and have agreed to each provision of this Agreement. You agree that you have been under no compulsion to sign this Agreement.

22. **DEFINITIONS.**

The following definitions apply to terms used this Agreement:

"Affiliate" - Any person or entity which controls, is controlled by or is under common control with one or more persons or entity/entities; in addition, as to the Franchisee, any owner of any interest in the Franchisee or the Franchise, any employee or agent of the Franchisee, and/or any independent contractor performing functions for, or on behalf of, the Franchisee, and any entity controlled by any of the foregoing.

[&]quot;Agreement" - This Franchise Agreement.

- "Attorneys' Fees" Includes, without limitation, legal fees, whether incurred in preparation of the filing of any written demand or claim, action, hearing, arbitration, or other proceeding to enforce the obligations of this Agreement, or during any such proceeding, plus all costs incurred in connection therewith.
- "Brand" The Techy brand, as applied to various goods and/or services as authorized by us from time to time.
- "Business Entity" Includes a corporation, partnership, joint venture, limited liability company, limited partnership, or other form of business recognized in any jurisdiction. If you are a Business Entity, then we may require each of your owners in our Business Judgment to guaranty your performance. Our current form of Owners Guaranty is attached as Exhibit 1 of this Franchise Agreement.
- "Business Judgment" Means that we are allowed to exercise our judgment however we consider to be appropriate in our sole and absolute discretion, without any limitation. You and we agree that when in this Agreement we describe instances in which we may exercise Business Judgment, we must and do have the unrestricted right to make decisions and/or take (or refrain from taking) actions. We have this right even if a particular decision/action may have negative consequences for you, a particular franchisee or group of franchisees. You understand and agree that the exercise of Business Judgment is critical to our role as Franchisor of the System and to our goals for its continuing improvement. This is a defined term for the purposes of this Agreement and is not intended to incorporate principles related to the application of the business judgment rule in a corporate law context.
- "Customary Representations, Warranties and Agreements" Includes commitments generally made by a transferor in connection with a transfer of a business and/or related assets, including but not limited to: representations as to ownership, condition and title to stock and/or assets, liens and encumbrances relating to the stock and/or assets, validity of contracts, and liabilities, contingent or otherwise, relating to the business/assets/entity to be acquired; full indemnification obligations and non-competition covenants by the transferor and each Affiliate, substantially similar to those required in Sections 7.4 and 8.2 of this Agreement; the delivery at closing of instruments transferring good and merchantable title to the assets purchased, free and clear of all liens and encumbrances (other than liens and security interests acceptable to us in our Business Judgment), and demonstrating that all sales, transfer and/or similar taxes are to be paid by the transferor through escrow if we so require; the transfer at closing of all licenses and permits which may be assigned or transferred.
- "Designated Equipment" Equipment that meets our requirements and which you must obtain and use in the operation of your Techy Store and if applicable, Techy Café and/or approved Wrapped Techy Van.
- "Fair Market Value" A price determined in a manner consistent with reasonable depreciation of any leasehold improvements, as applicable, and the items purchased. Fair Market Value does not include any factor or increment for any goodwill related to, or for, the Intellectual Property or any of its components such as the Marks, since you don't own the Intellectual Property. If you and we are unable to agree on the Fair Market Value of any assets, then it will be determined by an independent appraiser selected by you and us, or an arbitrator if you can we can't agree.
- "**Franchise**" The right to operate a single Techy Store and if applicable, Techy Café at the Premises under the terms of this Agreement and if applicable, operation of a Wrapped Techy Van.
- "Franchise Advisory Council" or "FAC" The advisory group selected (or which may be selected) in accordance with this Agreement, which shall provide Input as provided in this Agreement and as we may request from time to time.

"Franchised Business" - The business operations conducted by, at or in connection with your Techy Store.

"Franchisor-Related Persons/Entities" - Techy, LLC, the Marketing and Technology Fund, the FAC and each and all of the following, whether past, current and/or future: each and all company(ies) and/or person(s) acting through, in concert with us and/or any of the foregoing, and/or as Affiliates of ours and/or of any of the foregoing; each and all of the partners, shareholders, officers, directors, agents, attorneys, accountants, and/or employees of us and/or any of the foregoing; and each and all of the predecessors, successors and/or assigns of us and/or any of the foregoing.

"General Release" - A general release, in the then-current form prescribed by us at the time such release is to be delivered, of any and all claims, liabilities and/or obligations, of any nature whatsoever, including those existing as of, and/or arising before, the date of any such release, however arising, known or unknown, whether against us and/or any or all of the Franchisor-Related Persons/Entities, and whether by you, any owner of you (if you are or become a Business Entity) and/or any Affiliate of any of the foregoing. A copy of our general releasing language as currently used by us (which is subject to change) is attached as Exhibit 1.2 and is approved by you.

"Good Standing" - You are in "Good Standing" if you (and each of your owners and Affiliates) are not in default of any obligation to us and/or any of the Franchisor-Related Persons/Entities, whether arising under this Agreement or any other agreement between you (and each of your owners and Affiliates) and us (and/or any of the Franchisor-Related Persons/Entities), the Manuals or other System requirements (collectively, the "Obligations"); provided that you are not in Good Standing if you have been in default of any Obligations and such defaults are incurable by nature and/or part of a series of repeated defaults as defined in this Agreement.

"Gross Revenues" - Gross Revenues includes all charges and/or revenues which are, or could be, received or earned by you (and/or any Affiliate):

- A. by, at or in connection with your Techy Store and if the approved addendum is signed by us, Techy Café, and if the approved addendum is signed by us, Wrapped Techy Van;
- B. relating to the kinds of goods or services available now or in the future through a Techy Store and if the appropriate addendum is signed by us, Techy Café, and if the approved addendum is signed by us, Wrapped Techy Van, and/or distributed in association with the Marks or the Franchise System;
- C. relating to the operation of any Similar Business;
- D. with respect to, any tenants and/or subtenants of yours on the Premises (including rent and other lease or Operating Agreement payments); and/or
- E. with respect to any co-branding activities.

All sales and/or billings, whether collected or not, will be included in Gross Revenues, with no deduction for credit card or other charges. Gross Revenues does not include sales tax collected and paid when due to the appropriate taxing authority and actual customer refunds, adjustments and credits.

"Immediate Family" - With respect to any person, "Immediate Family" includes that person's spouse and/or domestic partner and each of their respective parents, guardians, grandparents, siblings, children, grandchildren, aunts, uncles, cousins, nieces and/or nephews.

- "Input" Advice and suggestions regarding specified matters. When we receive Input from the FAC or any other franchisee group we will retain the ultimate decision-making authority and responsibility for all matters for which Input is sought. FAC (or any other franchisee group) Input, votes or other collective actions will not be binding on us unless we have otherwise agreed in writing. FAC (or any other franchisee group) approval or consent will not be required as a pre-condition to any decision and/or action we may take.
- "Intellectual Property" Includes, regardless of the form or medium involved, (i) all Techy Software, including the data and information processed or stored thereby; (ii) the Manuals and all other directives, policies or information we issue from time to time; (iii) all customer relationships and information; (iv) the Marks; (v) all Confidential Information and our trade secrets; and (vi) all other proprietary, copyrightable and/or trade secret information and materials developed, acquired, licensed or used by us in our operation of the System.
- "Manuals" Specifications, standards, policies and procedures prescribed by us and published to you in any media (including electronic) and which are to be followed in the operation of your Techy Store and if applicable, the Techy Café or Wrapped Techy Van, as they may be changed or eliminated by us in our Business Judgment.
- "Marketing and Technology Fund" The fund established and defined under Section 11.1.
- "Marks" The trademarks, service marks and other commercial symbols now and/or in the future owned by (or licensed to) us to identify the services and/or products offered by Techy Stores or if applicable, the Techy Cafés or Wrapped Techy Vans, including (but not limited to) "Techy Repairs & Smart Home Installs Powered by DrPhoneFix®" logo, the Trade Dress and other logos and identifiers designated by us from time to time.
- "Post Termination Provisions" Those promises contained in this Agreement that survive its expiration, Transfer, Repurchase, or Termination for any reason, including without limitation the confidentiality, non-competition, indemnification, and dispute resolution and other provisions contained in Articles 19, 20 and 21.
- "Premises" The facility in which you will operate a single Techy Store.
- "**Products**" and "**Services**" Goods, products and services designated by us from time to time for use, sale or otherwise to be provided (and/or used) at and/or from your Techy Store and/or in association with the Marks.
- "Repurchase" Repurchase includes (but is not limited to) any acquisition by us (and/or any of the Franchisor-Related Persons/Entities) of your rights in and/or to any of the following: i.) this Agreement; ii.) the Franchise; iii.) the ownership of the Franchisee; iv.) your Techy Store; v.) your Techy Café; vii) Wrapped Techy Van; or vi.) any lease or Operating Agreement or assets associated with any of the foregoing.
- "Techy Store" / "Techy Café" / "Wrapped Techy Van" The Techy Store or if applicable, the Techy Café and/or Wrapped Techy Van, you are franchised to operate by this Agreement.
- "Similar Business" Any enterprise that offers, is otherwise involved in, or deals with any goods, products and/or services, which are substantially similar to those goods, Products and/or Services now or in the future authorized by us to be offered at or from Techy Stores, Techy Cafés, Wrapped Techy Vans (including any such enterprise and/or entity awarding franchises or licenses to operate or be involved with any such

business). Our receipt of any continuing license fees with respect to any Similar Business is not an approval of your involvement with any Similar Business.

"Special Accounts" - Classes of special customers (which may include national accounts, other large businesses, government agencies, and/or otherwise) as designated by us from time to time in our Business Judgment.

"System" - The distinctive format and method of doing business developed and used for the operation of a Techy Store, and subject to change by us at any time in our Business Judgment.

"System Standards" - Standards prescribed by us in our Business Judgment from time to time, in the Manuals or elsewhere, for the operation, marketing and otherwise of Techy Stores.

"Terminate" or "Termination" - "Terminate" or "Termination" when used in this Agreement means the Termination or cancellation of your rights and our obligations under this Agreement for any reason before the initial term expires. All of our rights are not cancelled on Termination since you have certain obligations that survive the ending of the Agreement in any manner, such as, but not limited to certain promises regarding non-competition, confidentiality and indemnity. Both of us are bound by the dispute resolution provisions (Article 19) this Agreement, even after the Agreement is ended for any reason.

"Territory" - The geographic area described in Exhibit 2.2.

"Trade Dress" - The Techy Store design and image authorized by us and subject to change by us at any time and in our Business Judgment.

"Techy Store" - A "Techy Store" means a "brick and mortar" retail facility located in a free-standing building or a shopping center accessible to the general public and using the Marks and Techy Repairs & Smart Home Installs Powered by DrPhoneFix® System whether Company-owned or franchised to a third party.

"Transfer" - Defined in Section 14.2.

"Us," "We," "Our" or "Franchisor" – Techy, LLC, a Florida Limited Liability Company.

"You," "Your," or "Franchisee" - The parties signing this Agreement as Franchisee. (If there is more than one Franchisee, each is jointly and severally obligated under this Agreement and all other agreements with us and/ or Franchisor-Related Persons/Entities). The term "you" is applicable to one or more persons or a Business Entity, as the case may be.

23. SPOUSAL CONSENT.

If you have obtained our approval and the Franchisee is a corporation, partnership, limited liability company or other form of legal entity, then the spouse of each of your owners, or, for, an individual(s), or subsequent to execution hereof you marry or you assign this Agreement to an individual(s), then in either event, the spouse(s) hereby jointly and severally personally and unconditionally guarantee(s) without notice, demand or presentment, the payment of all of your monetary obligations under this Agreement as if each were an original party to this Agreement in his or her individual capacity. All such spouses further agree to be bound by restrictions upon your activities upon transfer, termination or expiration of this Agreement as if each were an original party to this Agreement in his or her individual capacity. All such spouses must execute a spousal consent in the form attached hereto as Exhibit 20. In the event of divorce

and re-marriage, or subsequent marriage, you covenant and agree to provide Franchisor with a properly executed spousal consent and guarantee, in the form prescribed by Franchisor.

[Signatures on page that follows.]

IN WITNESS WHEREOF, you and we have executed and delivered this Agreement on the day and year first above written.

THIS AGREEMENT WILL NOT BECOME EFFECTIVE UNLESS AND UNTIL SIGNED BY THE PRESIDENT OR A VICE PRESIDENT OF FRANCHISOR. NO FIELD REPRESENTATIVE OR OTHER PERSON IS AUTHORIZED TO EXECUTE THIS AGREEMENT FOR FRANCHISOR.

WITNESS	FRANCHISEE (Individual)
(Signature)	(Signature)
	(Printed Name)
WITNESS	
(Signature)	(Signature)
(Printed Name)	(Printed Name)
	FRANCHISEE (Corp., LLC or Partnership)
	(Legal Name of Franchisee Entity)
WITNESS	a (Jurisdiction of Formation) (Corp., LLC or Partnership)
(Signature)	By:
(Printed Name)	(Printed Name)
	Its:(Title)
	FRANCHISOR TECHY, LLC
WITNESS	
	By:
	Its:

EXHIBIT 1 OWNER'S GUARANTY AND ASSUMPTION OF OBLIGATIONS

In consideration	of, and as an inducement to	o, the execution by T	Techy, LLC	, a Florida Limited
Liability Company ("Fra	anchisor") of a franchise ag	greement of even da	te herewith	(the "Agreement")
between Franchisor and _		_, a(n)		(state of formation)
	_(type of entity: LLC, LLP,	corporation, etc.) (the	e "Business I	Entity Franchisee"),
each of the undersigned h	nereby personally and uncon-	ditionally, jointly and	l severally:	

- (1) guarantees to Franchisor, its Affiliates, the Franchisor-Related Persons/Entities (as defined in the Agreement) and each of their successors and assigns, for the term of the Agreement, and for any renewal/successor franchise term, and thereafter as provided in the Agreement, that the Business Entity Franchisee will punctually pay and perform, each and every undertaking, agreement and covenant set forth in the Agreement, as currently set forth and as amended and/or otherwise changed in the future, including any successor franchise agreement;
- (2) agrees to be personally bound by, and personally liable for, the breach of, each and every provision in the Agreement (including all confidentiality, non-competition, indemnity and Post Termination Provisions), as currently set forth and as amended or otherwise changed in the future, including any successor franchise agreement; and
- (3) agrees to be personally bound by, and personally liable for, each past, current and/or future obligation of the Business Entity Franchisee to Franchisor, its Affiliates, the Franchisor-Related Persons/Entities and each of their successors and assigns.

The undersigned intending that the guarantees and other obligations herein be unqualifiedly general and without limitation in scope, nature and/or effect. Franchisor, and/or its Affiliates, the Franchisor-Related Persons/Entities and each of their successors and assigns, need not bring suit first against the undersigned in order to enforce this guarantee and may enforce this guarantee against any or all of the undersigned as it chooses in its/their sole and absolute discretion.

Each of the undersigned waives: presentment, demand, notice of demand, dishonor, protest, nonpayment, default and all other notices whatsoever, including (without limitation): acceptance and notice of acceptance, notice of any contracts and/or commitments, notice of the creation and/or existence of any liabilities under the Agreement or otherwise and of the amounts, terms or otherwise thereof; notice of any defaults, disputes or controversies between the Franchisor and the Business Entity Franchisee or otherwise, and any settlement, compromise or adjustment thereof; any right the undersigned may have to require that an action be brought against Franchisor, Business Entity Franchisee or any other person as a condition of liability, and any and all other notices and legal or equitable defenses to which he or she may be entitled.

Each of the undersigned consents and agrees that:

- (1) his or her direct and immediate liability under this guaranty will be joint and several;
- (2) he and/or she will render any payment or performance required under the Agreement on demand if the Business Entity Franchisee fails or refuses to do so punctually;
- (3) such liability will not be contingent or conditioned on pursuit by Franchisor or otherwise of any remedies against the Business Entity Franchisee or any other person;

- (4) such liability will not be diminished, relieved or otherwise affected by any extension of time, credit or other indulgence which Franchisor or otherwise may from time-to-time grant to the Business Entity Franchisee or to any other person, including, without limitation, the acceptance of any partial payment or performance, or the compromise or release of any claims, none of which will in any way modify or amend this guaranty, which will be continuing and irrevocable during the term of the Agreement and any renewal/successor franchise term;
- (5) the liabilities and obligations of the undersigned, whether under this document or otherwise, will not be diminished or otherwise affected by the Termination, rescission, expiration, renewal, award of a successor franchise, modification or otherwise of the Agreement;
- (6) terms not defined in this document shall have the meanings assigned in the Agreement; and
- (7) the provisions of Articles 18 through 22 of the Agreement are incorporated in and will apply to this document as if fully set forth herein and shall apply to any dispute involving the Franchisor, its Affiliates, the Franchisor-Related Persons/Entities and each of their successors and assigns and any of the undersigned.

In connection with such guarantee and the Franchisor (a) not requiring that the Franchise be in the name of one or more of the Guarantors and/or (b) not requiring the payment of a full transfer fee in connection with any related transfer from the undersigned to the Business Entity Franchisee, each of the undersigned hereby grants a General Release of any and all claims, liabilities and/or obligations, of any nature whatsoever, however arising, known or unknown, against the Franchisor, its Affiliates, the Franchisor-Related Persons/Entities and each of their successors and assigns.

IN WITNESS WHEREOF, each of the undersigned has here unto affixed his or her signature on the same day and year as the Agreement was executed.

PERCENTAGE OF OWNERSHIP

		OF BUSINESS ENTITY FRANCHISEE
	(Signature)	%
(Printed Name)		
	(Signature)	%
(Printed Name)		
	(Signature)	%
(Printed Name)		
	(Signature)	%
(Printed Name)		

GUARANTOR(S)

Business Entity Franchisee:		
	, a	corporation/LLC
By(Signature)		
(Printed Name)		
Its		
Date:		

EXHIBIT 1.2 CURRENT FORM OF RELEASING LANGUAGE

(SUBJECT TO CHANGE BY FRANCHISOR)

Release - General Provisions. The Franchisee(s), jointly and severally, hereby release and forever discharge each and all of the Franchisor-Related Persons/Entities (as defined below) of and from any and all causes of action, in law or in equity, suits, debts, liens, defaults under contracts, leases, or any Operating Agreement, agreements or promises, liabilities, claims, demands, damages, losses, costs or expenses, of any nature whatsoever, howsoever arising, known or unknown, fixed or contingent, past or present, that the Franchisee(s) (or any of them) now has or may hereafter have against all or any of the Franchisor-Related Persons/Entities by reason of any matter, cause or thing whatsoever from the beginning of time to the date hereof (the "Claims"), it being the mutual intention of the parties that this release be unqualifiedly general in scope and effect and that any Claims against any of the Franchisor-Related Persons/Entities are hereby forever canceled and forgiven.

THE FRANCHISEE(S) ACKNOWLEDGE THAT THEY ARE FAMILIAR WITH THE PROVISIONS OF CALIFORNIA CIVIL CODE SECTION 1542, WHICH PROVIDES AS FOLLOWS:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM MUST HAVE MATERIALLY AFFECTED HIS SETTLEMENT WITH THE DEBTOR."

THE FRANCHISEE(S), BEING AWARE OF THIS CODE SECTION, HEREBY EXPRESSLY WAIVE ALL OF THEIR RIGHTS THEREUNDER AS WELL AS UNDER ANY OTHER STATUTES OR COMMON LAW PRINCIPLES OF SIMILAR EFFECT OF ANY APPLICABLE JURISDICTION, INCLUDING, WITHOUT LIMITATION, CALIFORNIA AND/OR JURISDICTIONS OF FRANCHISEE(S)' RESIDENCE AND LOCATION OF FRANCHISED UNIT, provided, that if this Release is given in connection with the award of a franchise, then this release shall not apply to Claims relating to the offer and sale of such franchise under the California Franchise Investment Law or any rule or order issued thereunder.

The Franchisee(s) expressly assume the risk of any mistake of fact or fact of which they may be unaware or that the true facts may be other than any facts now known or believed to exist by Franchisee(s), and it is the Franchisee(s) intention to forever settle, adjust and compromise any and all present and/or future disputes with respect to all matters from the beginning of time to the date of this document finally and forever and without regard to who may or may not have been correct in their understanding of the facts, law or otherwise. All releases given by the Franchisee(s) are intended to constitute a full, complete, unconditional and immediate substitution for any and all rights, claims, demands and causes of action whatsoever which exist, or might have existed, on the date of this document. The Franchisee(s) represent and warrant that they have made such independent investigation of the facts, law and otherwise pertaining to all matters discussed, referred to or released in or by this document as the Franchisee(s), in the Franchisee(s) independent judgment, believe necessary or appropriate. The Franchisee(s) have not relied on any statement, promise, representation or otherwise, whether of fact, law or otherwise, or lack of disclosure of any fact, law or otherwise, by the Franchisor-Related Persons/Entities or anyone else, not expressly set forth herein, in executing this document and/or the related releases.

Franchisee(s) Initials:	

No Assignment or Transfer of Interest. The Franchisee(s) represent and warrant that there has been, and there will be, no assignment or other transfer of any interest in any Claims that the Franchisee(s) may have against any or all of the Franchisor-Related Persons/Entities, all Claims having been fully and finally extinguished, and the Franchisee(s) agree to forever indemnify and hold the Franchisor-Related Persons/Entities harmless from any liability, claims, demands, damages, losses, costs, expenses or attorneys' fees incurred by any of the Franchisor-Related Persons/Entities as a result of any person asserting any interest in any of the Claims and/or any voluntary, involuntary or other assignment or transfer, or any rights or claims under any assignment, transfer or otherwise. It is the intention of the parties that this indemnity does not require payment by any of the Franchisor-Related Persons/Entities as a condition precedent to recovery against the Franchisee(s) under this indemnity.

Franchisee(s) Initials:
Attorneys' Fees. If the Franchisee(s), or anyone acting for, or on behalf of, the Franchisee(s) or claiming to have received, by assignment or otherwise, any interest in any of the Claims, commence, join in, or in any manner seek relief through any suit (or otherwise) arising out of, based upon or relating to any of the Claims released hereunder or in any manner asserts against all or any of the Franchisor-Related Persons/Entities any of the Claims released hereunder, the Franchisee(s) agree to pay all attorneys' fees and other costs incurred by any of the Franchisor-Related Persons/Entities in defending or otherwise responding to said suit or assertion directly to the Franchisor-Related Persons/Entities incurring such costs.
Franchisee(s) Initials:
"Franchisor-Related Persons/Entities." Franchisor, Franchisor's affiliates, any advertising or marketing fund, any Franchisee Advisory Group and each of the following, whether past, current or future: companies and/or persons acting through and/or in concert with us and/or with any of the foregoing; partners, shareholders, officers, directors, agents, attorneys, accountants, and/or employees of ours and/or of any of the foregoing; and predecessors, successors and/or assigns of ours and/or of any of the foregoing.
Franchisee(s) Initials:
Date of Releases, Joint and Several Liability. The releases granted hereunder shall be deemed effective as of the date hereof. The liabilities and obligations of each of the Franchisee(s) (and any other person/entity providing releases to the Franchisor-Related Persons/Entities) shall be joint and several. Franchisee(s) Initials:
Printed Name:
Signature:

EXHIBIT 2.2 PROTECTED TERRITORY & SITE SELECTION

The "Protected Territory" is as follows:

The Protected Territory is a one-mile radius from the Approved Site or an area having a day time population of 50,000 (U.S. Census Bureau) people, whichever is the lesser area.

Note: Boundary lines include only the area within the boundary line and extend only to the middle of the boundary demarcation (for example, only to the middle of a street or highway.) You have no rights under this Agreement or otherwise with respect to a facility on the other side of the boundary line, street or highway or otherwise, and no matter how close to such boundary a facility may be, regardless of the distance from, impact on, or vicinity of, your Techy Store or the number of Techy Stores, other outlets or otherwise in any area or market. Your rights are limited as set forth in the Franchise Agreement.

The Approved Site is as follows:	
FRANCHISOR:	FRANCHISEE:
TECHY, LLC	
By:	Signature
Its:	Printed Name
Date:	Date:
	Signature
	Printed Name
	Date:

THIS COLLATERAL ASSIGNMENT OF LEASE (this "Assignment") is entered into as of

EXHIBIT 3.2 B COLLATERAL ASSIGNMENT OF LEASE

20 between ("Franchisee") and Techy
LLC, a Florida Limited Liability Company ("Franchisor.")
Subject to the provisions hereof, the Franchisee, to secure its obligations to the Franchisor under
the franchise agreement between the Franchisor and the Franchisee for the operation of a Techy Stor
franchise, dated, 20_ (the "Franchise Agreement"), hereby assigns, transfers an
sets over unto Franchisor [and/or such person(s)/entity(ies) as Franchisor may from time-to-time designate
all of Franchisee's right, title and interest, whether as tenant or otherwise, in, to and under that certain leas
(the "Lease"), a copy of which is attached to this Assignment, dated, 20, between Franchise
and ("Landlord"), respecting that property commonly known a
(the "Premises"). Th
Franchisor shall have no liabilities or obligations of any kind arising from, or in connection with, thi
Assignment, the Lease or otherwise (including, but not limited to, any obligation to pay rent and/or other
amounts) until and unless the Franchisor, in its sole and absolute discretion, takes possession of th
Premises pursuant to the terms hereof and expressly (and in writing) assumes the rights and obligations of
Franchisee under the Lease, the Franchisor only being responsible for those obligations accruing after th
date of such assumption.

The Franchisee agrees to indemnify and hold harmless the Franchisor from and against all claims and demands of any type, kind or nature made by the Landlord or any third party that arise out of or are in any manner connected with the Franchisee's use and occupancy of the Premises subject to the Lease.

The Franchisee represents and warrants to the Franchisor that the Franchisee has full power and authority to assign the Lease and its interest in the Lease, as evidenced by the execution of the Franchise Addendum to Lease Agreement attached to the Lease and incorporated therein by reference.

The Franchisor will not take possession of the Premises until and unless the Franchisee defaults (and/or until there is a termination, cancellation, rescission or expiration of the Franchisee's rights) under the Lease, any sublease, the Franchise Agreement or other agreement between the Franchisee and the Franchisor (or any affiliate). In such event, the Franchisor (or its designee) shall have the right, and is hereby empowered, (but has no obligation) to take possession of the Premises, expel Franchisee therefrom, and, in such event, Franchisee shall have no further right, title or interest in or under the Lease or to the Premises, all such rights thereby passing to the Franchisor or its designee, in each case without the Landlord's further consent. The Franchisee agrees to do all acts necessary or appropriate to accomplish such assignment on the Franchisor's request. The Franchisee will reimburse the Franchisor for the costs and expenses incurred in connection with any such retaking, including, without limitation, the payment of any back rent and other payments due under the Lease (whether such payments are made by a separate agreement with the Landlord or otherwise), attorney's fees and expenses of litigation incurred in enforcing this Assignment, costs incurred in reletting the Premises and costs incurred for putting the Premises in good working order and repair.

Franchisee agrees that it will not suffer or permit any surrender, termination, amendment or modification of the Lease without the prior written consent of Franchisor, and will notify Franchisor of any impending negotiations in accordance with section 3.2 of the Franchise Agreement. Throughout the term of the Franchise Agreement, Franchisee agrees that it shall elect and exercise on a timely basis all options to extend the term, or renew or assume in bankruptcy, the Lease, unless Franchisor otherwise agrees in

writing. Upon failure of Franchisee to so elect to extend or renew or assume the Lease, Franchisee hereby appoints Franchisor as its true and lawful attorney-in-fact to exercise such options in the name, place and stead of Franchisee for the sole purpose of effecting any extension, renewal or assumption, in each case for the account of the Franchisee and without any liability or obligation of the Franchisor

Failure of the Franchisor to exercise any remedy hereunder shall not be construed or deemed to be a waiver of any of its rights hereunder. The rights and remedies of the Franchisor under this Assignment are cumulative and are not in lieu of, but are in addition to, any other rights and remedies which the Franchisor shall have under or by virtue of the Franchise Agreement or otherwise. The terms, covenants, and conditions contained herein shall bind the Franchisee and its successors and assigns, and inure to the benefit of the Franchisor and its successors and assigns. In the event of any dispute between the parties regarding this Assignment, or any matter related in any way to it, the dispute resolution provisions (including, but not limited to, mediation, binding arbitration, waiver of jury trial and limitation of damages) of the Franchise Agreement between the Franchisor and the Franchisee shall apply. The arbitrator in any such proceeding shall have the full power and authority to grant an appropriate award to give full effect to this Assignment, expelling the Franchisee from the Premises and awarding possession to the Franchisor, as well as granting such other relief as may be proper and fair at law and by equity. If there is more than one Franchisee, their obligations hereunder will be joint and several.

This Assignment, any memorandum hereof or any financial statement related hereto may be recorded by, and at the expense of, the Franchisor. The Franchisee hereby appoints the Franchisor as its attorney in fact to execute any and all documents and to take any and all such actions, as are necessary or appropriate to record such instrument referenced above.

Notwithstanding anything to the contrary contained herein, the Franchisee agrees to indemnify, defend and hold harmless the Franchisor with respect to all obligations and liabilities, including, without limitation, the obligations to pay all rent and other monies due under the Lease, that arise after the date of any assignment of the Lease that transpires under this Assignment; provided, however, nothing hereunder shall affect any obligations or covenants of the Franchisee owed under its Franchise Agreement with the Franchisor, including, without limitation, any post-termination covenant not to compete.

FRANCHISEE:		
Signature	Signature	
Printed Name	Printed Name	
Date:	Date:	

LANDLORD APPROVAL:

The undersigned Landlord hereby consents to and approves of the above-described Collateral Assignment of Lease by the Franchisee to the Franchisor and further agrees that immediately upon notice to the Landlord, the Franchisor shall have the right to succeed the Franchisee as the tenant under the Lease without further action or consent by any of the parties hereto; provided, however, nothing in the Assignment or in

the consent and approval by the Landlord to the Assignment shall affect any other rights of the Landlord under the Lease.

LANDLORD	FRANCHISOR		
	TECHY, LLC		
Ву:	By:		
Its:	Its:		
Date:	Date:		

Techy, LLC

EXHIBIT 3.3 SITE SELECTION ACKNOWLEDGMENT AND ADA CERTIFICATION FORM

Techy, LLC, a Florida Limited Liability Company, ("Franchisor," or "we," "us" or "o	ır") anc
("Franchisee(s)" or "you") are parti	es to a
ranchise agreement dated, 20 (the "Franchise Agreement") for the operation of a Tec	hy Store
t the location identified below (the "Store"). This Form refers to the following proposed locat	ion (the
site"), which we have accepted and consented to under the terms of a Franchise Agreement:	
STORE ADDRESS)	

You understand and agree that:

- You have independently selected and are solely responsible for the selection, development and operation of the site. Alternatively, if we propose a site to you, you have independently investigated those locations and you are solely responsible for your selection, development and operation of the site.
- Our consent to (or proposal of) the site is not, and should not be relied upon as, a recommendation or endorsement of such location, nor a representation or warranty as to the suitability of the site for any purpose, the likelihood of any success at such site, or otherwise. We cannot and do not guarantee the success of any location, whether selected by you or suggested by us.
- Although we, or companies referred by or associated with us, may assist you in site location, identification, financing and/or development by providing consultation, lease or Operating Agreement negotiation, evaluation and/or other assistance, including references to potential locations, contractors and other professionals (individually and collectively referenced as "site-related matters") neither we nor any Franchisor-Related Persons/Entities will have any liability with respect to any location to be selected, obtained and/or used by you or for any site-related matters. The sole responsibility for the selection and/or approval of the site and all site related matters is your own.
- We and/or any Franchisor-Related Persons/Entities may have made, or may make, available to you standard and/or site specific plans and specifications to be used by you in the construction of your Store. You are solely responsible for obtaining architectural, engineering and other applicable professional services to prepare surveys, site and foundation plans and adapt any plans and specifications to the site and all applicable laws, regulations and ordinances.
- In signing this form, you acknowledge that you have not relied on any assistance or input from us or any Franchisor-Related Persons/Entities with respect to the site and any site related matter, and you waive any claims you may have against us and/or any Franchisor-Related Persons Entities in connection with the site and any such matters.
- You acknowledge that you (and each of your owners, if you are a Business Entity) have had the opportunity and been advised by us to have this Agreement and all other documents reviewed by your own attorney, and that you have read, understood, had an opportunity to discuss and agreed to each provision of this Agreement. You agree that you have been under no compulsion to sign this Agreement.
- You agree that this Form, along with the Franchise Agreement between us, contains the final, complete and exclusive expression of the terms of your and our agreement with respect to the subject matter of this document and supersedes all other agreements and/or representations of any kind or nature. Any

understandings, agreements, representations, or otherwise (whether oral or written) which are not fully expressed in this document and the Franchise Agreement are expressly disclaimed by you.

ADA Certification:

In accordance with Section 3.3 of the Franchise Agreement, Franchisee certifies to Franchisor that to the best of Franchisee's knowledge, the Techy Store and its adjacent areas comply with all applicable federal, state and local accessibility laws, statutes, codes, rules, regulations and standards, including but not limited to the Americans with Disabilities Act. Franchisee acknowledges that it is an independent contractor and the requirement of this certification by Franchisor does not constitute ownership, control, leasing or operation of the Techy Store. Franchisee acknowledges that Franchisor has relied on the information contained in this certification. Furthermore, Franchisee agrees to indemnify Franchisor and Franchisor-Related Persons/Entities, and each of their respective officers, directors, and employees, in connection with any and all claims, losses, costs, expenses, liabilities, compliance costs, and damages incurred by such indemnified party(ies) as a result of any matters associated with Franchisee's compliance (or failure to comply) with the Americans with Disabilities Act, as well as the costs, including attorneys' fees, related to the same.

In the event of any dispute concerning or relating to this document and/or any of the transactions and/or matters to which it may apply, such dispute will be resolved in accordance with the dispute resolution provisions of the Franchise Agreement (including, but not limited to, Articles 19, 20 and 21 of the Franchise Agreement, providing for BINDING ARBITRATION and WAIVER OF JURY TRIAL among other terms). Terms not defined in this document shall have the same meaning as they do in the Franchise Agreement.

All signers are jointly and severally responsible for the representations and promises described in this Acknowledgment and Certification Form.

FRANCHISEE (Individual)	FRANCHISEE (Corp. or LLC)
(Signature)	(Legal Name of Franchisee Entity)
(Printed Name)	a (Jurisdiction of Formation) (Corp. or LC)
(Signature)	By: (Signature)
(Printed Name)	(Printed Name)
	Its:

EXHIBIT 8.1 EMPLOYEE CONFIDENTIALITY AGREEMENT

(<u>Note to Franchisee</u>: This is a form which has not been checked by us for compliance with local laws and should be reviewed by your attorney for your protection and to maximize enforceability. You are responsible for ensuring that the terms of the agreement used by you comply with all applicable laws, since they may vary from one state or province to another.)

In consideration of the employment of the below named Employee, and as inducement for disclosure by _______ [franchisee entity or individual name] (the "Franchisee"), doing business as an independent Franchisee, for the continuation of such employment, and for the compensation which I have received and may receive during the period of such employment, I, the undersigned Employee, hereby agree that during my employment with the Franchisee and for any post-term periods specified in this Agreement:

- (1) My employment by the Franchisee will be in accordance with the policies, rules and regulations of the Franchisee, as the same now exist, or as they may be established or modified from time to time.
- (2) The Franchisee has, subject to a Franchise Agreement with Techy, LLC (the "Franchisor") acquired specified rights to use certain "Confidential Information," which includes all information (current and future) relating to the operation of an franchised store or the franchise System, including, among other things, all: i) Manuals, training, techniques, processes, policies, procedures, systems, data and know how regarding the development, marketing, operation and franchising of Techy Repairs & Smart Home Installs Powered by DrPhoneFix franchised stores; ii) designs, specifications and information about Products and Services, iii) all information regarding customers and suppliers, customer, supplier and product lists, technical processes and know how, specifications, manuals, notes, reports, memoranda, data, equipment and/or secured areas (in written, audio, magnetic and/or electronic format), including any statistical and/or financial information and all lists, together with various designs, techniques, know-how, marketing concepts and information, operating procedures and technical information and ancillary products, services and techniques and other related applications which are not generally known in the industry or to the public, and, in addition, iv) any other items that an arbitrator or court deems reasonably appropriate for protection.
- (3) "Confidential Information" is not intended to include any information that: is or subsequently becomes publicly available (other than by breach of any legal obligation), or became known to you other than through a breach of a legal obligation.
- (4) By virtue of my employment by the Franchisee, the Employee will or may have access to Confidential Information.
 - (5) With reference to the Confidential Information, the Employee agrees as follows:
 - a. The Confidential Information is a valuable trade secret licensed to the Franchisee by the Franchisor and/or related companies, and I will not use the Confidential Information other than within the course and scope of my employment responsibilities and functions.
 - b. Not to release or divulge any Confidential Information unless first expressly authorized to do so in writing by a superior or an officer of the Franchisee; provided, however, that during the period of my employment, Employee will be permitted to release or divulge the same, or any portion thereof, to persons employed or otherwise closely associated with the Franchisee, but

- only to the extent that such persons have a need to know the same within the course and scope of their employment by, or close association with, the Franchisee (for example, attorneys and/or accountants retained by the Franchisee.)
- c. Any and all publications/copies/disclosures of the Confidential Information in any form, which may be presented to me or to which I may be granted access, are on loan and will at all times remain, the exclusive property of the Franchisee and/or the Franchisor; the Confidential Information is being given to me in trust and confidence; and I will accept the same subject to such trust.
- d. During the period of my employment by the Franchisee, I will take all necessary steps to safeguard and maintain the secrecy and confidentiality of the Confidential Information in my possession or control, including (by way of illustration and not limitation) (i) securing the Confidential Information in locked or otherwise secured files; and (ii) refraining from making copies or reproductions of the Confidential Information, or any portions thereof, unless necessary for the carrying out of my employment responsibilities, or if first expressly authorized to do so by a superior or an officer of the Franchisee.
- e. Upon the termination of my employment by the Franchisee, I will immediately return to the Franchisee any and all Confidential Information and all copies thereof, which may have been entrusted to me or which I may have generated or copied, as well as any physical property of the Franchisee, including books, tapes, dvd's, equipment, and the like, whether proprietary or not, which I may have in my possession or control.
- f. My obligations with respect to the Confidential Information will continue beyond the period of my employment.
- All inventions, discoveries, developments, improvements, innovations, and writings, whether or not eligible for patent and/or copyright protection (hereinafter collectively referred to as "Innovations" or "Inventions" as may be appropriate), conceived or made by me either solely or in concert with others, during the period of my employment by the Franchisee (including, but not limited to, any period prior to the date of this Agreement) whether or not made or conceived during working hours, which (a) relate in any manner to the existing or contemplated business, or the development of activities, of the Franchisee and/or the Franchisor, or (b) are suggested by, or result from, my work for the Franchisee, or (c) result from my use of the Franchisee's time, materials, or facilities, will be the sole and exclusive property of the Franchisor. Any Inventions made by me, or disclosed by me to a third party, or described in a patent application of mine, within nine (9) months following the period of my employment by the Franchisee, will be presumed to have been conceived or made by me during the period of my employment with the Franchisee, unless I can prove they were entirely conceived and made by me following the period of such employment.
- (7) I will promptly make a full disclosure to the Franchisor, and hold in trust for the sole right and benefit of the Franchisor, any and all Inventions which I may solely or jointly conceive, write, develop, reduce to practice; or cause to be conceived, written, developed, or reduced to practice, during the period of time I am employed by the Franchisee, and thereafter in accordance with the provisions of this Agreement.
- (8) I hereby assign and agree to assign to the Franchisor, all of my right, title and interest in and to all my Inventions, if any, and agree, during and subsequent to my employment, to execute and deliver to the Franchisor, ownership, title and exclusive rights therein, all without charge.

- (9) I hereby assign and agree to assign to the Franchisor all of my right, title and interest in and to any and all United States and foreign patents and copyrights covering my Inventions, and all reissues, registrations and renewals thereof. I further agree, during and subsequent to my employment, to aid (i) in the prosecution of any United States or foreign applications for Letters Patent or the registration of copyrights covering such inventions and (ii) in the enforcement of any such patents or copyrights. In this connection, I will, at the Franchisor's request and expense, execute, acknowledge and deliver any and all documents and oaths, and take such further action considered necessary by the Franchisor for the foregoing purposes, without charge.
- (10) In the event the Franchisor is unable, for any reason whatsoever, to secure my signature to any lawful and necessary documents required to assign, apply for, or prosecute any United States or foreign applications for Letters Patent or the registration of copyrights in and to my Inventions or otherwise which belong to the Franchisor by virtue of the provisions of this Agreement or otherwise, I hereby irrevocably designate and appoint the Franchisor and its duly authorized officers and agents as my agent and attorney-in-fact, to act for and in my behalf and stead to execute and file any such assignments and applications, and to do all other lawfully permitted acts to further the prosecution and issuance of Letters Patent thereon and/or registrations of copyrights with the same legal force and effect as if executed by me.
- (11) My compensation as an employee of the Franchisee will cover any Inventions which I may conceive or make hereunder, and I will not be entitled to any additional compensation therefore.
- (12) I represent to the Franchisee and the Franchisor that I have no right, title or interest in or to any invention which has been made, conceived or reduced to practice by me (solely or jointly with others) prior to my employment by the Franchisee.
- (13) My services, and the Confidential Information which may be entrusted to me, are unique, and, if I breach this Agreement, the Franchisee and the Franchisor may not be adequately compensated by damages. Therefore, if I violate the terms of this Agreement, either during or after my employment, the Franchisee and the Franchisor will be entitled, in addition to all other remedies available to either, to equitable relief by injunction or otherwise, thereby enjoining or restraining me, and those persons acting in concert with me, from the continuation of any breaches hereof. The right to equitable relief granted in the foregoing sentence will not preclude the Franchisee or the Franchisor from seeking actual money damages from me or any other party in the event of a breach or threatened breach of this Agreement.
- Ouring my employment by the Franchisee, and for one (1) year after termination of such employment, I will not conduct, operate, consult, advise or in any manner be associated, directly or indirectly, with any business or operation substantially similar to or competitive with that conducted by the Franchisee within the Franchisee's Territory (as defined by Franchisee's franchise agreement). Such restriction includes the furnishing and/or use of the Confidential Information to any person and/or entity, whether gratuitously, on a consulting basis, as an owner, shareholder, partner, employee or associate. For informational purposes, the Territory as it currently exists is shown on an attachment to this Agreement.
- (15) Nothing contained in this Agreement will be construed to prevent me from engaging in a lawful profession, trade or business after my employment with the Franchisee. I confirm that I possess valuable skills unrelated to the Franchised Business and have the ability to be self-supporting and employed regardless of the restrictions described in this Agreement. I also acknowledge that the restrictions of this Agreement will not prevent me from practicing a lawful profession, trade, or business and are limited to the express restrictions detailed herein. This Agreement will be construed only as one which prohibits me from engaging in practices unfair to the Franchisee, and which are in violation of the confidence and trust reposed in me by the Franchisee with respect to its Confidential Information.

- (16) The parties agree to the following dispute resolution provisions:
- a. Any litigation, claim, dispute, suit, action, controversy, or proceeding of any type whatsoever including any claim for equitable relief and/or where you are acting as a "private attorney general," suing pursuant to a statutory claim or otherwise, between or involving you and us on whatever theory and/or facts based, and whether or not arising out of this Agreement, and including any dispute involving the Franchisor, ("Claim") will be processed in the following manner, you and we each expressly waiving all rights to any court proceeding, except as expressly provided below at sub-sections (g) and (h) below:
- (1) First, discussed in a face-to-face meeting held at a neutral location reasonably near your franchised location within thirty (30) days after either you or we give written notice to the other proposing such a meeting;
- (2) Second, if unresolved, submitted to non-binding mediation for a minimum of four (4) hours before (a) the American Arbitration Association ("AAA") or its successor or (b) Judicial Arbitration and Mediation Service ("JAMS") or its successor, if the AAA cannot conduct such mediation. The mediator shall be a neutral person experienced in franchising. We will pay the costs of the first four (4) hours of any mediation, and no mediation is required to extend beyond such four (4) hour period. Any party may be represented by counsel and may, with permission of the mediator, bring persons appropriate to the proceeding.
- Third, if unresolved, submitted to Arbitration before and in accordance with the arbitration rules of (a) the American Arbitration Association ("AAA") or its successor or (b) Judicial Arbitration and Mediation Service ("JAMS") or its successor, if the AAA cannot conduct such arbitration. Arbitration may be filed prior to a face-to-face meeting and/or mediation, with such face-to-face meeting and/or mediation to follow as quickly thereafter as possible. Arbitration and mediation/meeting may proceed concurrently. The arbitrator must be a lawyer substantially experienced in franchising. Judgment on any arbitration award may be entered in any court having jurisdiction thereof, or of Franchisor or of Franchisee or of Employee, subject to the opportunity for appeal to a panel, as contemplated below. The arbitrator's award will be in writing.
- (4) Fourth, a final award by an arbitrator (there will be no appeal of interim awards or other interim relief) may be appealed within thirty (30) days of such final award. Appeals will be conducted before a three (3)-arbitrator panel appointed by the same organization as conducted the arbitration, each member of which must be a lawyer substantially experienced in franchising. The arbitration panel will not conduct any trial de novo or other fact-finding function. Such panel's decision will be in writing, may be entered in any court having jurisdiction and will be BINDING, FINAL AND NON-APPEALABLE. On request by either party, the arbitration panel will provide to all disputants a reasoned opinion with findings of fact and conclusions of law.
- (5) If required by applicable law for any arbitration provision (including the appeal process) to be enforceable (for example, to preserve constitutionally or statutorily provided rights), the arbitrator or a court will, as soon as possible, appropriately allocate between you and us the fees of the arbitrator(s) and/or his/her related organization, or require us to advance a portion of such fees subject to possible reimbursement, or otherwise address such issues so as to allow the mediation/arbitration to proceed and may adjust such allocations appropriately during the arbitration process for such purpose. In addition, you or we can choose to deposit all fees required by any arbitration organization, subject to later allocation between you and us, as provided above.

- (6) If the organizations specified by this Agreement to conduct any mediation and/or arbitration are unable or unwilling to conduct such proceeding(s), and the parties to the dispute cannot agree on an appropriate substitute organization or person to conduct such proceeding(s), then a court of competent jurisdiction shall designate an appropriate organization or person to conduct such proceeding(s).
- b. <u>Location</u>. Any mediation/arbitration (and any arbitration appeal) will be conducted exclusively at a neutral location in the county and state in which our then-current headquarters is located, which may change from time to time, and be attended by you and us, and/or designees authorized to make binding commitments on each of our respective behalf. You and we agree that the provisions of this Article will control, notwithstanding any language included in our franchise disclosure documents due to state requirements suggesting that the provisions of any section might be unenforceable due to a failure to have a meeting of the minds or otherwise. Neither you nor we have any expectation that the provisions of this (or any other) section will be unenforceable or that they will not be enforced. You understand and agree that one effect of this paragraph may be that mediation, arbitration and other related costs may be greater, and it may be more difficult for you to proceed, than if those proceedings took place in a location near your residence or business. If any court determines that this provision is unenforceable for any reason, mediation/arbitration (and any arbitration appeal) will be conducted at a neutral location reasonably near your franchised location.
- c. Arbitration Authority. Arbitrators shall apply all applicable law, and a failure to apply the applicable law in accordance with Section 19.8 C. of the Franchise Agreement shall be deemed an act in excess of authority. The arbitrator shall be appointed within thirty (30) days of the filing of any demand for arbitration. Any arbitration hearing shall be held within ninety (90) days of the applicable appointment. The arbitrator shall decide all questions relating in any way to the parties' agreement to arbitrate, including but not limited to arbitrability, applicability, subject matter, timeliness, scope, remedies, and any alleged fraud in the inducement or claimed unconscionability. The arbitrator can issue summary orders disposing of all or part of a claim and provide for temporary restraining orders, preliminary injunctions, injunctions, attachments, claim and delivery proceedings, temporary protective orders, receiverships and other equitable and/or interim/final relief. Each party consents to the enforcement of such orders, injunctions, etc. by any court having jurisdiction. The subpoena powers of the arbitrator with respect to witnesses to appear at the arbitration proceeding shall not be subject to any geographical limitation. No award in arbitration will have any effect of preclusion or collateral estoppel in any other adjudication or arbitration.
- d. <u>Intention to Arbitrate</u>. The parties each knowingly waive all rights to a trial and select arbitration as the sole means to resolve disputes, except as expressly provided otherwise in this Agreement, understanding that arbitration may be less formal than a court or jury trial, may use different rules of procedure and evidence and that appeal is generally less available, and that the fees and costs associated with a face-to-face meeting, mediation and/or arbitration may be greater than in civil litigation, but still strongly preferring a face-to-face meeting, mediation and/or arbitration as provided in this Agreement believing that arbitration has overriding advantages from a business standpoint. The Franchisee intends to, and the Employee expressly agrees that the Franchisee may, fully enforce each of the provisions of this Agreement, including those relating to arbitration, venue, choice of laws, or otherwise, having had an express meeting of the minds regarding each of such matters.
- e. <u>Choice of Laws.</u> Notwithstanding any provision of this Agreement or otherwise relating to which state or other laws this Agreement will be governed by, any provisions of state, or other law to the contrary, the Franchisee and the Employee mutually intend and agree that (1) the arbitrator shall decide any and all questions relating in any way to the parties' agreement (or claimed agreement) to arbitrate, including but not limited to applicability, subject matter, timeliness, scope, remedies, claimed unconscionability and any alleged fraud in the inducement and/or the enforcement of the agreement to

arbitrate contained herein and that this Agreement and all related matters shall be governed exclusively by the Federal Arbitration Act (9 U.S.C. § 1 et seq.) and the federal common law of arbitration and (2) the Franchisee and the Employee mutually intend and agree (and have expressly had a meeting of the minds) to fully enforce all of the provisions of this Agreement and all other documents signed by the Franchisee and the Employee, including (but not limited to) all venue, choice-of-laws, mediation/arbitration provisions and other dispute avoidance and resolution provisions and to rely on federal preemption under the Federal Arbitration Act (9 U.S.C. § 1 et seq.).

- f. No Bond. No party will be required to post a bond in order to obtain any injunctive or other equitable relief.
- g. <u>Intellectual Property Claims</u>. These dispute resolution provisions apply to any claims/arbitration between the parties (including any dispute involving the Franchisor and/or any person/entity related in any way to it), and/or by any owner and/or affiliate thereof, or which could be brought in their behalf or by any successor; provided that any claims or disputes relating primarily to the validity of the Franchisor's Marks and/or any Intellectual Property licensed to the Franchisee will be subjected to court proceedings; provided that only the portion of any claim or dispute relating primarily to the validity of the Marks and/or any Intellectual Property licensed to the Franchisee and requesting equitable relief shall be subject to court action, and any portion of such claim seeking monetary damages will be subject to the process outlined above.
- h. <u>Litigation</u>. Subject to the foregoing obligations regarding mediation/arbitration, any litigation (for example, to enforce such obligations) between the undersigned (or involving the Franchisor) will be held in the United States District Court encompassing the Franchisee's then-current headquarters (the "Proper Federal Court"). Proceedings will be held only in the Proper Federal Court, subject to the following exceptions:
- (i) if a basis for federal jurisdiction does not exist, then any such proceeding shall be brought exclusively before a court in the most immediate state judicial district encompassing the Franchisee's then-current headquarters and having subject matter jurisdiction (the "Proper State Court");
- (ii) proceedings to remove or transfer a matter to the Proper Federal Court and/or to compel arbitration as contemplated by this Agreement may be brought in the court where the applicable action is pending and/or the Proper State or Federal Court; and
- (iii) any action primarily with respect to any real and/or personal property (including any action in unlawful detainer, ejectment or otherwise) may be brought in any court of competent jurisdiction and/or the Proper State or Federal Court.
- (17) Upon the termination of the Employee's employment, the Franchisee may notify anyone thereafter employing me of the existence and provisions of this Agreement.
- (18) The employee understands that his/her employment is at will, and that just as the Employee may terminate his/her employment at any time and for any reason (or for no reason), the Franchisee may do the same, unless a fixed term is specified herein or in another writing, executed by the Employee and the Franchisee.
- (19) The Employee represents that he/she has no existing agreements with, obligations to, or interest in any other party that keep the Employee from complying with his/her obligations under this Agreement, or which may give rise to a conflict of interest, except those identified on the attached list signed by the Employee and the Franchisee. If no list is attached, the Employee agrees that there are no

such agreements, obligations or interests on my part. In addition, the Employee agrees to promptly disclose in writing to his/her superior any future agreements, obligations and/or interests which may preclude or conflict with his/her obligations hereunder.

- (20) The Employee will not use on behalf of, or divulge to, the Franchisee, or its agents or employees, during his/her employment by the Franchisee, confidential or trade secret information acquired during any prior employment of his/hers or from any other source outside of the Franchisee, provided, of course, that the Employee knows or should know of its nature as confidential or a trade secret.
- (21) The Employee understands and confirms that he/she has no authority whatsoever to make any commitment or enter into any arrangement or contract on behalf of the Franchisee unless authorized by an officer of the Franchisee in writing.
- (22) The Franchisee and the Employee agree that this Agreement supersedes any prior oral agreement and/or written agreement by and between them relating generally to the subject matter of this Agreement; the Employee represents and warrants that there are no such prior oral agreement and/or written agreement.
- (23) The Franchisee and the Employee agree that, if it is determined that any provision of this Agreement is illegal or unenforceable, such provision will be enforced to the fullest extent permissible under governing law and such determination will solely affect such provision and not impair the remaining provisions of this Agreement. The time period of the restrictions described in this Agreement will be extended by the length of time during which the Employee is in breach of any such provision of this Agreement.
- (24) The Franchisee and the Employee agree that this Agreement will be construed, and the validity, performance and enforcement hereof will be governed by the laws of the State in which the Franchisee's headquarters is located.
- (25) A waiver by the Franchisee or Franchisor of any breach of this Agreement on the Employee's part will not operate as or be construed as a waiver of any subsequent breach hereof.
- (26) This Agreement will inure to the benefit of and be enforceable by the Franchisee and the Franchisor, and any successors and assigns of the foregoing, and that it will be binding upon the Employee, his/her executors, administrators, legatees, distributees, heirs and other successors in interest. The Franchisor is an intended third-party beneficiary of this Agreement and may protect its interests by enforcing the parties' obligations, but the Franchisor is not a party to this Agreement, is not the employer of, and has no obligations to, the Employee.
- (27) The Employee has read the foregoing provisions, understands that this Agreement defines the terms and conditions under which the Franchisee is willing to employ or continue to employ the Employee, is executing this Agreement and agreeing to abide by its provisions voluntarily, and the Franchisee has given the Employee a copy of this Agreement for his/her future reference so as to avoid any possible oversights or misunderstandings regarding its provisions.

(Date and Signatures on Following Page)

Dated:	, 20 at	City	,	State
FRANCHISEE:			EMPLOYEE:	
			Employee's name	
By:			Employee's signature	
Its:				

EXHIBIT 10. EXECUTIVE ORDER 13224 AND RELATED CERTIFICATIONS

If the Franchisee is an individual or individuals, the Franchisee certifies that he/she/they are not, nor to my/our best knowledge have I/us been designated, a terrorist and/or a suspected terrorist, nor am I/us associated and/or affiliated in any way with any terrorist and/or suspected terrorist person and/or organization, as defined in U.S. Executive Order 13224 and/or otherwise.

If the Franchisee is a company, the person(s) signing on behalf of the Franchisee certifies(y) that, to the Franchisee's and such person's best knowledge, neither the Franchisee, such person, and/or any owners, officers, board members, similar individuals and/or affiliates/associates of the Franchisee have been designated, a terrorist and/or a suspected terrorist, nor is the Franchisee or any such persons and/or affiliates/associates owned, controlled, associated and/or affiliated in any way with any terrorist and/or a suspected terrorist person and/or organization, as defined in U.S. Executive Order 13224 and/or otherwise.

Franchisee agrees to fully comply and/or assist Franchisor in its compliance efforts, as applicable, with any and all laws, regulations, Executive Orders or otherwise relating to antiterrorist activities, including without limitation the U.S. Patriot Act, Executive Order 13224, and related U.S. Treasury and/or other regulations, including properly performing any currency reporting and other obligations, whether relating to the Franchise or otherwise, and/or required under applicable law. The indemnification responsibilities provided in the Franchise Agreement cover the Franchisee's obligations hereunder.

FRANCHISEE				
Ву:				
Printed Name:				
Title:				
Date:				

EXHIBIT 23. SPOUSAL CONSENT

The undersigned person(s) hereby repre	esent(s) to Techy, LLC that he/she is the spouse of
the individual franchisee(s) who has executed a	n Techy, LLC Franchise Agreement dated the
day of, 20_ or that he/she is the sp	ouse of one of the Owners of the legal entity who
executed the Techy, LLC Franchise Agreement	of the same date. In consideration of the grant by
Techy, LLC to the Franchisee as herein provi	ided, each of the undersigned spouses agrees, in
consideration of benefits received and to be rece	ived by each of them, jointly and severally, and for
themselves, their heirs, legal representatives and	l assigns that they, and each of them, will be firmly
bound by all of the terms, provisions and co	nditions of the foregoing Techy, LLC Franchise
Agreement, that they and each of them jointly a	and severally do hereby unconditionally guarantee
the full and timely performance by the Franchis	see of each and every obligation of the Franchisee
under the aforesaid Franchise Agreement, inclu	ading, without limitation, any indebtedness of the
Franchisee arising under or by virtue of the a	aforesaid Franchise Agreement. The undersigned
jointly and severally further agree(s) to be boun	d by the in-term and post-term covenants set forth
in Section 10 and in Paragraph 17.4 of the afore	said Franchise Agreement.
WITNESS:	Signature:
TITILIDO.	Digitation.
	Printed Name:

EXHIBIT E-1 IN-STORE (WALMART) ADDENDUM TO TECHY, LLC

FRANCHISE AGREEMENT

7	THIS	S AD	DENDUM	I is enter	red into	and made t	his the _		_da	y of	,
20,	by	and	between	Techy,	LLC	(hereinafter	referred	to	as	"Franchisor"),	and
			, (h	ereafter 1	eferred	to as "Franch	isee").				
				<u>v</u>	<u>V I T N</u>	<u>ESSETH</u>	<u>:</u>				

WHEREAS, Franchisee and Franchisor are contemporaneously entering into a Franchise Agreement; and

WHEREAS, Franchisee desires to enter into Franchisor's In-Store (Walmart) Addendum to operate a franchise solely within a Walmart location; and

WHEREAS, Franchisee understands Franchisee must enter into a separate sublease agreement for the In-Store (Walmart) space with Franchisor's affiliate, DrPhoneFix, Inc. and also enter into a Start-Up Package agreement with DrPhoneFix, Inc. to prepare the subleased space in a ready to use condition which shall include Franchisor's required furnishings, signage, equipment and start-up inventory; and

WHEREAS, Franchisee and Franchisor desire to clarify or modify certain provisions of the Franchise Agreement as set forth hereinafter.

- **NOW, THEREFORE,** in consideration of the mutual and several covenants and obligations contained herein, the parties hereto agree that this Addendum to the parties Franchise Agreement shall be attached to the executed Franchise Agreement and shall partially amend and revise the parties Franchise Agreement as follows:
 - 1. Notwithstanding anything to the contrary in **Paragraph 2.2** "Grant" of the Franchise Agreement, Franchisee has requested Franchisor to grant a Franchise for Franchisor's In-Store (Walmart) franchise concept and Franchisor hereby grants to Franchisee the right and obligation (The "In-Store Walmart" franchise), and Franchisee accepts and undertakes the obligation to operate one (and only one) In-Store (Walmart) franchise location during the term defined in the Franchise Agreement and use the System and Marks within a Walmart retail store located at _______ (the "Site").
 - 2. Notwithstanding anything to the contrary in **Section 3** and specifically **Paragraphs 3.1, 3.2 and 3.4**, Franchisee understands and agrees that Franchisee is required to enter into a sublease agreement for the In-Store (Walmart) location with Franchisor's affiliate, DrPhoneFix, Inc. Franchisee acknowledges and agrees that the location within the Walmart store is located at the address in 1 above, that the

designation of a location is not a representation nor a warranty of success or suitability of the Site, and Franchisee agrees that neither Franchisor nor DrPhoneFix, Inc. shall have any liability whatsoever with respect to any role either Franchisor or DrPhoneFix, Inc. had in the selection of the Site. Franchisor shall not be responsible for failure of the In-Store (Walmart) location at the Site to meet Franchisee's expectations as to revenue or operational criteria. Further, Franchisor shall have a right to terminate the Franchise Agreement if the underlying Lease Agreement for the In-Store (Walmart) location at the Site is terminated or expires and neither Franchisee nor Franchisor nor its affiliate, DrPhoneFix, Inc., shall have any liability whatsoever to Franchisee except that Franchisee may obtain any unearned payments made in advance to DrPhoneFix, Inc., which shall be refunded only if such termination does not result in a breach of the sublease or breach of the Franchise Agreement. Neither Franchisor nor DrPhoneFix, Inc. shall have any obligation to provide Franchisee an alternative location in the event Franchisee's sublease expires or is terminated. Franchisee accepts this Franchise Agreement and Addendum with full and complete understanding that the grant of this Franchise contains no representation, warranty, promise or assurance of renewal or extension of the sublease agreement nor of the Franchisee's ability or right to obtain additional franchised locations and Franchisor has the absolute and sole discretion whether to offer or grant additional In-Store (Walmart) franchised locations to Franchisee. Franchisee further acknowledges and agrees that should the sublease expire or be terminated, Franchisee shall be solely responsible for finding another Techy Store Site location pursuant to Section 3 of the Franchise Agreement and within the Protected Territory set forth in Exhibit 2.2 of the Franchise Agreement.

- 3. Notwithstanding anything to the contrary in the Franchise Agreement, Franchisee understands and agrees to the following terms:
 - A. Franchisee understands and acknowledges that other franchisees of Franchisor may be granted franchise agreements at different times and different situations and Franchisee acknowledges that the provisions of such agreements may vary substantially from those contained in this Addendum and the Franchise Agreement to which it is attached and made a part of. Franchisee's obligations may also differ substantially from those of other franchisees.
 - B. Franchisor reserves the right in its sole discretion to vary its specifications, standards and operating practices and requirements among franchisees, including, without limitation, those relating to equipment, signage, operations, inventory and services. Franchisor may impose such variations to address differing or unique circumstances or for reasons Franchisor, in its discretion, deems good and sufficient. Franchisee understands and acknowledges that Franchisor and its affiliates have the right and discretion to treat franchisees differently even if they are similarly situated.

- C. Franchisee understands, agrees and acknowledges that Walmart may offer products and services for sale at the Walmart store Site, or elsewhere, that are the same or similar to the products and services offered for sale by Franchisee at the Walmart In-Store (Walmart) location.
- D. Upon execution of the Sublease Agreement, Franchisee shall also pay to DrPhoneFix, Inc. a Start-Up Package fee of \$______ for preparing the space in a ready to use condition which shall include Franchisor's designated furnishings, signage equipment and start-up inventory. The Start-Up Package fee is non-refundable.

MAKES NO WARRANTIES, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO, IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE FOR AN ITEM, WORK OR PRODUCTS PROVIDED BY DRPHONEFIX, INC. FRANCHISOR SHALL NOT BE RESPONSIBLE NOR LIABLE FOR ANY DIRECT, SPECIAL, INCIDENTAL, CONSEQUENTIAL OR EXEMPLARY DAMAGES.

THIS ADDENDUM shall be considered an integral part of the Franchise Agreement between the parties. Except as modified or supplemented by this Addendum, all of the rights and obligations of the parties set out in the Franchise Agreement are binding on the parties hereto.

To the extent this Addendum shall be deemed to be inconsistent with any terms or conditions of the Franchise Agreement or Exhibits or Attachments thereto, the terms of this Addendum shall govern.

[Signatures on page that follows.]

IN WITNESS WHEREOF, Franchisor has caused these presents to be executed in its name and on its behalf by its proper corporate officer, and Franchisee has hereunder affixed his/her/its hand and seal all on the day and year first above written.

WITNESS	FRANCHISEE (Individual)
(Signature)	(Signature)
	(Printed Name)
WITNESS	
(Signature)	(Signature)
	(Printed Name)
	FRANCHISEE (Corp., LLC or Partnership)
	(Legal Name of Franchisee Entity)
ATTEST	a(Jurisdiction of Formation) (Corp., LLC or Partnership)
	By:
	(Signature)
	Its:(Title)
	FRANCHISOR TECHY, LLC
ATTEST	
	By:
	Its:

EXHIBIT E-2 START-UP PACKAGE ADDENDUM TO TECHY, LLC

FRANCHISE AGREEMENT

20, by and	DENDUM is entered into and made this theday of, between Techy, LLC (hereinafter referred to as "Franchisor"), and, (hereafter referred to as "Franchisee").
	<u>W I T N E S S E T H</u> :
WHEREA Agreement; and	S, Franchisee and Franchisor are contemporaneously entering into a Franchise
	S , Franchisee and Franchisor desire to clarify or modify certain provisions of ement as set forth hereinafter.
obligations contained Agreement shall be	IEREFORE, in consideration of the mutual and several covenants and ed herein, the parties hereto agree that this Addendum to the parties Franchise attached to the executed Franchise Agreement and shall partially amend and ranchise Agreement as follows:
	nstanding anything to the contrary in the Franchise Agreement, Franchisee ands and agrees to the following terms:
A.	Franchisee understands and acknowledges that other franchisees of Franchisor may be granted franchise agreements at different times and different situations and Franchisee acknowledges that the provisions of such agreements may vary substantially from those contained in this Addendum and the Franchise Agreement to which it is attached and made a part of. Franchisee's obligations may also differ substantially from those of other franchisees.
В.	Franchisor reserves the right in its sole discretion to vary its specifications, standards and operating practices and requirements among franchisees, including, without limitation, those relating to equipment, signage, operations, inventory and services. Franchisor may impose such variations to address differing or unique circumstances or for reasons Franchisor, in its discretion, deems good and sufficient. Franchisee understands and acknowledges that Franchisor and its affiliates have the right and discretion to treat franchisees differently even if they are similarly situated.
C.	Upon execution of the Franchise Agreement, Franchisee shall pay to

DrPhoneFix, Inc. a Start-Up Package fee of \$______ for preparing

the store space approved by Franchisor in a ready to use condition which shall include Franchisor's designated furnishings, signage equipment and start-up inventory. The Start-Up Package fee is non-refundable.

FRANCHISOR MAKES NO WARRANTIES, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO, IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE FOR AN ITEM, WORK OR PRODUCTS PROVIDED BY DRPHONEFIX, INC. FRANCHISOR SHALL NOT BE RESPONSIBLE NOR LIABLE FOR ANY DIRECT, SPECIAL, INCIDENTAL, CONSEQUENTIAL OR EXEMPLARY DAMAGES.

THIS ADDENDUM shall be considered an integral part of the Franchise Agreement between the parties. Except as modified or supplemented by this Addendum, all of the rights and obligations of the parties set out in the Franchise Agreement are binding on the parties hereto.

To the extent this Addendum shall be deemed to be inconsistent with any terms or conditions of the Franchise Agreement or Exhibits or Attachments thereto, the terms of this Addendum shall govern.

[Signatures on page that follows.]

IN WITNESS WHEREOF, Franchisor has caused these presents to be executed in its name and on its behalf by its proper corporate officer, and Franchisee has hereunder affixed his/her/its hand and seal all on the day and year first above written.

WITNESS	FRANCHISEE (Individual)
(Signature)	(Signature)
	(Printed Name)
WITNESS	
(Signature)	(Signature)
	(Printed Name)
	FRANCHISEE (Corp., LLC or Partnership)
	(Legal Name of Franchisee Entity)
ATTEST	a(Jurisdiction of Formation) (Corp., LLC or Partnership)
	By: (Name)
	(Signature)
	Its:(Title)
	FRANCHISOR
ATTEST	TECHY, LLC
	By:
	Its:

EXHIBIT E-3 TECHY CAFÉ ADDENDUM TO TECHY, LLC

FRANCHISE AGREEMENT

	HIS ADDENDUM to the Franchise Agreement is made and entered into thisday
	, 20, by and between Techy, LLC (hereinafter referred to as "Franchisor").
and	, (hereafter referred to as "Franchisee").
	<u>W I T N E S S E T H :</u>
W Agreement	HEREAS , Franchisee and Franchisor are contemporaneously entering into a Franchise; and
operate a	HEREAS, Franchisee desires to enter into Franchisor's "Techy Café" Addendum to Techy Café solely in conjunction with Franchisee's franchise, pursuant to the parties Agreement; and
	HEREAS, Franchisor and Franchisee desire to clarify or supplement certain provisions chise Agreement as set forth hereinafter.
obligations Agreement	DW, THEREFORE, in consideration of the mutual and several covenants and contained herein, the parties hereto agree that this Addendum to the parties Franchise shall be attached to the executed Franchise Agreement and shall partially amend, revised the parties Franchise Agreement as follows:
1.	Franchisor grants Franchisee the right to establish and operate a Techy Café using Franchisor's System, Marks, and other Intellectual Property at the Site and at no other location (temporary or permanent).
2.	The term of this Addendum shall begin immediately and shall expire or terminate at the same time as the Franchise Agreement. Franchisee shall start operating the Techy Café no later than days after the execution of this Addendum.
3.	The definition of "Gross Revenue" in the Franchise Agreement shall be enlarged to include all revenue received from the Techy Café. All calculations of "Continuing Weekly Royalty Fee" under paragraph 9.2 of the Franchise Agreement ("Continuing Weekly Royalty Fee and Payment Dates") shall include all Gross Revenue from the Techy Café, and all reports of Gross Revenue, shall also include all Gross Revenue from or relating to the Techy Café.

4. Notwithstanding any contrary provision in the Franchise Agreement, Franchisee shall have a right to operate a Techy Café in addition to the Techy Store at the Site. The

- Techy Café shall be used for no purpose other than a Techy Café pursuant to this Addendum and the Franchise Agreement.
- 5. Franchisee shall be solely responsible to seek and obtain all licenses, consents and other approvals of any governmental authority required for Franchisee to lawfully open and operate the Techy Café.
- 6. Franchisee shall at Franchisee's expense effect leasehold improvements and obtain and install all fixtures, furniture and equipment at the Techy Café as required to comply with Franchisor's requirements and specifications as set forth in the Manual(s).
- 7. Franchisee shall not start construction of the Techy Café until final plans and specifications have been approved in writing by Franchisor. Franchisee shall not modify any plans and specifications approved by Franchisor without first obtaining Franchisor's written consent to the modification.
- 8. Franchisee shall be solely responsible to diligently design, construct, equip, prepare and open the Techy Café.
- 9. Franchisor shall have the right, but no obligation, from time to time to inspect the construction of the Techy Café while work is in progress. Franchisee shall make any modifications to the construction of the Techy Café as required by Franchisor.
- 10. Franchisee shall cause all signs used in connection with the Techy Café, whether exterior or interior, to conform to Franchisor's sign criteria, including but not limited to criteria for color, size, design and location. Franchisee shall obtain Franchisor's written consent for all signs and sign location before installation.
- 11. No vending machine, amusement device, video game, or other gaming or entertainment device of any nature shall be installed or used at the Tech Café without Franchisor's prior written consent.
- 12. Franchisee shall defend, indemnify and hold harmless Franchisor and all of its officers, agents, employees and indemnitees, as may be defined in the Franchise Agreement, from all losses, cost and expenses incurred in connection with any action, claim, suit or other proceeding, based in any way on any aspect of operation of the Techy Café.
- 13. All non proprietary goods, services and supplies required to operate the Techy Café shall be purchased by Franchisee only from suppliers and vendors designated or approved in writing by Franchisor.
- 14. In addition to The Training you will receive when you begin your Techy Store Training Program pursuant to the Franchise Agreement, You or if we approve a manager of your Techy Café, then your Techy Café manager, must successfully complete our Training Program for the Techy Café which is one day of training in addition to the Techy Store

Training Program. We may charge a reasonable fee for training of additional and/or subsequent managers. We can choose to eliminate or shorten training for persons previously trained or with comparable experience. All other provisions of Section 5 ("Training and Guidance") of the Franchise Agreement shall be in full force and effect.

- 15. Franchisee shall maintain for the Techy Café and personnel, all of the types and amounts of insurance that the Franchise Agreement requires.
- 16. Franchisor reserves the right in its sole discretion to vary its specifications, standards and operating practices and requirements among franchisees, including, without limitation, those relating to equipment, signage, operations inventory and services. Franchisor may impose such variations to address differing or unique circumstances or for reasons Franchisor, in its discretion, deems good and sufficient. Franchisee understands and acknowledges that Franchisor and its affiliates have the right and discretion to treat franchisees differently even if they are similarly situated.
- 17. In addition to the effects of termination or expiration under the Franchise Agreement, upon termination or expiration of the Franchise Agreement, Franchisee shall:
 - A. Pay all sums due and owing relating to the operation of the Techy Café;
 - B. At the request of Franchisor, assign all of Franchisee's interest in any lease for the Techy Café and if requested, provide Franchisor with all necessary assistance to enable Franchisor to take possession of the Techy Café;
 - C. Immediately deliver to Franchisor all computer software used by Franchisee and all data storage used in operating the Techy Café;
 - D. If Franchisor elects not to assume possession of the Techy Café after termination or expiration of the Franchise Agreement, Franchisee shall promptly perform sufficient redecoration and remodeling as needed, in Franchisor's sole judgement, to distinguish the Techy Café from its previous appearance as a Techy Café.

THIS ADDENDUM shall be considered an integral part of the Franchise Agreement between the parties. Except as modified or supplemented by this Addendum, all of the rights and obligations of the parties set out in the Franchise Agreement are binding on the parties hereto.

To the extent this Addendum shall be deemed to be inconsistent with any terms or conditions of the Franchise Agreement or Exhibits or Attachments thereto, the terms of this Addendum shall govern.

[Signatures on page that follows.]

IN WITNESS WHEREOF, Franchisor has caused these presents to be executed in its name and on its behalf by its proper corporate officer, and Franchisee has hereunder affixed his/her/its hand and seal all on the day and year first above written.

WITNESS	FRANCHISEE (Individual)
(Signature)	(Signature)
	(Printed Name)
WITNESS	
(Signature)	(Signature)
	(Printed Name)
	OR
	FRANCHISEE (Corp., LLC or Partnership)
	(Legal Name of Franchisee Entity)
ATTEST	a(Jurisdiction of Formation) (Corp., LLC or Partnership)
	By:(Signature)
	(Printed Name)
	Its:(Title)
ATTEST	FRANCHISOR TECHY, LLC
ATTEST	By:
	Its:

EXHIBIT E-4 WRAPPED TECHY VAN ADDENDUM TO TECHY, LLC

FRANCHISE AGREEMENT

THIS ADDENDUM to the Franchise Agreement is made and entered into this _____ day

	, 20, by and between Techy , LLC (hereinafter referred to as "Franchisor"),
and	, (hereafter referred to as "Franchisee").
	<u>W I T N E S S E T H</u> :
right to ope	HEREAS, Franchisee and Franchisor entered into a Franchise Agreement dated, 20 ("Franchise Agreement") pursuant to which Franchisee was granted the rate one Techy Store at the Site and use the Marks and System for the operation of the within the Protected Territory described in Exhibit 2.2 of the Franchise Agreement;
Addendum	HEREAS, Franchisee desires to enter into Franchisor's "Wrapped Techy Van" to operate a Mobile Van solely in conjunction with Franchisee's Techy Store franchise, the parties Franchise Agreement; and
	HEREAS, Franchisor and Franchisee desire to clarify or supplement certain provisions chise Agreement as set forth hereinafter.
obligations Agreement	DW, THEREFORE, in consideration of the mutual and several covenants and contained herein, the parties hereto agree that this Addendum to the parties Franchise shall be attached to the executed Franchise Agreement and shall partially amend, revise ent the parties Franchise Agreement as follows:
1.	Franchisor grants Franchisee the right to operate a Wrapped Techy Van using Franchisor's System, Marks, and other Intellectual Property within the Franchisee's Protected Territory and nowhere else. Franchisee must comply and meet Franchisor's specifications as to the Van model, Van age, vehicle type, interior configuration and signs and exterior vehicle wrap.
2.	The term of this Addendum shall begin immediately and shall expire or terminate at the same time as the Franchise Agreement. Franchisee shall start operating the Wrapped Techy Van no later than days after the execution of this Addendum.
3.	The definition of "Gross Revenue" in the Franchise Agreement shall be enlarged to include all revenue received from the Wrapped Techy Van. All calculations of "Continuing Weekly Royalty Fee" under paragraph 9.2 of the Franchise Agreement ("Continuing Weekly Royalty Fee and Payment Dates") shall include all Gross

- Revenue from the Wrapped Techy Van, and all reports of Gross Revenue, shall also include all Gross Revenue from or relating to the Wrapped Techy Van.
- 4. Notwithstanding any contrary provision in the Franchise Agreement, Franchisee shall operate a Wrapped Techy Van solely within the Protected Territory set forth in Exhibit 2.2 of the Franchise Agreement. The Wrapped Techy Van shall be used for no purpose other than a Wrapped Techy Van operated pursuant to this Addendum and the Franchise Agreement. The Wrapped Techy Van shall be used exclusively in the operation of the franchised business. Only one Wrapped Techy Van is authorized by this Addendum and the Wrapped Techy Van must meet all of Franchisor's approved standards and only a Franchisor approved Wrapped Techy Van may be used by Franchisee.
- 5. Franchisee shall be solely responsible to seek and obtain all licenses, consents and other approvals of any governmental authority required for Franchisee to lawfully operate the Wrapped Techy Van within the Protected Territory.
- 6. Franchisee shall at Franchisee's expense lease or purchase the Wrapped Techy Van which shall comply with all specifications and requirements set forth in the Manual(s) of Franchisor. All computer hardware and software, inventory, van wrap, furniture, fixtures and equipment for the Wrapped Techy Van must be installed and used as required to comply with Franchisor's specifications set forth in the Manual(s) or other written notifications from Franchisor to Franchisee. Franchisee shall not modify any specifications for the Wrapped Techy Van, without first obtaining Franchisor's written consent to the modification.
- 7. Franchisee shall defend, indemnify and hold harmless Franchisor and all of its officers, agents, employees and indemnitees, as may be defined in the Franchise Agreement, from all losses, cost and expenses incurred in connection with any action, claim, suit or other proceeding, based in any way on any aspect of operating the Wrapped Techy Van.
- 8. All non proprietary goods, inventory, services and supplies used for the operation of the Wrapped Techy Van shall be purchased by Franchisee only from suppliers and vendors designated or approved in writing by Franchisor.
- 9. Franchisee shall maintain for the Wrapped Techy Van and personnel, all of the types and amounts of insurance that the Franchise Agreement requires.
- 10. Franchisor reserves the right in its sole discretion to vary its specifications, standards and operating practices and requirements among franchisees, including, without limitation, those relating to equipment, signage, operations inventory and services. Franchisor may impose such variations to address differing or unique circumstances or for reasons Franchisor, in its sole discretion, deems good and sufficient. Franchisee

understands and acknowledges that Franchisor and its affiliates have the right and discretion to treat franchisees differently even if they are similarly situated.

- 11. In addition to the effects of termination or expiration under the Franchise Agreement, upon termination or expiration of the Franchise Agreement, Franchisee shall:
 - 1. Pay all sums due and owing relating to the operation of the Wrapped Techy Van:
 - At the request of Franchisor, assign all of Franchisee's interest in any lease for the Wrapped Techy Van and if requested, provide Franchisor with all necessary assistance to enable Franchisor to take possession of the Wrapped Techy Van;
 - 3. Immediately delivery to Franchisor all computer software used by Franchisee, all data storage used in operating the Wrapped Techy Van;
 - 4. If Franchisor elects not to assume possession of the Wrapped Techy Van, Franchisee shall within twenty-four (24) hours promptly perform sufficient redecoration, de-identification and remodeling as needed, in Franchisor's sole judgement, to distinguish the Wrapped Techy Van from its previous appearance as a Wrapped Techy Van.
- 12. Upon execution of this Addendum and the Franchise Agreement, Franchisee shall pay to DrPhoneFix, Inc. \$______ for inventory, van computer hardware and software, van wrap and interior van signage.
- 13. FRANCHISOR MAKES NO WARRANTIES, EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO, IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE FOR AN ITEM, WORK OR PRODUCTS PROVIDED BY DRPHONEFIX, INC. FRANCHISOR SHALL NOT BE RESPONSIBLE NOR LIABLE FOR ANY DIRECT, SPECIAL, INCIDENTAL, CONSEQUENTIAL OR EXEMPLARY DAMAGES.

THIS ADDENDUM shall be considered an integral part of the Franchise Agreement between the parties. Except as modified or supplemented by this Addendum, all of the rights and obligations of the parties set out in the Franchise Agreement are binding on the parties hereto.

To the extent this Addendum shall be deemed to be inconsistent with any terms or conditions of the Franchise Agreement or Exhibits or Attachments thereto, the terms of this Addendum shall govern.

IN WITNESS WHEREOF, Franchisor has caused these presents to be executed in its name and on its behalf by its proper corporate officer, and Franchisee has hereunder affixed his/her/its hand and seal all on the day and year first above written.

WITNESS	FRANCHISEE (Individual)
(Signature)	(Signature)
	(Printed Name)
WITNESS	
(Signature)	(Signature)
	(Printed Name)
	OR
	FRANCHISEE (Corp., LLC or Partnership)
	(Legal Name of Franchisee Entity)
ATTEST	a (Jurisdiction of Formation) (Corp., LLC or Partnership)
	By: (Name)
	(Signature)
	Its:

ATTEST	FRANCHISOR TECHY, LLC	
	By:	
	Its:	

EXHIBIT "F" TO THE DISCLOSURE DOCUMENT	
FORM OF AREA DEVELOPMENT AGREEMENT	



AREA DEVELOPMENT AGREEMENT

Table of Contents

1.1	Designated Development Area And Development	1
	2.1 Term	
	3.1 Renewal	
	4.1 Development Conditions And Procedures	
	5.1 Fees	
	5.2 Extension Fee.	
	6.1 Operations.	
	7.1 Defaults And Termination.	
	8.1 Transfers	
	9.1 Relationship Of The Parties	
	10.1 Personal Guarantees.	
	11.1 Payments And Legal Compliance.	6
	12.1 Incorporated Provisions Of Techy Franchise Agreement	
	13.1 Miscellaneous Provisions.	
	Exhibit 1.1 Designated Development Area	
	Exhibit 1.2Developmen	ıt Schedule
Obl	Exhibit 1.3 Owner's Guaranty And Assumption Of Business Entity Franchise oligations	ee's

TECHY, LLC AREA DEVELOPMENT AGREEMENT

THIS A	AREA DEVELOPME	ENT AGRE	EMENT (the	: "Agreeme	ent") is made	and entered
into this	day of	_, 20, (the	e "Effective 1	Date") by a	and between	Techy, LLC
("Franchisor", "	'we" or "us") and				, ("Area Do	eveloper," or
"you"), with ref	ference to the following	ng facts:				

- A. Techy, LLC developed a System and methods of operating Franchised Stores that provide, at the retail level, operation of businesses for the repair of cell phones, tablets, computers, game consoles and smart home related hardware and other electronic devices, and the sale of specified electronic devices, accessories and other supplies and items that we may designate from time to time under the trade name Techy. We refer to these businesses as "Techy Stores". We grant franchises for qualified persons to own and operate Techy Stores using the Techy System and the Techy brand name and trademarks.
- B. You desire to have the right and obligation to open a specified number of Techy Stores in a designated territory on or before certain dates. We desire to grant to you such right and obligation, subject to the terms and upon the conditions contained in this Agreement.

You and we agree to the following provisions.

1.1 Designated Development Area and Development

- A. Your Development Area does not include the right to open a Techy Store in Walmart locations within your Development Area. We retain the absolute right to open or franchise to others the right to open a Techy Store in any Walmart location, even if the Walmart location is within your Development Area. We will not award a Techy Franchise for the operation of a Techy Store to anyone other than you, or open a Techy Store owned by us or an Affiliate ("Company Owned Store"), in the area designated on the attached Exhibit 1.1 (the "Designated Development Area") for the term of this Agreement, subject to all of the conditions and provisions of this Agreement.
- B. You commit to sign the then current version of the Techy Franchise Agreement (and all related documents then customarily in use by us), and pay the full Initial Franchise Fees, and to open and operate the number of Techy Stores (the "Development Stores") in the Designated Development Area, in each case by no later than the dates provided on Exhibit 1.2 (the "Development Schedule"). You understand that the then-current form of Techy Franchise Agreement may differ in material respects from our present Franchise Agreement. Our current form of Techy Franchise Agreement has been fully reviewed and approved by you. Terms not defined in this Agreement have the same meanings as in the Franchise Agreement.
- C. Only Techy Stores newly established by you in the Designated Development Area will count to satisfy the requirements of this Agreement and the Development Schedule. Any franchises acquired by transfer or to be located outside the Designated Development Area will <u>not</u> satisfy such requirements.
- D. Your rights in the Designated Development Area are only as expressly stated in 1.1 A, above, and are subject to certain rights reserved by us. You understand and agree that Section 2 of the Franchise Agreement, captioned "Grant and Term", expresses our respective rights in the

Designated Development Area during the term of this Agreement. You expressly acknowledge reading and understanding Section 2 of the Franchise Agreement.

E. We are not granting you with this Agreement the right to sell or to solicit offers to buy Techy Franchises. You represent that all Techy Franchises to be obtained by you are for your sole ownership, use and operation. You will not have the right to "sublet" any area within the Designated Development Area or to sub-franchise or assign any rights under this Agreement.

2.1 <u>Term.</u>

Unless earlier terminated in accordance with its provisions, this Agreement shall commence on the Effective Date indicated on Page One and automatically expire upon the earlier of: (i) the latest Required Date for Opening of a Techy Store or sometimes referred to as "Development Store", provided on Exhibit 2.1; or (ii) the actual Opening Date for the last scheduled Techy Store.

3.1 Renewal.

Area Developer shall have no right to a renewal or successor agreement.

4.1 Development Conditions and Procedures

- A. You are solely responsible for locating, securing and developing the site for each Techy Store according to any then applicable Techy standards, guidelines and/or specifications.
- B. You shall submit to us such information regarding a proposed site as we require, in the form and manner requested by us, together with the terms of any proposed lease. We may seek such additional information that we consider necessary, but our approval will not be unreasonably withheld. The terms of any such lease shall comply with the provisions of the Franchise Agreement applicable to such proposed site. Our consent to any site shall not be construed as a recommendation or warranty as to suitability or the success of the Techy Store to be located there. We make no such assurances of any kind.
- C. Upon our written consent to any approved site, and subject to the conditions to the offer of a franchise provided in 4.1 D., below, you will pay to us the Initial Franchise Fee for each approved Development Store and sign the then current form of Franchise Agreement along with any related documents then customarily used by us, including any appropriate receipts for disclosure documents. We may in connection with our evaluation of the proposed site require and consider financial statements or similar information satisfactory to us demonstrating financial capability and compliance with Section 6.1 C of this Agreement.
- D. Regardless of any other provision of this Agreement, you understand that we have the right to deny you a Franchise and/or not permit you to open a Development Store if any of the following conditions are not met:
- (i) you are not in compliance with any Performance Standards under any Franchise Agreement and/or are in default under this Agreement, any Franchise Agreement, the Manuals, or any other agreement with us and/or any Franchisor Related Person/Entity between you and any other party if related to your Franchise Business;

- (ii) you have not delivered all completed and signed documents currently required by us;
- (iii) you have not appointed a manager for the Development Store who meets our then current training and other standards.
- E. You and we acknowledge and agree that the deadlines set forth in the Development Schedule (Exhibit 1.2) are the <u>essence</u> of this Agreement. No modification or amendment to the Development Schedule or any consent to or waiver of any deadline or other obligation of this Agreement will either (i) create any obligation to grant additional modifications, amendments, consents or waivers or (ii) be effective unless made by mutual written agreement. Any modification or amendment to the Development Schedule or otherwise will be in our Business Judgment, as that term is defined in the Franchise Agreement, and may be subject to conditions, including (but not limited to) a General Release or a reduction of the Designated Development Area and Development Stores.

5.1 <u>Fees</u>.

The Initial Franchise Fee under the Area Development Agreement for the first Techy Store is \$15,000, for the second Techy Store the discounted Initial Franchise Fee is \$10,000, the discounted Initial Franchise Fee for the third and each additional Techy Store is \$5,000. Upon execution of this Area Development Agreement you must pay a lump sum non-refundable Area Development Fee of \$15,000 for the first Techy Store plus one-half of the Initial Franchise Fee for each discounted Techy Store required to be developed as set forth in the Exhibit 1.2 "Development Schedule." By your signature below you promise to open and operate a total of ______ Techy Stores and to pay an Area Development Fee of \$______ when you sign this Area Development Agreement and to pay the remaining additional balance of \$______ as set forth in the following chart for a total Area Development Fee of \$______ :

		Amount Paid On	
Number of Franchise		Signing	Balance Due At the time you sign
Stores Required	Total	Development	Each of the Remaining Franchise
To Be Opened	Development Fee	Agreement	Agreements
Two	\$25,000	\$20,000	0(1) + 5k(2)
Three	\$30,000	\$22,500	0(1) + 5k(2) + 2.5k(3)
Four	\$35,000	\$25,000	0(1)+5k(2) + 2.5k(3) + 2.5k(4)
Five			\$0(1)+\$5k(2)+\$2.5k(3)+\$2.5k(4)+
1.116	\$40,000	\$27,500	\$2.5k(5)

Note that the second franchise has a reduced fee of \$10,000 and when added to the \$15,000 fee for the first franchise makes a total Development Fee of \$25,000 for the two stores. However, only half of the fee for the second franchise is paid when signing the Development Agreement. The remaining half (\$5,000) is paid upon signing the franchise agreement to open the second franchise.

For each additional area development franchise in excess of 5 Franchise Stores, the Initial Franchise
Fee will be \$5,000 of which \$2,500 will be added to the Development Fee and the balance of \$2,500
will be payable upon signing each additional Franchise Agreement in excess of 5 franchises.

Signature	Printed Name and Date

You agree that the grant of rights to you under this Area Development Agreement constitutes the sole consideration for the payment of your Area Development Fee and that the total Area Development Fee is fully earned by us upon your signature to this Area Development Agreement.

5.2 Extension Fee.

If you fail to comply with your minimum Development Schedule set forth in Exhibit 1.2, we may, in our sole discretion, grant an extension of your Development Schedule for a non-refundable extension fee of from \$5,000 to \$20,000 per remaining franchise to be opened, depending upon the amount of time extended.

6.1 Operations.

- A. We require the Chief Executive Officer or another officer/owner for or of the Area Developer to attend and successfully complete such initial and/or on-going training as we designate and/or to meet with us from time-to-time to review Area Developer operations.
- B. We require that you employ a manager for each Development Store who will meet such standards (including training requirements) as are set forth in the then-current Franchise Agreement and/or Manuals issued by us, to supervise and coordinate the operation of each Techy Store.

7.1 <u>Defaults and Termination.</u>

- A. This Agreement may be terminated by us upon your receipt of written notice and without opportunity to cure, except as may be required by law, if:
- (1) You attempt to sell, assign, Transfer or encumber in whole or in part any or all rights and obligations under this Agreement in conflict with Article 8.1, below;
- (2) You fail to meet on a timely basis any of the provisions of the Development Schedule, including without limitation, any Development Store Opening Date or Fee Payment;
- (3) You commit to a lease unapproved by us, or begin the development and/or operation of a Development Store, without having complied with the terms of this Agreement, including without limitation Article 4.1, above;
- (4) You or any of your owners are judged bankrupt, become insolvent, make an assignment for the benefit of creditors, are unable to pay his/her/their debts as they become due, or a petition under any bankruptcy law is filed against you or any of your owners or a receiver or other custodian is appointed for a substantial part of your assets or the assets of any owner;
- (5) You or any of your owners is convicted of, or pleads no contest to, a felony or to any crime or offense that may adversely affect the good will associated with the Marks;
 - (6) You commit any other breach of this Agreement.

- B. Any default by you under this Agreement may be regarded by us as a default by you under the Franchise Agreement. Any default under any Franchise or other agreement with us or with any Franchisor Related Person/Entity may be considered by us to be a default under this Agreement. Any default by you under any agreement, lease or other obligation relating to the Franchise Business may be regarded as a default under this Agreement, regardless of whether or not any such agreements are between you and us. In all instances we shall be entitled to all rights and remedies available to us under the respective agreements, at law and in equity. If you default under any Franchise or other agreement, we may require you to stop all development activities/Development Store openings under this Agreement unless and until you are permitted by us in writing to continue.
- C. Upon a termination/expiration of this Agreement you have no rights under this Agreement, including without limitation, no right to the award of any further franchises or any refunds of any amounts paid. We shall be free to develop the Territory in any manner we choose, including establishing Company Owned Stores and franchised units in the Territory, subject only to the terms of any applicable unit franchise agreements.
- D. Notwithstanding the provisions of Section 7.1 A (2), above, if the opening of a Development Store is physically prevented by circumstances beyond human control, such as fire, flood, earthquake, riot, war, or other similar circumstance, then you will be allowed such additional time as is reasonably necessary to open such Development Store, but not longer than six months. Such an extension shall be available exclusively to the Store subject to the interrupted development and shall not apply to any subsequent unit development deadlines or requirement dates.

8.1 Transfers

- A. This Agreement is personal to you and based upon individual skills, resources, special qualities and characteristics and is not assignable, whether voluntarily or by operation of law, without our express written consent, which we may grant, condition or withhold in our Business Judgment. We may choose among other things to apply the provisions applicable to transfers as contained in the most current Techy Franchise Agreement between us, including those regarding Rights of First Refusal, to any transfer of this Agreement. Any assignment by you must be accompanied by a concurrent assignment to the same assignee of all of your interests in each Development Store and each related Techy Franchise Agreement. Any consent to transfer by us will be conditioned upon our receipt of a nonrefundable transfer fee of \$10,000 in addition to any such fees due under the applicable unit Franchise Agreements. This area development transfer fee may be reduced or waived by us in our sole Business Judgment.
- B. This Agreement is assignable by us, in whole or in part, without your consent and shall inure to the benefit of our successors and assigns. We have no liability to you upon such an assignment.
- C. This Agreement shall not be deemed to diminish in any way any rights of first refusal and/or rights of repurchase held by us under any Techy Franchise Agreement with you, whether effective now or in the future. If we purchase from you substantially all of the operating assets of the Development Stores, or all of their franchises are terminated or repurchased, you will have no rights, and we will have no obligations, under this Area Development Agreement.

9.1 Relationship of the Parties.

You and we are independent contractors. Nothing in this Agreement is intended to make either you or us a general or special agent, joint venturer, partner, or employee of the other for any purpose. There is no fiduciary, trust or other special relationship between us. Expenses, obligations, taxes and other liabilities incurred by you in connection with your performance of the services contemplated by this Agreement shall be your sole responsibility. Neither you nor we shall act as the agent of the other, nor guarantee or become in any way responsible for the obligations, debts or expenses of the other.

10.1 Personal Guarantees.

Each of the owners signing this Agreement hereby personally guarantees, jointly and severally, the full payment and performance of each and all of the Area Developer's obligations under this Agreement.

11.1 Payments and Legal Compliance.

- A. You will promptly pay when due any and all taxes, accounts, liabilities and other indebtedness of every kind incurred by you as a result of your acts or omissions under or associated with this Agreement, any Franchise Agreement, the operation of any Techy Store or otherwise and will hold harmless and indemnify us and each of the Franchisor-Related Persons/Entities from such taxes, accounts, liabilities and other indebtedness.
- B. You will comply with all federal, state and local laws and regulations and shall timely obtain any and all permits, certificates or licenses necessary for the activities contemplated under this Agreement and/or any Franchise Agreement.

12.1 Incorporated Provisions of Franchise Agreement.

As provided in Section 1.1 and Exhibit 1.2, you are, or shall become, concurrent with the signing of this Agreement, a Techy Franchisee under the terms of a Techy Franchise Agreement substantially similar to the current Techy Franchise Agreement. The following provisions of the Techy Franchise Agreement shall apply to this Agreement in the following manner:

- A. <u>Licensed Marks</u>. The provisions of Section 6 of the Techy Franchise Agreement shall apply to the use of the Marks under this Agreement and are incorporated herein.
- B. <u>Relationship and Indemnification</u>. You and we are independent contractors, as provided in Section 9.1, above. Therefore, the provisions of Sections 7.1 through 7.5 of the Techy Franchise Agreement shall apply in the same manner to you in your Area Development activities as they do to your Techy Store Franchise Business and are incorporated herein. References made in such Sections to your Techy Store Franchise and Franchise Business operations shall be deemed to include your operations under this Agreement, including without limitation the indemnification obligations of Section 7.4 of the Techy Franchise Agreement.
- C. <u>Confidential Information; Exclusive Relationship</u>. The provisions of Section 8 of the Techy Franchise Agreement shall apply in the same manner to you in your Area Development activities as they do to your Techy Franchise Business and are incorporated herein. References

made in Section 8 to your Techy Franchise Agreement and Franchise Business operations shall be deemed to include your operations under this Agreement.

- D. <u>Records and Reports.</u> The provisions of Section 12.2 of the Techy Franchise Agreement and the reporting and record requirements described therein shall apply in the same manner to your Area Development activities as they do to your Techy Franchise Business and are incorporated herein.
- E. <u>Dispute Avoidance and Resolution, Notice and Acknowledgments</u>. The provisions of Sections 19, 20 and 21 of the Techy Franchise Agreement shall apply to this Agreement and are incorporated herein, including without limitation those providing for mediation and mandatory binding arbitration, waiver of jury trial, limitations on damages and venue. The parties (and each of their owners) waive all rights **to a jury trial**.
- F. <u>Rights on Termination, Expiration, Transfer, and/or Repurchase</u>. The provisions of Section 17 of the Techy Franchise Agreement shall apply to this Agreement and are incorporated herein.

13.1 Miscellaneous Provisions.

- A. Whole Agreement; Amendment. This Agreement contains the final, complete and exclusive expression of the terms of your and our agreement with respect to the subject matter, and supersedes and replaces any and all prior and/or concurrent understandings, agreements, representations or otherwise (whether oral or written) between you and us; provided that this Agreement is not intended to amend or supersede any Techy Franchise Agreement and/or related documents to which you are a party. This Agreement cannot be modified or changed except by written document signed by all parties. Notwithstanding anything to the contrary herein, nothing in this Agreement or any related agreement is intended to disclaim the representations made in our franchise disclosure document provided to you.
- B. <u>Severability</u>. Whenever there is any conflict between any provisions of this Agreement and any present or future statute, law, ordinance or regulation contrary to which the parties have no legal right to contract, the latter shall prevail; but in such event the provisions of this Agreement thus affected shall be limited only to the extent necessary to bring them within the requirements of the law. In the event that any part, article, paragraph, sentence or clause of this Agreement shall be held to be indefinite, invalid or otherwise unenforceable, the indefinite, invalid or unenforceable provision shall be deemed deleted, and the remaining part of this Agreement shall continue in full force and effect.
- C. <u>Binding Agreement</u>. This Agreement shall be binding upon and inure to the benefit of the successors and assigns of Franchisor, and shall be binding upon and inure to the benefit of you and your respective heirs, executors, administrators, successors and assigns, subject to the prohibitions against assignment contained herein.
- D. <u>Joint and Several Liability</u>. If "you" consist of more than one person or entity, or a combination thereof, the obligations and liabilities of each such person or entity to us are joint and several.
- E. <u>No Waiver</u>. No waiver by any party of any breach or series of breaches or defaults in performance, shall be deemed to be a waiver of any other breach or default; and no failure,

refusal or neglect of ours to exercise any right, power or option given to us under this Agreement shall be deem to be a waiver of any right, power or option.

- F. <u>Acknowledgments</u>. In addition to those acknowledgments contained in the Franchise Agreement and incorporated by reference pursuant to Section 12.1 E, above, Area Developer and each of its owners expressly acknowledge that:
- (1) Neither the Area Developer, any of the owners nor anyone else have received or relied on (nor have we or anyone else provided) any <u>oral or written</u>: sales, income or other projections of any kind or nature or any statements, representations, or otherwise which stated or suggested any level or range of actual or potential sales, costs, income, expenses, profits, cash flow, or otherwise with respect to any Techy Store or Area Development rights. <u>We cannot reliably predict, forecast or project future performance, revenues, profits or otherwise of any Techy Store or with respect to any Area Development rights</u>. If any such information, promises, representations and/or warranties has been provided to Area Developer, any of the owners or anyone else, they haven't been authorized and may not be relied on; and
- (2) A complete ready-to-sign copy of this agreement as signed by Area Developer was received by Area Developer (and each of the owners, as applicable) at least seven calendar days prior to the earlier of its execution, or the payment of any amounts; and
- (3) A complete copy of the Techy Franchise Disclosure Document, together with all exhibits, was received at least fourteen calendar days prior to the earlier of Area Developer (and each of the owners, as applicable) signing this agreement or the paying of any amounts.

G. <u>Notices</u>. All written notices and reports permitted or required to be delivered by the provisions of this Agreement will be deemed so delivered at the time delivered by hand, immediately on transmission by facsimile transmission or other electronic system, including e-mail or any similar means, one (1) business day after being placed in the hands of a commercial courier service for overnight delivery, or three (3) business days after placement in the United States Mail by Registered or Certified Mail, Return Receipt Requested, postage prepaid and addressed to us at Techy, LLC, 3000 South West 4th Avenue, Fort Lauderdale, Florida 33315, (or our then-current headquarters), to the attention of our Executive Manager, and to you at the following address:

All parties have obtained the advice of counsel in connection with entering into this Agreement, understand the nature of this Agreement, and intend to be bound by its terms.

[Signatures on page that follows.]

FRANCHISOR:

TECHY, LLC A Florida Limited Liability Com	pany
By:	
Its	Date —
AREA DEVELOPER (Individua	1)
Signature	Date
Printed Name	
Signature	Date
Printed Name	_
OR	
AREA DEVELOPER (Corp., LI must be accompanied by approp	
Legal Name of Franchisee Entity	
a	
Jurisdiction of Formation	Corporation, LLC or Partnership
By:Name	<u></u>
Signature	Date
Title:	
OWNERS	

EXHIBIT 1.1 DESIGNATED DEVELOPMENT AREA

The State of	, United States of America
of the boundary no rights under boundary line, s may be, regardl	ry lines include only the area within the boundary line and extend only to the middle demarcation (for example, only to the middle of a street or highway.) You have this Agreement or otherwise with respect to a facility on the other side of the treet or highway or otherwise, and no matter how close to such boundary a facility ess of the distance from, impact on, or vicinity of, your Techy Development Store Techy Stores, other outlets or otherwise in any area or market.
	Describe Development Area below:
Initials	

EXHIBIT 1.2 DEVELOPMENT SCHEDULE

Stores	Required Dates for Signin	ng	Required Date for
to be	of Separate Franchise Ag	Opening of Techy Stores	
<u>Developed</u>	and Related Documents a		
	Payment of Development	Fees	
	in Full		
_			
1			
•			
2			
3			
3			
4			
5			
6			
_			
7			
O			
8			
9			
,			
10			
The Techy Stor	res in the foregoing list are	the Techy "De	evelopment Stores."
·		·	1
Initials:	_		
	_		

EXHIBIT 1.3

OWNER'S GUARANTY AND ASSUMPTION OF BUSINESS ENTITY FRANCHISEE'S OBLIGATIONS

In consideratio	on of, and as an inducement to, the	execution by Techy, LLC, ("Franchis	or")
of the Area Developme	ent Agreement of even date herewi	ith (the "Agreement") between Franch	isor
and	, a(n)	(the "Area Developer"), each of	the
		jointly and severally: (1) guarantees	
Franchisor, its affiliates	s, the Franchisor-Related Persons/	Entities (as defined in the Agreement)	and
each of their successor	s and assigns, for the term of the	Agreement and thereafter as provide	d in
the Agreement, that the	e undersigned will be bound by, a	and punctually pay and perform, each	and
every undertaking, agre	eement and covenant set forth in the	he Agreement; (2) agrees to be person	ally
bound by, and personal	lly liable for, the breach of, each a	and every provision in the Agreement;	and
(3) agrees to be perso	onally bound by, and personally	liable for, each obligation of the A	Area
Developer to Franchiso	or and/or any company affiliated of	r related in any way with or to Franchi	sor,
including all past, cur	rrent and/or future obligations of	of the Area Developer, the undersig	ned
		I without limitation in scope, nature and	
effect. Franchisor (and	I/or its affiliates) need not bring su	uit first against the undersigned in orde	er to
enforce this guarantee	and may enforce this guarantee a	against any or all of the undersigned a	as it
chooses in its Business	Judgment.		

Each of the undersigned waives:

- (1) acceptance and notice of acceptance by Franchisor, of the foregoing undertakings;
- (2) notice of demand for payment of any indebtedness or nonperformance of any obligations hereby guaranteed;
- (3) protest and notice of default to any party with respect to the indebtedness or nonperformance of any obligations hereby guaranteed;
- (4) any right the undersigned may have to require that an action be brought against Franchisor, Area Developer or any other person as a condition of liability; and
- (5) any and all other notices and legal or equitable defenses to which he or she may be entitled.

Each of the undersigned consents and agrees that:

- (1) his or her direct and immediate liability under this guaranty will be joint and several;
- (2) he and/or she will render any payment or performance required under the Agreement on demand if the Area Developer fails or refuses to do so punctually;
- (3) such liability will not be contingent or conditioned on pursuit by Franchisor of any remedies against the Area Developer or any other person;

- (4) such liability will not be diminished, relieved or otherwise affected by any extension of time, credit or other indulgence which Franchisor may from time-to-time grant to the Area Developer or to any other person, including without limitation the acceptance of any partial payment or performance, or the compromise or release of any claims, none of which will in any way modify or amend this guaranty, which will be continuing and irrevocable during the term of the Agreement;
- (5) the liabilities and obligations of the undersigned, whether under this document or otherwise, will not be diminished or otherwise affected by the termination, rescission, expiration or otherwise of the Agreement; and
- (6) the provisions of Articles 19, 20 and 21 of the Agreement are incorporated in and will apply to this document as if fully set forth herein and shall apply to any dispute involving the Franchisor and any of the undersigned.

In connection with such guarantee and the Franchisor (a) not requiring that the Franchise be initially awarded in the name of one or more of the Guarantors and/or (b) not requiring the payment of a full transfer fee in connection with any related transfer from the undersigned to the Area Developer, each of the undersigned hereby grants a General Release of any and all claims, liabilities and/or obligations, of any nature whatsoever, however arising, known or unknown, against the Franchisor and/or any or all of the Franchisor-Related Persons/Entities.

IN WITNESS WHEREOF, each of the undersigned has here unto affixed his or her signature on the same day and year as the Agreement was executed.

GUARANTOR(S)	PERCENTAGI OF Area Develo	E OF OWNERSHIP oper
		%
		%
		%
		%
Area Developer:		
, a	·	
Ву		
Its		
Franchise Agreement Number:		

EXHIBIT F-1

IN-STORE (WALMART) ADDENDUM

FOR

TECHY, LLC

AREA DEVELOPMENT AGREEMENT

THIS ADDENDUM is entered into this the day of	, 20, by and
between Techy, LLC (hereinafter referred to as "Franchisor"), and	,
(hereafter referred to as "Area Developer" or "You").	

WITNESSETH:

WHEREAS, Area Developer and Franchisor are contemporaneously entering into an Area Development Agreement; and

WHEREAS, Area Developer desires to develop Franchisor's In-Store (Walmart) franchises and enter into Franchisor's Franchise Agreement and In-Store (Walmart) Addendum for each In-Store location to be developed; and

WHEREAS, Area Developer understands and agrees that when a Franchise Agreement and In-Store (Walmart) Addendum are entered into for each In-Store (Walmart) location that a separate sublease agreement for the In-Store space within a Walmart store location must be executed with Franchisor's affiliate, DrPhoneFix Inc., and additionally, that you must engage DrPhoneFix, Inc. to perform Start-Up Package services to prepare the subleased premises in a ready to use condition.

NOW, THEREFORE, in consideration of the mutual and several covenants and obligations contained herein, the parties hereto agree that this In-Store (Walmart) Addendum to the parties Area Development Agreement shall be attached to the executed Area Development Agreement and shall partially amend and revise the parties Area Development Agreement as follows:

- 1. In **Paragraph A** of the Preamble to the Area Development Agreement, Area Developer acknowledges and agrees that the stores franchised under this concept shall also be known as In-Store (Walmart) franchises and In-Store (Walmart) franchise locations.
- 2. Notwithstanding anything to the contrary in **Paragraph B** of the Preamble to the Area Development Agreement, Area Developer desires to have the right and obligation to open a specified number of Techy In-Store (Walmart) franchise locations (rather than Techy Stores) in a designated territory on or before certain dates.
- 3. In **Paragraph 1.1 A.** of the Area Development Agreement, notwithstanding anything to the contrary, Area Developer is granted a limited right to open Techy In-Store (Walmart) franchises at locations within the Designated

Development Area and such In-Store (Walmart) franchises will only be located within Walmart stores within the Designated Development Area.

- 4. In Paragraph 1.1 B. of the Area Development Agreement, notwithstanding anything to the contrary, you commit to sign the then current version of Franchisor's Franchise Agreement and In-Store (Walmart) Addendum to Techy, LLC Franchise Agreement (and all related documents then customarily in use by Franchisor) and pay the full Initial Franchise Fees, sign a sublease agreement with DrPhoneFix, Inc. and engage DrPhoneFix, Inc. to perform Start-Up Package services to prepare the subleased premises for each location in a ready to use condition, and open and operate the number of Techy In-Store (Walmart) franchise locations by no later than the dates provided in Exhibit 1.2 of the Area Development Agreement, herein the "Development Stores" and "Development Schedule". Our current form of Techy Franchise Agreement and Addendum has been fully reviewed and approved by Area Developer. Terms not defined in this Addendum have the same meaning as the Area Development Agreement and In-Store (Walmart) Addendum for Techy, LLC Area Development Agreement.
- 5. **Paragraph 1.1 C.** is amended in its entirety to read as follows:
 - **1.1 C.** Only Techy In-Store (Walmart) franchise locations, newly established by you in the Designated Development Area will count to satisfy the requirements of this Agreement and the Development Schedule. Any franchises acquired by transfer or to be located outside the Designated Development Area will not satisfy such requirements.
- 6. In **Paragraph 4.1 A.** you must enter into a sublease agreement with DrPhoneFix, Inc. for the In-Store (Walmart) franchise premises.

THIS ADDENDUM shall be considered an integral part of the Area Development Agreement between the parties. Except as modified or supplemented by this Addendum, all of the rights and obligations of the parties set out in the Area Development Agreement are binding on the parties hereto.

To the extent this Addendum shall be deemed to be inconsistent with any terms or conditions of the Area Development Agreement or Exhibits or Attachments thereto, the terms of this Addendum shall govern.

[Signatures on page that follows.]

IN WITNESS WHEREOF, Franchisor has caused these presents to be executed in its name and on its behalf by its proper corporate officer, and Franchisee has hereunder affixed his/her/its hand and seal all on the day and year first above written.

WITNESS	AREA DEVELOPER (Individual)
(Signature)	(Signature)
	(Printed Name)
WITNESS	
(Signature)	(Signature)
	(Printed Name)
	AREA DEVELOPER (Corp., LLC or Partnership)
	(Legal Name of Franchisee Entity)
ATTEST	a(Jurisdiction of Formation) (Corp., LLC or Partnership) By:(Name)
	(Signature) Its:
	(Title) FRANCHISOR TECHY, LLC
ATTEST	· -
	Its:

EXHIBIT "G" TO THE DISCLOSURE DOCUMENT

LIST OF STATE AGENCIES/AGENTS FOR SERVICE OF PROCESS

State Agencies/Agents for Service of Process				
STATE	ADDRESS	PHONE		
California - Los Angeles Department of Financial Protection and Innovation	320 West 4th Street, Suite 750 Los Angeles, CA 90013-2344	213-576-7500		
California – Sacramento Department of Financial Protection and Innovation	2101 Arena Boulevard Sacramento, CA 95814-4017	916-445-7205		
California - San Diego Department of Financial Protection and Innovation	1350 Front Street, Room 2034 San Diego, CA 92108	619-525-4233		
California - San Francisco Department of Financial Protection and Innovation	One Sansome Street, Suite 600 San Francisco, CA 94104-4428	415-972-8565		
Hawaii Commissioner of Securities	335 Merchant Street, Room 203 Honolulu, HI 96813-2921	808-586-2722		
Illinois Illinois Attorney General	500 South Second Street Springfield, IL 62706	217-782-4465		
Indiana Secretary of State	200 West Washington Street, Room 201 Indianapolis, IN 46204	317-232-6681		
Maryland Administrator: Office of the Attorney General Securities Division	200 St. Paul Place, 20th Floor Baltimore, MD 21202-2020	410-576-6360		
Agent for Service of Process Maryland Securities Commissioner	200 St. Paul Place Baltimore, MD 21202-2020			
Michigan Department of Commerce, Corporations and Securities Bureau	525 W. Ottawa Street Lansing, MI 48913	517-373-7117		
Minnesota Minnesota Commissioner of Commerce	85 7th Place East, Suite 280 St. Paul, MN 55101-2198	651-539-1600		
New York Administrator: New York State Department of Law Investor Protection Bureau	28 Liberty Street, 21st Floor New York, NY 10005	212-416-8222		
Agent for Service of Process Secretary of State	99 Washington Avenue Albany, NY 1223	212-416-8211		
North Dakota North Dakota Securities Commissioner	600 East Boulevard Avenue, State Capital, 5th Floor Bismarck, ND 58505	701-328-2910		
Rhode Island Department of Business Regulation	1511 Pontiac Avenue Cranston, RI 02920	401-462-9587		
South Dakota Director of Division of Securities	124 S. Euclid Ave., Suite 104 Pierre, SD 57501-3185	605-773-4823		
Virginia Clerk of the State Corporation Commission	1300 E. Main Street, 1st Floor Richmond, VA 23219-3630	804-371-9051		
Washington Administrator Dept. of Financial Institutions Securities Division Agent for Service of Process Director of Dept. of Financial Institutions	P.O. Box 9033 Olympia, WA 98501 150 Israel Road, SW Tumwater, WA 98501	360-902-8760		
Wisconsin	345 W. Washington Ave., 4th Floor Madison, WI 53703	608-266-8557		

EXHIBIT "H" TO THE DISCLOSURE DOCUMENT

STATE SPECIFIC AND OTHER ADDENDA AND RIDERS

CALIFORNIA ADDENDUM

The following modifications are to the Techy, LLC Franchise Disclosure Document, Franchise Agreement and Area Development Agreement and may supersede, to the extent then required by valid applicable state law, certain provisions of the Franchise Agreement.

THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE DISCLOSURE DOCUMENT.

- 1. California Business and Professions Code Sections 20000 through 20043 provide rights to you concerning termination, transfer or non-renewal of a franchise. If the Franchise Agreement or Area Development Agreement contains provisions that are inconsistent with the law, the law will control.
- 2. The Franchise Agreement provides for termination upon bankruptcy. This provision may not be enforceable under Federal Bankruptcy Law (11 U.S.C.A. Sec. 101 et seq.).
- 3. The Franchise Agreement contains covenants not to compete which extend beyond the termination of the agreements. These provisions may not be enforceable under California law.
- 4. Section 31125 of the California Corporation Code requires the franchisor to provide you with a disclosure document before asking you to agree to a material modification of an existing franchise.
- 5. Neither the franchisor, any person or franchise broker in Item 2 of the Disclosure Document is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C.A. 79a et seq., suspending or expelling such persons from membership in such association or exchange.
- 6. You must sign a general release if you renew or transfer your franchise. California Corporation Code 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code 31000 through 31516). Business and Professions Code 20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code 20000 through 20043).
- 7. Prospective franchisees are encouraged to consult private legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5 Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of a franchise agreement restricting venue to a forum outside the State of California.
- 8. You will not receive an exclusive territory. You may face competition from other channels of distribution or competitive brands that we control.
- 9. OUR WEBSITE, http://www.TechyCompany.com HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF

FINANCIAL PROTECTION AND INNOVATION. ANY COMPLAINTS CONCERNING THE CONTENT OF THIS WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION AT www.dfpi.ca.gov.

- 10. In Item 6, the maximum interest allowed in California is 10% per annum. Item 6 is modified in California to comply with California law and charging a maximum of 10% per annum for "Late Payment Charges".
- 11. The Franchise Agreement contains a liquidated damage clause. Under California Civil Code Section 1671, certain liquidated damage clauses are unenforceable.
- 12. The Franchise Agreement requires binding arbitration. The arbitration will occur in Broward County, Florida with cost being borne by each respective party unless the claim is for amounts owed by franchisee to franchisor, in which case all cost and expenses relating to the claim shall be borne by franchisee if franchisor prevails on the claim. Prospective franchisees are encouraged to consult private legal counsel to determine the applicability of California and federal laws (such as Business and Professional Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provision of a franchise agreement restricting venue to a forum outside the State of California.
- 13. The Franchise Agreement requires application of the laws of Florida. This provision may not be enforceable under California law.

It is agreed that the applicable foregoing state law Addendum, supersede any inconsistent portion of the Disclosure Document or Franchise Agreement, but only to the extent they are then valid requirements of an applicable and enforceable state law, and for only so long as such state law remains in effect.

Dated this day of, 20		
TECHY, LLC A Florida limited liability company	FRANCHISEE	
Ву:	(Signature)	
Title:	(Print Name)	

ILLINOIS ADDENDUM TO TECHY, LLC FRANCHISE AGREEMENT

This addendum to the Franchise Agreement is agreed to by and between Techy, LLC and the Franchisee identified below, to amend and revise said Franchise Agreement as follows:

- 1. Illinois law governs the relationship between the parties to the franchise.
- 2. In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in a franchise agreement that designates jurisdiction and venue in a forum outside of the State of Illinois is void. However, a franchise agreement may provide for arbitration to take place outside of Illinois.
- 3. Your rights upon Termination and Non-Renewal are set forth in sections 19 and 20 of the Illinois Franchise Disclosure Act.
- 4. In conformance with section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.
- 5. Notwithstanding anything heretofore in the Franchise Agreement, nothing in any agreement is intended to disclaim the express representations made in the Franchise Disclosure Document, its exhibits and amendments.

To the extent this Addendum shall be deemed to be inconsistent with any term or conditions of said Franchise Agreement or Exhibits or attachments thereto, the terms of the Illinois Franchise Disclosure Act as stated in this Addendum shall govern.

	ndersigned hereby acknowledges having read this nd by all of its terms, and agrees it shall become
TECHY, LLC A Florida limited liability company	FRANCHISEE
By:	(Signature)
Title:	(Print Name)

ILLINOIS ADDENDUM TO TECHY, LLC AREA DEVELOPMENT AGREEMENT

This addendum to the Area Development Agreement is agreed to by and between Techy, LLC and the Franchisee identified below, to amend and revise said Area Development Agreement as follows:

- 1. Illinois law governs the relationship between the parties to the franchise.
- 2. In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in the Area Development Agreement that designates jurisdiction and venue in a forum outside of the State of Illinois is void. However, an area development agreement may provide for arbitration to take place outside of Illinois.
- 3. Your rights upon Termination and Non-Renewal are set forth in sections 19 and 20 of the Illinois Franchise Disclosure Act.
- 4. In conformance with section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.
- 5. Notwithstanding anything heretofore in the Area Development Agreement, nothing in any agreement is intended to disclaim the express representations made in the Franchise Disclosure Document, its exhibits and amendments.

To the extent this Addendum shall be deemed to be inconsistent with any term or conditions of said Area Development Agreement or Exhibits or attachments thereto, the terms of the Illinois Franchise Disclosure Act as stated in this Addendum shall govern.

IN WITNESS WHEREOF, each of the undersigned hereby acknowledges having read this Addendum, understands and consents to be bound by all of its terms, and agrees it shall become effective the ____ day of ______, 20__.

TECHY, LLC
A Florida limited liability company

By:______

(Signature)

Title:______

(Print Name)

ILLINOIS ADDENDUM TO TECHY, LLC DISCLOSURE DOCUMENT

1. Your rights upon termination and non-renewal of a Franchise Agreement are set forth in sections 19 and 20 of the Illinois Franchise Disclosure Act.

Section 4 of the Illinois Franchise Disclosure Act provides that any provision in a Franchise Agreement that designates jurisdiction or venue outside of the State of Illinois is void. However, a Franchise Agreement may provide for arbitration in a venue outside of Illinois.

2. Illinois law governs the relationship between the parties to this franchise.

Section 41 of the Illinois Franchise Disclosure Act provides that any condition, stipulation, or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.

MARYLAND ADDENDUM TO TECHY, LLC FRANCHISE AGREEMENT

This addendum to the Franchise Agreement is agreed to by and between TECHY, LLC and the Franchisee identified below, to amend and revise said Franchise Agreement as follows:

The Franchise Agreement is amended by the addition of the following language to the original language that appears therein:

Any limitation on the period of time arbitration and/or litigation claims must be brought shall not act to reduce the 3 year statute of limitations afforded a franchisee for bringing a claim arising under the Maryland Franchise Registration and Disclosure Law. Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

Any general release required as a condition of renewal, sale, and/or assignment/transfer shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law.

The acknowledgments or representations of the franchisee made in the franchise agreement and questionnaire which disclaim the occurrence and/or acknowledge the non-occurrence of acts that would constitute a violation of the Franchise Law are not intended to nor shall they act as a release, estoppels or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.

To the extent this Rider shall be deemed to be inconsistent with any term or conditions of said Franchise Agreement or Exhibits or attachments thereto, the terms and conditions as stated in this Rider shall govern.

The provision in the franchise agreement which provides for termination upon bankruptcy of the franchisee may not be enforceable under federal bankruptcy law (11 U.S.C. Section 101 et seq.).

Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

Notwithstanding anything heretofore in the Franchise Agreement, nothing in any agreement is intended to disclaim the express representations made in the Franchise Disclosure Document, its exhibits and amendments.

IN WITNESS WHEREOF, each of the undersigned hereby acknowledges having read this Rider, understands and consents to be bound by all of its terms, and agrees it shall become effective the ____ day of ______, 20___.

TECHY, LLC
A Florida limited liability company

By:______

(Signature)

Title:_____

(Print Name)

MARYLAND ADDENDUM TO TECHY, LLC AREA DEVELOPMENT AGREEMENT

	addendum to the Area Developmented, 20	ment Agreement by and between Techy, LLC and
1. The Ar	rea Development Agreement is m	nodified by the inclusion of the following language:
	the Maryland Franchise Regist	ce of forum is not applicable to claims arising under tration and Disclosure Law. Developer may sue in under the Maryland Franchise Registration and
2. The ap		Development Agreement are amended to include the
		or waiver of liability required as a condition of nment/transfer shall not apply to any liability under ration and Disclosure Law.
	franchisees to assent to any rele purchasing a franchise. Our Fr the occurrence and/or acknowle a violation of the Franchise Law representations are not intended	osure Law prohibits us from requiring prospective case, estoppel or waiver of liability as a condition of anchise Agreement requires franchisees to disclaim edge the non-occurrence of acts that would constitute in order to purchase your franchise. Therefore, such to nor shall they act a s a release, estoppel or waiver the Maryland Franchise Registration and Disclosure
agreement is in		n the Area Development Agreement, nothing in any representations made in the Franchise Disclosure
them in the Fra		ed herein shall have the same meaning as ascribed to pressly modified hereby, the agreements shall remain ms.
		hereto have duly executed and delivered this Rider Agreement on the day and year first above written.
FRANCHISO TECHY, LLC A Florida limite		DEVELOPER
By:		(Signature)
Title		

(Print Name)

MARYLAND ADDENDUM TO TECHY, LLC DISCLOSURE DOCUMENT

Anything to the contrary in the Techy, LLC disclosure document, the following provisions shall supersede and apply to all franchises offered and sold in the State of Maryland:

Item 17 is amended to state:

"Any general release required as a condition of renewal, sale, and/or assignment/transfer shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law."

Item 17 is amended to add:

"A franchisee may bring a lawsuit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law."

The provision in the franchise agreement which provides for termination upon bankruptcy of the franchisee may not be enforceable under federal bankruptcy law (11 U.S.C. Section 101 et seq.).

Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

STATE AMENDMENT TO THE FRANCHISE AGREEMENT TECHY, LLC PURSUANT TO THE MINNESOTA FRANCHISE DISCLOSURE LAW

	This Rider to the Franchise	Agreement by	and between	TECHY,	LLC and	Franchisee i	İS
dated _	, 20						

1. The Franchise Agreement is amended by the inclusion of the following:

The franchisor will protect the franchisee's right to use the trademarks, service marks, trade names, logotypes or other commercial symbols or indemnify the franchisee from any loss, costs or expenses arising out of any claim, suit or demand regarding the use of the name.

2. The Franchise Agreement is amended by the inclusion of the following:

With respect to franchises governed by Minnesota law, the franchisor will comply with Minn. Stat. Sec. 80C, 14 Subds. 3, 4, and 5 which require, except in certain specified cases, that a franchisee be given 90 days notice of termination (with 60 days to cure) and 180 days notice for non-renewal of the franchise agreement.

3. The Franchise Agreement is supplemented by the inclusion of the following:

Minn. Stat. § 80C.21 and Minn. Rule 2860.4400J prohibit us from requiring litigation to be conducted outside Minnesota. In addition, nothing in the DISCLOSURE DOCUMENT or agreement can abrogate or reduce any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction.

4. The Franchise Agreement is amended by the inclusion of the following:

Under Minn. Rule 2860.4400J, a franchisee cannot waive any rights, the franchisee cannot consent to the franchisor obtaining injunctive relief, although the franchisor may <u>seek</u> injunctive relief and a franchisee cannot be required to consent to waiver of a jury trial. In addition, a court will determine if a bind is required.

5. The Franchise Agreement is amended by the inclusion of the following:

Under Minn. Rule 2860.4400D, we are prohibited from requiring you to sign a general release.

All initially capitalized terms not defined herein shall have the same meaning as ascribed to them in the Franchise Agreement. Except as expressly modified hereby, the Franchise Agreement shall remain in full force and effect per its terms

ADDENDUM TO TECHY, LLC DISCLOSURE DOCUMENT AS REQUIRED BY THE STATE OF MINNESOTA

Minnesota law provides franchisees with certain rights regarding termination and nonrenewal of their franchises. As in effect in November 1990, Minn. Stat. Ann. Sec. 80.C.14, Subd. 3, 4 and 5 require, except in certain specified cases, that a franchisee be given 90 days notice of termination (with 60 days to cure) and 180 days notice for nonrenewal of a franchise agreement.

The franchisor will protect the franchisee's right to use the trademarks, service marks, trade names, logotypes or other commercial symbols or indemnify the franchisee from any loss, costs or expenses arising out of any claim, suit or demand regarding the use of the name.

Under Minn. Rule 2860.4400J, (1) a franchisee cannot waive any rights, (2) the franchisee cannot consent to the franchisor obtaining injunctive relief, although the franchisor may <u>seek</u> injunctive relief, and (3) a franchisee cannot be required to consent to waiver of a jury trial. In addition, a court will determine whether a bond is required.

Minn. Stat. §80C.21 and Minn. Rule 2860.440J prohibit us from requiring litigation to be conducted outside Minnesota. In addition, nothing in the DISCLOSURE DOCUMENT or agreement can abrogate or reduce any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to any procedure, forum or remedies provided for by the laws of the jurisdiction. Under Minn. Rule 2860.4400D, we are prohibited from requiring you to sign a general release.

ADDENDUM TO OFFERING PROSPECTUS FOR THE STATE OF NEW YORK

1. The following information is added to the cover page of the Franchise Disclosure Document:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT A OR YOUR PUBLIC LIBRARY FOR SOURCES OF INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THE FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND NEW YORK STATE DEPARTMENT OF LAW, BUREAU OF INVESTOR PROTECTION AND SECURITIES, 28 LIBERTY STREET, 21ST FLOOR, NEW YORK, NEW YORK 10005. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS WITH ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. The following is added at the end of Item 3:

Except as provided above, with regard to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

- A. No such party has an administrative, criminal, or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable civil or misdemeanor allegations.
- B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature, or financial condition of the franchise system or its business operations.
- C. No such party has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the 10 year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.
- D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought

by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.

3. The following is added to the end of Item 4:

Neither the franchisor, its affiliate, its predecessor, officers, or general partner during the 10-year period immediately before the date of the offering circular: (a) filed as debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code; (b) obtained a discharge of its debts under the bankruptcy code; or (c) was a principal officer of a company or a general partner in a partnership that either filed as a debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code or that obtained a discharge of its debts under the U.S. Bankruptcy Code during or within 1 year after that officer or general partner of the franchisor held this position in the company or partnership.

- 4. The following is added to the end of Item 5:
 - 5. The initial franchise fee constitutes part of our general operating funds and will be used as such in our discretion.
- 5. The following is added to the end of the "Summary" sections of Item 17(c), titled "Requirements for franchisee to renew or extend" and Item 17(m), entitled "Conditions for franchisor approval of transfer":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687.4 and 687.5 be satisfied.

6. The following language replaces the "Summary" section of Item 17(d), titled "**Termination by you (franchisee)**":

You may terminate the agreement on any grounds available by law.

7. The following is added to the end of the "Summary" section of Item 17(j), titled "Assignment of contract by us (franchisor):

However, no assignment will be made except to an assignee who in good faith and judgment of the franchisor, is willing and financially able to assume the franchisor's obligations under the Franchise Agreement.

8. The following is added to the end of the "Summary" sections of Item 17(v), titled "Choice of forum", and Item 17(w). titled "Choice of law":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchisor or upon the franchisee by Article 33 of the General Business Law of the State of New York.

NORTH DAKOTA ADDENDUM TO TECHY, LLC FRANCHISE AGREEMENT

This Addendum to the Techy, LLC Franchise Agreement by and between Techy, LLC and Franchisee is dated $___$, 20 $_$.

1. Section 15 of the Franchise Agreement is amended by the inclusion of the following language:

Under North Dakota Franchise Investment Law ("NDFIL") Section 51-19-09 we are prohibited from requiring you to sign a general release and therefore Section 15 of the Franchise Agreement is modified to eliminate this requirement.

2. Section 17.4 B of the Franchise Agreement is amended by the inclusion of the following language:

The Commissioner has determined that requiring franchisee to consent to termination or liquidated damages is unfair, unjust, and inequitable within the intent of Section 51-19-09 of the ("NDFIL") and accordingly Franchisor modifies Section 17.4 B of the Franchise Agreement to eliminate this requirement.

3. Section 8.2 B (2) of the Franchise Agreement is amended by the inclusion of the following language:

The Commissioner has held that covenants restricting competition contrary to Section 9-08-06 of the North Dakota Century Code, without further disclosing that such covenants may be subject to this statute are unfair, unjust, or inequitable within the intent of Section 51-19-09 of the ("NDFIL") and accordingly Franchisor modifies Section 8.2 B (2) of the Franchise Agreement to add the following "Covenants" not to compete such as those mentioned above are generally considered unenforceable in the State of North Dakota.

4. Section 19.2 (B) of the Franchise Agreement is amended by the inclusion of the following language:

The Commissioner has determined that franchise agreements, which provide that parties agree to the arbitration or mediation of dispute at a location that is remote from the site of the franchisee's business, are unfair, unjust, or inequitable within the intent of Section 51-19-09 of the ("NDFIL") and accordingly Franchisor modifies Section 19.2 (B) of the Franchise Agreement to provide the site of any arbitration or mediation be agreeable to all parties and may not be remote from the franchisee's place of business.

5. Section 19 of the Franchise Agreement is amended by the inclusion of the following language:

The Commissioner has held that requiring franchisees to consent to jurisdiction of courts outside of North Dakota is unfair, unjust, or inequitable within the intent of Section 51-19-09 of the ("NDFIL") and accordingly Franchisor modifies Section 19 of the Franchise Agreement to delete this provision.

6. Section 19 of the Franchise Agreement is amended by the inclusion of the following language:

No rights under North Dakota law are waived.

7. Section 19 of the Franchise Agreement is amended by the inclusion of the following language:

The Commissioner has held that franchise agreements, which specify that they are to be governed by the laws of a state other than North Dakota, are unfair, unjust, or inequitable within the intent of Section 51-19-09 of the ("NDFIL") and accordingly Franchisor modifies Section 19 of the Franchise Agreement to replace the word "Florida" with the word "North Dakota."

8. Section 19 of the Franchise Agreement is amended by the inclusion of the following language:

The Commissioner has determined the waiver of a trial by jury is unfair, unjust, and inequitable within the intent of Section 51-19-09 of the ("NDFIL") and therefore Franchisor modifies the Section 19 to delete this waiver whenever it occurs in Section 19 of the Franchise Agreement.

9. Section 19 of the Franchise Agreement is amended by the inclusion of the following language:

The Commissioner has determined the waiver of exemplary and punitive damages to be unfair, unjust, and inequitable within the intent of Section 51-19-09 of the ("NDFIL") and therefore Franchisor modifies Section 19 of the Franchise Agreement to delete the waiver of exemplary and punitive damages

IN WITNESS WHEREOF, each of the un Addendum, understands and consents to be bound effective the day of, 20	·
FRANCHISOR TECHY, LLC A Florida limited liability company	FRANCHISEE
Ву:	(Signature)
Title:	(Print Name)

NORTH DAKOTA ADDENDUM TO TECHY, LLC AREA DEVELOPMENT AGREEMENT

This Addendum to the Techy,	, LLC Area Develo	pment Agreement by	and between	Techy,
LLC and Developer is dated	, 20			

1. Section 12 of the Area Development Agreement is amended by the inclusion of the following language:

The Commissioner has determined that requiring franchisee to consent to termination or liquidated damages is unfair, unjust, and inequitable within the intent of Section 51-19-09 of the ("NDFIL") and accordingly Franchisor modifies Section 12 of the Area Development Agreement to eliminate this requirement.

2. Section 12 of the Area Development Agreement is amended by the inclusion of the following language:

The Commissioner has held that covenants restricting competition contrary to Section 9-08-06 of the North Dakota Century Code, without further disclosing that such covenants may be subject to this statute are unfair, unjust, or inequitable within the intent of Section 51-19-09 of the ("NDFIL") and accordingly Franchisor modifies Section 12 of the Area Development Agreement to add the following "Covenants" not to compete such as those mentioned above are generally considered unenforceable in the State of North Dakota.

3. Section 12 of the Area Development Agreement is amended by the inclusion of the following language:

The Commissioner has determined that franchise agreements, which provide that parties agree to the arbitration or mediation of dispute at a location that is remote from the site of the franchisee's business, are unfair, unjust, or inequitable within the intent of Section 51-19-09 of the ("NDFIL") and accordingly Franchisor modifies Section 12 of the Area Development Agreement to provide the site of any arbitration or mediation be agreeable to all parties and may not be remote from the franchisee's place of business.

4. Section 12 of the Area Development Agreement is amended by the inclusion of the following language:

The Commissioner has held that requiring franchisees to consent to jurisdiction of courts outside of North Dakota is unfair, unjust, or inequitable within the intent of Section 51-19-09 of the ("NDFIL") and accordingly Franchisor modifies Section 12 of the Area Development Agreement to delete this provision.

5. Section 12 of the Area Development Agreement is amended by the inclusion of the following language:

No rights under North Dakota law are waived.

6. Section 12 of the Area Development Agreement is amended by the inclusion of the following language:

The Commissioner has held that franchise agreements, which specify that they are to be governed by the laws of a state other than North Dakota, are unfair, unjust, or inequitable within the intent of Section 51-19-09 of the ("NDFIL") and accordingly Franchisor modifies Section 12 of the Area Development Agreement to replace the word "Florida" with the word "North Dakota."

7. Section 12 of the Area Development Agreement is amended by the inclusion of the following language:

The Commissioner has determined the waiver of a trial by jury is unfair, unjust, and inequitable within the intent of Section 51-19-09 of the ("NDFIL") and therefore Franchisor modifies the Section 12 to delete this waiver whenever it occurs in Section 12 of the Area Development Agreement.

8. Section 19 of the Franchise Agreement is amended by the inclusion of the following language:

The Commissioner has determined the waiver of exemplary and punitive damages to be unfair, unjust, and inequitable within the intent of Section 51-19-09 of the ("NDFIL") and therefore Franchisor modifies Section 12 of the Franchise Agreement to delete the waiver of exemplary and punitive damages

•	bound by all of its terms, and agrees it shall become
FRANCHISOR TECHY, LLC	DEVELOPER
A Florida limited liability company	
By:	
	(Signature)
Title:	
	(Print Name)

NORTH DAKOTA

ADDENDUM TO THE TECHY DISCLOSURE DOCUMENT

1. Section 17 (c) of the Franchise Disclosure Document is amended by the following:

The Commissioner has determined that it is unfair, unjust, and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law to require the franchisee to sign a general release upon renewal of the franchise agreement.

2. Section 17 (i) of the Franchise Disclosure Document is amended by the following:

The Commissioner has determined that it is unfair, unjust, and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law to require the franchisee to consent to termination or liquidated damages in the franchise agreement.

3. Section 17 (r) of the Franchise Disclosure Document is amended by the following:

The Commissioner has held that covenants restricting competition are contrary to Section 9-08-06 of the North Dakota Century Code, without further disclosing that such covenants may be subject to this statute, they are unfair, unjust, or inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law.

4. Section 17 (u) of the Franchise Disclosure Document is amended by the following:

The Commissioner has determined that franchise agreements, which provide that parties agree to arbitration or mediation of disputes at a location that is remote from the site of the franchisee's business, are unfair, unjust, and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law.

5. Section 17 (v) of the Franchise Disclosure Document is amended by the following:

The Commissioner has held that requiring franchisees to consent to the jurisdiction of courts outside of North Dakota is unfair, unjust, and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law.

6. Section 17 (w) of the Franchise Disclosure Document is amended by the following:

The Commissioner has held that franchise agreements, which specify that they are governed by the laws of another state other than North Dakota, are unfair, unjust, and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law.

FRANCHISOR	FRANCHISEE
Techy, LLC	
A Florida limited liability company	
By:	
<u> </u>	(Signature)
Title:	
	(Print Name)

WASHINGTON ADDENDUM TO TECHY, LLC FRANCHISE AGREEMENT

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

IN WITNESS WHEREOF, the parties he concurrently with the execution of the Franchise Ag	reto have duly executed and delivered this Addendu greement on this day of, 20
TECHY, LLC A Florida limited liability company	FRANCHISEE
By:	(Signature)
Title:	

(Print Name)

WASHINGTON ADDENDUM TO TECHY, LLC AREA DEVELOPMENT AGREEMENT

	This Addendum to the Area	a Development	Agreement by a	and between	Techy, LLC	and Developer
is dated	, 20_	<u>_</u> .				

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the Area Development Agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the Area Development Agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the Area Development Agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the Area Development Agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the Area Development Agreement or elsewhere are void and unenforceable in Washington.

IN WITNESS WHEREOF, the parties here concurrently with the execution of the Area Development	to have duly executed and delivered this Adderment Agreement on	
TECHY, LLC A Florida limited liability company	FRANCHISEE	
By:	(Signature)	
Title:	(Print Name)	

WASHINGTON ADDENDUM TO TECHY, LLC DISCLOSURE DOCUMENT

The State of Washington has a statute, RCW 19.100,180, which may supersede the Franchise Agreement in your relationship with the Franchisor including the areas of termination and renewal of your Franchise. There also may be court decisions which may supersede the Franchise Agreement in your relationship with the franchisor including the areas of termination and renewal of your Franchise.

For Washington Franchisees, the site of mediation or arbitration shall be determined by the mediator or arbitrator at the time of mediation or arbitration.

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 10.100 RCW shall prevail.

A release or waiver of rights executed by a Franchisee shall not include rights under the Washington Franchise Investment Protection Act except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, rights or remedies under the Act such as a right to a jury trial may not be enforceable.

Transfer fees are collectable to the extent that they may reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

The undersigned does hereby acknowledge receipt of	f this addendum.
Dated this day of, 20	
TECHY, LLC A Florida limited liability company	FRANCHISEE
By:	(Signature)
Title:	(Print Name)

EXHIBIT "I" TO THE DISCLOSURE DOCUMENT

SECURED PROMISSORY NOTE	

SECURED PROMISSORY NOTE

\$	
FOR VALUE RECEIVED, The undersigned	(hereinafter called the "MAKER"), promises to pay to
the order of Techy, LLC (hereinafter sometime, together with the hold	(hereinafter called the "MAKER"), promises to pay to ler of this note, called "PAYEE"), at 3000 South West 4th Avenue, Fort Lauderdale,
Florida 33315 or at such other place as PAYEE may designate from	
Dollars, together with interest thereon at the rate of seven percent (7	(%) per annum. Payment of the Secured Promissory Note shall be made in lawful
	llments of principal and interest of(\$) Dollars, payable on
the of each month, the first installment being due on	, 20 and the final installment being due and payable on
In the event of default in the payment of any one or more of said inst same fall due, or the failure of any maker, endorser, surety or guaran of the obligations or any part thereof or upon the happening of any of any of the following with respect to MAKER or the OBLIGORS: doe imminent or threatened; insolvency; commission of an act of bankr appointment of a committee of any creditors or liquidating agent; indebtedness; making or sending notice of an intended bulk transfer; demand, to furnish any financial information allowed by law or to performance of any of MAKER'S obligations or any of the co AGREEMENT" being executed contemporaneously herewith; if Mexecuted contemporaneously herewith; MAKER suffers any termin proceedings, suit or action (at law or in equity, or under any provision reorganization, composition, extension, arrangement, wage earner OBLIGORS; entry of judgment or issuance of a warrant of attachmost MAKER or any of the OBLIGORS of any proceeding for enforceme reason, or on the happening of any one or more of said events of defany of the OBLIGORS to declare the obligations immediately due and to waive the right to make such election. PAYEE may note the fac noted hereon such election to accelerate shall be effective. Upon any accelerated date of maturity at the rate of eight percent (8%) add on it MAKER and each of the OBLIGORS hereby severally (a) waives as or laws of Florida or any other state as to personal property; (b) waive any party and all other requirements necessary to hold MAKER and agrees that time of payment may be extended or other indulgence gra as to MAKER or any of the OBLIGORS; (d) agrees to pay after defan Promissory Note, including reasonable attorneys' fees. MAKER and each and every one of the OBLIGORS and shall inure The purpose of this loan is for business reasons and MAKER ackna agricultural purposes and that MAKER is (are) not consumers as defenses or remedies granted a "consumer" and also waive the righ law.	allments, provided MAKER has ten (10) days to cure any default when and as the tor hereof (hereinafter called the "OBLIGORS") to pay when due or perform any me or more of the following events: (a) abandonment of collateral; (b) occurrence leath (if an individual) or dissolution (if a partnership or corporation); if insolvency ruptcy; assignment for the benefit of creditors; calling of a meeting of any creditors; offering to or receiving from any creditors a composition or extension of any the whole or partial suspension or liquidation of MAKER'S business; failing, after operation in the payment of permit inspection of books or records of account; default in the payment or operation of the contained or referred to herein or in the "SECURITY IAKER defaults in any of the terms of the "SECURITY AGREEMENT" being nation of the Franchise Agreement between the parties; commencement of any on of the Bankruptcy Act or amendment thereto) for adjudication as a bankruptcy and of the Bankruptcy Act or amendment thereto) for adjudication as a bankruptcy and of the Bankruptcy Act or amendment thereto) for adjudication as a bankruptcy and of the Bankruptcy of MAKER or the OBLIGORS or against any property anist property of MAKER or any of the OBLIGORS or commencement against not of a money judgment, or in the event PAYEE shall deem itself insecure for any fault, PAYEE shall have the right at its election and without notice to MAKER or dipayable with interest to date. No delay in making such election shall be construed to acceleration, both principal and accrued interest shall bear interest from the interest or highest legal rate allowed, until paid. To this debt or any extension thereof all rights of exemption under the Constitution res demand, presentment, protest, notice of protest, notice of dishonor, suit against any of the OBLIGORS liable for the payment of the Secured Promissory Note; (c) anted without notice of or consent to such an action and without release of liability ultrall cost of collecting, or securing or attemp
property of even date herewith, a copy of which is attached he	Acto and meorporated netern.
ATTEST:	
	(SEAL)
	(SEAL)
	(SEAL)

GUARANTY

For value received and in consideration of any extension made by or upon the within Secured Promissory Note and Security Agreement (the "NOTE"), the undersigned (if more than one, jointly and severally) hereby (a) unconditionally guarantees to the PAYEE hereof, its successors and assigns, and every subsequent holder of the note (all collectively called the "PAYEE"), irrespective of the genuineness, validity, regularity, or enforceability thereof, or the obligations evidenced thereby, or of any collateral therefore, or the existence or extent of any such collateral, and irrespective of any other circumstances or condition, that all sums stated therein to be payable thereunder (principal, interest and charges) shall be promptly paid in full when due, in accordance with the provisions thereof at maturity, by acceleration or otherwise, and in case of extension of time of payment in whole or in part, all said sums shall be promptly paid when due, in accordance with the provisions thereof at maturity by acceleration or otherwise; (b) consents that from time to time, without notice to the undersigned, payment of any of said sums may be extended in whole or in part by indulgence, or otherwise, and also that any of said collateral may be sold or exchanged; and (c) agrees that the obligations of the undersigned shall be and remain unaffected (i) by any understanding or agreement that any other person, firm, or corporation was or is to sign or become bound on or for the NOTE; or (ii) by the death or bankruptcy of any one or more of the undersigned, if more than one, and in case of any such death or bankruptcy, by failure of the PAYEE to file claim against the estate of said decedent or bankrupts, as the case may be, for the amount of such decedent's or such bankrupt's liability hereunder; or (iii) by the failure of PAYEE to perfect any security interest in any collateral securing this NOTE, to the existence or extent of any such collateral. This instrument is executed under the seal of each of the undersigned.

WITNESSES:		
		(SEAL)
	(Signature)	
	(address)	(city) (state) (zip)
	(address)	(City) (state) (Zip)
		(SEAL)
	(Signature)	
	(11)	(',) (, ,) (',)
	(address)	(city) (state) (zip)

EXHIBIT "J" TO THE DISCLOSURE DOCUMENT

SECURITY AGREEMENT	

SECURITY AGREEMENT

THIS SECURITY AG	REEMI	ENT made ar	nd entered int	o this the	day of	, 20, by
and between						,
		, spo	ouse (hereina	after collectiv	vely referred to as "D	DEBTOR") and
Techy, LLC (hereinafte	er referi	red to as "SE	CURED PAF	RTY");		
			<u>W I T N E S S</u>	<u>S E T H</u> :		
WHEREAS:						
(a) been contemporaneously		-	nent (hereinaft	er sometimes	"Agreement") for a Tec	thy franchise has
(b) contemporaneously herev Secured Promissory Note	with, DE				Secured Promissory No ARTY in a sum certain a	
(c) granting unto SECURED					ess referred to in (b) aboreral as hereafter defined.	
NOW, THEREF follows:	₹ORE, ir	1 consideration	n of the mutual	l covenants cor	ontained herein, the partic	es hereto agree as
1. GRAN interest in and to all accollectively referred to as by performance of servic DEBTOR work papers contemporaneously here including without limitati a security interest in and proceeds thereof, contract now or hereafter existing forth in the Franchise Agproperty which may be (hereinafter referred to as Security Agreement or proconnection with any obligation).	ccounts, s "Accounts, ces for a for the object, with, plusion, future to all fur to rights, gor acquing greement describes s "COLI rovided i	as hereinafter unts"). Account customer, and operation of the use the intangilities accounts that initure, fixture general intangilities, all of white executed contest under the LATERAL"). in any COLLA	er defined, tog unts used herein d the accounts the Techy fran- ible rights that at are obtained to es, equipment, tibles, chattel pa- ich or any porti- atemporaneousl heading "Dese The SECURE ATERAL docu	gether with an n shall mean as receivable gruchise pursuant flow from the thereby. DEB inventory, new aper, document ion of which make the properties of Present in the properties of the present in the	a contract right to payme rowing from said right, int to the Franchise Agriche existence of each an TOR further grants to SE w or hereafter existing or its thereby, products and may be located at the DEF and proceeds from the sale roperty" or "Additional all enjoy all of the rights	ereto (hereinafter ent not yet earned together with the reement executed devery account. ECURED PARTY acquired, and all proceeds whether BTOR address ser e thereof and any Security" below s contained in the
Description of Property	New or Used	Make	Model	Vehicle License #	Vehicle Identification Number	Primary Use Of Collateral
		<u> </u>				

As "Additional Security" DEBTOR agrees to execute and deliver to SECURED PARTY a mortgage, trust deed, assignment of interest or covenant not to convey or encumber any real estate owned by DEBTOR or any guarantor. This includes property beneficially owned by DEBTOR as guarantor. The form of said instrument shall be determined by Secured Party. In conjunction with said mortgage or trust deed, DEBTOR warrants that said real

estate located at	is	unencumbere	d other	than a	mortgage t	o
	_ in the approximate amount of \$		and that	said rea	l estate has a	ın
approximate market value of \$	(Attach a separate sh	neet to show sa	ame dat	a concer	ning all othe	er
pieces of real estate).						

- 2. <u>INSURANCE</u>: The DEBTOR agrees to maintain the COLLATERAL subject to this Security Agreement in good condition and not to sell or dispose of same without prior written approval of SECURED PARTY. DEBTOR further agrees to secure the COLLATERAL and keep it insured under the Security Agreement against all loss, damage or destruction due to fire, theft and all types of physical damage, in the sum and by policies adequate at all times to protect the interest of the SECURED PARTY hereunder and otherwise satisfactory to the SECURED PARTY. DEBTOR shall, upon request, furnish the SECURED PARTY evidence of such insurance. Upon failure of DEBTOR to do so, the SECURED PARTY may procure such insurance, and in that event, DEBTOR shall pay the premium therefore upon demand, as an additional part of the obligation secured hereunder.
- 3. <u>OTHER OBLIGATIONS</u>: This Security Agreement will further cover DEBTOR'S obligations under that certain Franchise Agreement being executed contemporaneously herewith including the obligation and duty to make Royalty payments pursuant to Section 9.2 of the Franchise Agreement and Advertising Fund payments pursuant to Section 11.1 of the Franchise Agreement, both as to existing and future payments.
- 4. <u>RISK OF LOSS</u>: Under no circumstances shall SECURED PARTY be deemed to assume any responsibility for or any obligation or duty with respect to any part or all of the COLLATERAL of any nature or time whatsoever or any matters or proceedings arising out of or relating thereto. The SECURED PARTY shall not be required to take any action of any kind to collect or protect any interest in the COLLATERAL. The SECURED PARTY shall not be liable or responsible in any way for the safe keeping, care or custody of any of the COLLATERAL or for any of the loss or damage thereto, or for any diminution in the value thereof, or for any act or default of any agent or bailee of the SECURED PARTY or for the collection of any proceeds, but the same shall be at the DEBTOR'S sole risk at all times.
- 5. <u>ATTACHMENT TO PROCEEDINGS</u>: Upon the sale, exchange or other disposition of any of the COLLATERAL, the security interest created hereunder shall, without breaking continuity, and without further formality or act, continue in and attach to the instruments for the payment of money, accounts receivable, contract rights, documents of title, shipping documents, chattel paper and all other cash and non-cash proceeds of such sale, exchange or disposition. Upon loss, destruction or physical damage to the COLLATERAL, the security interest shall continue in and attach to any insurance proceeds payable as a result of such occurrence.

6. FURTHER WARRANTIES AND AGREEMENTS OF DEBTOR:

- (a) The DEBTOR will not cancel or abandon the COLLATERAL nor will DEBTOR use the COLLATERAL in violation of any law nor give a security interest in, assign, sell, lease, transfer or mortgage or in any way encumber the COLLATERAL without the prior written consent of SECURED PARTY. DEBTOR will perform or comply with the terms of any lease covering the premises wherein the COLLATERAL is located and any other orders, ordinances or laws of any governmental body or agency concerning such premises or the conduct of business therein. DEBTOR agrees to pay all rents, taxes, assessments and charges levied against the COLLATERAL and other claims which are or may become liens against the COLLATERAL or any part hereof and for the use, storage, maintenance or repair thereof, and upon DEBTOR'S failure to do so, SECURED PARTY may pay them, and when paid, the same shall be added to the unpaid principal balance due hereunder; however, SECURED PARTY shall be under to duty to pay them.
- (b) That DEBTOR will take care of the COLLATERAL, will not commit or permit any waste thereon, will keep the same repaired, and will at all times maintain the same in as good condition as it is now in, reasonable wear and tear excepted.
- (c) DEBTOR will allow SECURED PARTY, in its sole discretion and before or after default: to inspect the COLLATERAL and inspect and copy all records relating to the COLLATERAL and the obligations; to terminate, on notice to DEBTOR, DEBTOR'S authority to sell, lease, otherwise transfer, manufacture, process or assemble, or furnish under contracts of service, any COLLATERAL as to which any such permission has been given; to require DEBTOR to give possession or control of the COLLATERAL to SECURED PARTY; to take any action DEBTOR is normally required to take in order to obtain, preserve and enforce the security interest created by this

agreement, and to maintain and preserve the COLLATERAL without notice to DEBTOR, and to add the costs of same to the obligations (but SECURED PARTY is under not duty to take such actions).

- (d) That without the necessity for notice of or consent of DEBTOR, SECURED PARTY may exercise any rights of DEBTOR with respect to any COLLATERAL including, without limitation thereto, the following rights: (a) to record or register in, or otherwise transfer into, the name of SECURED PARTY or its nominee any part of COLLATERAL without disclosing that SECURED PARTY'S interest is that of a secured party; (b) to pledge or otherwise transfer any or all of the obligations and/or COLLATERAL whereupon any pledgee or transferee shall have all the rights of SECURED PARTY hereunder, and SECURED PARTY shall thereafter be fully discharged and relieved from all responsibility and liability for the COLLATERAL so transferred but shall retain all rights and powers hereunder as to all COLLATERAL not so transferred; (c) to take possession of any COLLATERAL and to receive any proceeds of any dividends and income on any COLLATERAL including money, and to hold the same as collateral or apply the same to any of the obligations, the manner, order and extent of such application to be in the sole discretion of SECURED PARTY; (d) to exercise any and all rights of voting, conversion, exchange, subscription or other rights or options pertaining to any COLLATERAL (e) and to liquidate, demand, sue for, collect compromise, receive and receipt for the cash or surrender value of any COLLATERAL.
- (e) To give notice in writing by Certified Mail to SECURED PARTY within 24 hours of the date of repossession as to any property of DEBTOR alleged to have been left on, upon or in the COLLATERAL at the time of repossession and such notice shall be an express condition precedent to any action or suit for loss or damage, and DEBTOR further agrees that SECURED PARTY may hold any such property without liability until a reasonable time after any such notice is received. DEBTOR expressly agrees if the COLLATERAL is repossessed at a time when DEBTOR is not in default, or there is a dispute as to this issue and notice of this fact is received by SECURED PARTY in writing and thereafter the COLLATERAL is returned to DEBTOR, then the damages therefore, if any, shall not exceed the fair rental value of the COLLATERAL for the time it was repossessed. DEBTOR hereby expressly consents to and invites SECURED PARTY and its agents to come upon any and all premises upon which the COLLATERAL is placed and warrants that any entry for the purpose of repossession of the COLLATERAL after default shall not be a trespass upon the premises and any such repossession shall not be a conversion of the COLLATERAL and agrees to indemnify and hold SECURED PARTY harmless against any actions, costs or expenses arising directly or remotely out of any attempt to enter such premises and repossess the COLLATERAL after default.
- (f) That SECURED PARTY shall be deemed to have exercised reasonable care in the custody and preservation of the COLLATERAL in its possession if it takes such reasonable actions for that purpose as DEBTOR shall request in writing, but SECURED PARTY shall have sole power to determine whether such actions are reasonable. Any omission to do any act not requested by DEBTOR shall not be deemed a failure to exercise reasonable care. DEBTOR shall be responsible for the preservation of the COLLATERAL and shall take all steps to preserve rights against prior parties. SECURED PARTY shall have the right to, but shall not be obligated to, preserve rights against prior parties; nor shall SECURED PARTY be liable for any failure to realize upon, or to exercise any right or power with respect to, any of the obligations or COLLATERAL, or for any delay in so doing.
- DEFAULT: In the event the DEBTOR defaults on any of the obligations secured by this security interest created hereunder, including but not limited to default in any of the provisions of the Secured Promissory Note executed contemporaneously herewith, the SECURED PARTY shall be entitled to any and all rights and remedies of a secured party or holder under the DEBTOR'S state Uniform Commercial Code, including, but not limited to, the repossession or taking of the COLLATERAL (if not already in its possession). Upon demand by SECURED PARTY, DEBTOR shall assemble the COLLATERAL and make it available to SECURED PARTY at a place designated by SECURED PARTY. Sale of the COLLATERAL may be made at any time and from time to time at any public or private sale, at the option of SECURED PARTY, without advertisement or notice to DEBTOR, except such notice as is required by law and cannot be waived. SECURED PARTY may purchase the COLLATERAL at any such sale (unless prohibited by statute) free from any equity of redemption and from all other claims. After deducting all expenses, for retaking, maintaining and selling the COLLATERAL, including court costs and attorneys' fees, and for collecting the proceeds of sale, SECURED PARTY shall have the right to apply the remainder of said proceeds in payment of, or as a reserve against any of the indebtedness owed SECURED PARTY by DEBTOR or obligations of DEBTOR to SECURED PARTY, the manner, order, and extent of such application to be in the sole discretion of SECURED PARTY. To the extent notice of any sale or other disposition of the COLLATERAL is required by law to be given to DEBTOR, the requirement of reasonable notice shall be met by sending such notice, as provided above, at least ten (10) days before the time of sale or disposition. DEBTOR shall remain liable to SECURED PARTY for the payment of any deficiency with interest as herein provided. However, SECURED PARTY shall not be obligated to

resort to any COLLATERAL but, at its election, may proceed to enforce any of the obligations in default against DEBTOR. Any notice required to be given to any person shall be deemed sufficient if mailed, postage paid, to such person's address as it appears in the Franchise Agreement, or, if none appears to any address in SECURED PARTY'S file.

- 8. <u>GENDER</u>: Plural or singular words used herein to designate, the DEBTOR shall be construed to refer to the maker or makers of this Security Agreement, whether one or more; and all covenants and agreements herein made by DEBTOR shall bind the heirs, personal representatives, successors, and assigns of DEBTOR and every option, right and privilege herein reserved or secured to SECURED PARTY shall inure to the benefit of or may be exercised by its successors and assigns.
- 9. <u>AMENDMENT</u>: No provision of this Security Agreement shall be modified, altered or limited in any way except by a written instrument expressly referring to this Security Agreement and to the provisions so modified, altered or limited, and signed by the DEBTOR and SECURED PARTY, or by their heirs, personal representatives, successors, assigns and guarantors.
- 10. <u>APPOINTMENT</u>: DEBTOR hereby irrevocably constitutes and appoints the SECURED PARTY Debtor's lawful Attorney-In-Fact for the purpose of executing form UCC-1 financing statements, continuation statements, and any other papers or documents necessary to protect the security interests created hereunder.
- 11. <u>WAIVER</u>: SECURED PARTY shall not by any act, delay, omission or otherwise be deemed to have waived any of its rights or remedies, and no waiver of any kind shall be valid unless in writing and signed by SECURED PARTY. All rights and remedies of SECURED PARTY under this Agreement and under any statute or rule of law shall be cumulative and may be exercised successively or concurrently. This Agreement shall continue in force and effect and shall secure any obligations of DEBTOR to SECURED PARTY, incurred or arising prior to the filing of record of a termination statement with respect hereto. SECURED PARTY shall have the right to correct patent errors herein.

If any term of the Security Agreement shall be held to be invalid, illegal or unenforceable under state or federal laws, the validity of all other terms hereof shall in no way be affected thereby.

12. MISCELLANEOUS PROVISIONS:

- (a) This Agreement shall be governed by and construed in accordance with the laws of the State of Florida.
- (b) This Agreement may be executed in any number of counterparts, each of which shall be deemed an original but all of which shall constitute one in the same instrument.
- (c) The headings in the paragraphs of this Agreement are inserted for convenience only and shall not constitute a part hereof.

[Signatures on page that follows.]

IN WITNESS WHEREOF, the DEBTOR has executed or caused this Security Agreement to be executed on the day and year first above written.

WITNESSES:		
	<u></u>	(SEAL)
	Signature	
	(A.11)	(0:1) \(\(\frac{1}{2}\) \(\frac{1}{2}\) \(\frac{1}2\) \(\frac{1}2\) \(\frac{1}2\) \(\frac{1}2\) \(\frac{1}2\) \
	(Address)	(City)(State)(Zip)
		(SEAL)
	Signature (Spot	
	<u> </u>	(6) (6) (7)
	(Address)	(City)(State)(Zip)
COUNTY)		
STATE OF)		
Subscribed and sworn to before me this the	_ day of	, 20
	NOTARY PUBLIC	
	MY COMMISSION EX	PIRES:

EXHIBIT "K" TO THE DISCLOSURE DOCUMENT

FRANCHISEE QUESTIONNAIRE

FRANCHISEE QUESTIONNAIRE

Prior to the final execution of a Franchise Agreement, this questionnaire must be completed in its entirety.

1. Full Name of	f Franchisee:			
2. Franchisee C	Office Location:			
[] Individual	_	[] Limited Partnership [] Other		
authorized to a		ual, indicate the capacity is chisee: (check applicable b		ndersigned is
[] General P	artner			
[] Other (ple	ease explain):			
5. Did Franchis	see receive a Franchise D	Disclosure Document?	[] Yes	[] No
7. Name of our 8. Have you diattorney, accou		ve who primarily worked v		is sale:
statement or p Franchise that		representative speaking evenues, profits or operatorate?		
answer	in the following blank	question 9, please provide lines. Attach additional ave the following lines bla	pages, if nece	

EXHIBIT "L" TO THE DISCLOSURE DOCUMENT

 	-
PRINCIPAL OWNER'S STATEMENT	

PRINCIPAL OWNER'S STATEMENT

This form must be completed by you if you have multiple owners or if you, or your franchised business, is owned by a business organization (like a corporation, partnership or limited liability company). We are relying on the truth and accuracy of this form in awarding the Franchise Agreement to you.

1. Form of Ov	vner. Franchisee is a (check of	one):	
(a)	Corporation		
(b)	Limited Liability Company		
(c)	Other		
	Specify:		
Fran	nchisee was formed under the la		
		(State)	
20, under the law under any name oth	s of the State of	rated or formed on Franchisee has not conducted busing diability company or partnership name is a list of all persons who have managements, etc.) and their positions are listed below:	and
Name of Pe	rson	Position	
-			
3. <u>Owners.</u> person who is one of additional sheets if n	f your owners and fully describ	es the full name and mailing address of es the nature of each owner's interest. Att	
Owner's Na	nme and Address	Description of Interest	
·			
governing the owner (e.g., operating agree	rship, management and other s	are copies of the documents and contra ignificant aspects of the business organization, articles of incorporation or organization	tion
		nt and complete as of, 20 extronically by any means Franchisor choo	

and if by electronic signature, the electronic signature shall be binding to the same extent as an original signature and shall be deemed an original signature with regard to this document. You are not entitled to challenge the validation or authenticity of the electronic signature or this document on the ground that it is not the original.

OWNER INDIVIDUALS
[Signature]
[Print Name]
[Signature]
[Print Name]
OR
CORPORATION, LIMITED LIABILITY COMPANY OR PARTNERSHIP
[Name]
By:
Printed Name:
Title:

EXHIBIT "M" TO THE DISCLOSURE DOCUMENT

DECEMPTS
RECEIPTS

State Effective Dates

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below.

State	Effective Date
California	Pending
Hawaii	
Illinois	Pending
Indiana	Pending
Maryland	Pending
Michigan	09/22/22
Minnesota	Pending
New York	Pending
North Dakota	Pending
Rhode Island	Pending
South Dakota	09/21/22
Virginia	Pending
Washington	Pending
Wisconsin	Pending

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

RECEIPT

(Retain this copy for your records)

This disclosure document summarizes certain provisions of the Franchise Agreement, Area Development Agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If Techy, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

New York requires that we give you this disclosure document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement for the payment of any consideration that relates to the franchise relationship.

Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

If Techy, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580 and the state agency in Exhibit "G".

The franchisor is Techy, LLC located at 3000 South West 4th Avenue, Fort Lauderdale, FL 33315. Its telephone number is (954) 931-6671.

Issuance Date: April 20, 2023

The franchise seller for this offering is ______ whose principal business address is 3000 South West 4th Avenue Fort Lauderdale, Florida 33315 and telephone number is (877) 928-8905.

Techy, LLC authorizes the respective state agencies identified in Exhibit "G" to receive service of process for it in their particular state.

I received a disclosure document dated April 20, 2023 that included the following Exhibits:

- Exhibit A Franchisee Operations Manual Table of Contents
- Exhibit B List of Current Franchisees
- Exhibit C List of Franchisees Who Have Left the System
- Exhibit D Financial Statements
- Exhibit E Franchise Agreement
- Exhibit E-1 In-Store (Walmart) Addendum to Franchise Agreement
- Exhibit E-2 Start-Up Package Addendum to Franchise Agreement
- Exhibit E-3 Techy Café Addendum to Franchise Agreement
- Exhibit E-4 Wrapped Techy Van Addendum to Franchise Agreement
- Exhibit F Area Development Agreement
- Exhibit F-2 In-Store (Walmart) Addendum to Area Development Agreement
- Exhibit G List of State Agencies/Agents for Service of Process
- Exhibit H State Specific and other Addenda and Riders
- Exhibit I Secured Promissory Note
- Exhibit J Security Agreement

	Principal Owner's Statement Receipt	
		Individual:
Date R	eceived	Prospective Franchisee Signature
		Print Name
		OR
		Legal Entity
		By:
		Name:
		On behalf of the following entity: Company Name:
Date R	eceived	Title:

RECEIPT

(Sign, date, and return this copy to us)

This disclosure document summarizes certain provisions of the Franchise Agreement, Area Development Agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If Techy, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

New York requires that we give you this disclosure document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement for the payment of any consideration that relates to the franchise relationship.

Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

If Techy, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580 and the state agency in Exhibit "G".

The franchisor is Techy, LLC, located at 3000 South West 4th Avenue, Fort Lauderdale, FL 33315. Its telephone number is (954) 931-6671.

Issuance Date: April 20, 2023

The franchise seller for this offering is ______ whose principal business address is 3000 South West 4th Avenue Fort Lauderdale, Florida 33315 and telephone number is (877)928-8905.

Techy, LLC authorizes the respective state agencies identified in Exhibit "G" to receive service of process for it in their particular state.

I received a disclosure document dated April 20, 2023 that included the following Exhibits:

- Exhibit A Franchisee Operations Manual Table of Contents
- Exhibit B List of Current Franchisees
- Exhibit C List of Franchisees Who Have Left the System
- Exhibit D Financial Statements
- Exhibit E Franchise Agreement
- Exhibit E-1 In-Store (Walmart) Addendum to Franchise Agreement
- Exhibit E-2 Start-Up Package Addendum to Franchise Agreement
- Exhibit E-3 Techy Café Addendum to Franchise Agreement
- Exhibit E-4 Wrapped Techy Van Addendum to Franchise Agreement
- Exhibit F Area Development Agreement
- Exhibit F-2 In-Store (Walmart) Addendum to Area Development Agreement
- Exhibit G List of State Agencies/Agents for Service of Process
- Exhibit H State Specific and other Addenda and Riders
- Exhibit I Secured Promissory Note
- Exhibit J Security Agreement

xhibit L	Franchisee Questionnaire Principal Owner's Statement Receipt		
		Individual:	
Date R	eceived	Prospective Franchisee Sig	nature
		Print Name	
		OR	
		Legal Entity	
		By:	
		Name:	
		On behalf of the following Company Name:	entity:
——————————————————————————————————————	eceived	Title:	